NEW STRATEGIES FOR TRANSITION METAL CATALYZED C-C AND C-N BOND

FORMATION

A Dissertation Submitted to the Graduate Faculty of the North Dakota State University of Agriculture and Applied Science

By

Praveen Kilaru

In Partial Fulfillment of the Requirements for the Degree of DOCTOR OF PHILOSOPHY

Major Department: Chemistry and Biochemistry

Fargo, North Dakota

North Dakota State University Graduate School

Title

NEW STRATEGIES FOR TRANSITION METAL CATALYZED C-C

AND C-N BOND FORMATION

By

Praveen Kilaru

The Supervisory Committee certifies that this disquisition complies with North Dakota

State University's regulations and meets the accepted standards for the degree of

DOCTOR OF PHILOSOPHY

SUPERVISORY COMMITTEE:

Prof. Pinjing Zhao

Chair

Prof. Mukund Sibi

Prof. Gregory Cook

Prof. Dean C. Webster

Approved:

08/13/2018 Date Prof. Gregory Cook

Department Chair

ABSTRACT

Transition metal catalysis emerged as an essential tool in the field of organic chemistry. In this context, transition metal catalyzed C-H bond functionalization is considered as an alluring strategy as it occurs with high atom-and-step economy. In the recent years, significant attention has been paid for the conversion of C-H bond into C-X (X = C, N, O, S, P...etc) bonds using transition metal catalysts. This thesis presents the development of new catalytic systems for the construction of C-C and C-N bonds through late transition metal-mediated C-H activation and decarboxylation reactions.

Chapter 1 introduces the background of transition metal catalyzed C-H bond functionalization. This chapter provide reported catalytic methods for the conversion of arene C-H bonds into various functional groups through transition metal mediated chelation-assisted C-H bond activation.

Chapter 2 describes the development of a new method for the synthesis of oxindoles via intramolecular alkene hydroarylation with N-aryl acrylamides using a Ru(II)/N-heterocyclic carbene (NHC) catalyst system. This reaction occurs with good substrate scope and synthetically useful tolerance of functional groups and does not require the assistance of additional directing group. Preliminary mechanistic results support a tandem sequence involving amide-directed aromatic C–H bond activation and intramolecular alkene arylmetalation.

Chapter 3 describes ruthenium-based decarboxylative alkenylation of heteroarenes through carboxylate directed C-H bond functionalization. The decarboxylative functionalization of heteroarenes occurs with high regioselectivity and a broad range of functional group tolerance. This decarboxylation proceeds without stoichiometric amounts of bases or oxidants and it is applicable for functionalization of various heteroarenes such as indole, pyrrole, thiophene,

iii

benzothiophene, and benzofuran at both C-2 and C-3 positions. The current protocol provides a straightforward approach for the synthesis of trisubstituted olefins with heteroarenes.

Chapter 4 explains the development of Rh/Ag-bimetallic catalyst system for decarboxylative amidation of *ortho*-substituted benzoic acids with 3-aryldioxazolones. The nature of *ortho*-substituents determines regioselectivity of this reaction through two forms of proposed chelation assistance: (1) A wide range of non-directing *ortho*-substituents led to *ortho*amidation products via carboxylate-directed C-H amidation and subsequent decarboxylation. (2) 2-Pyridyl and analogous DGs led to *ipso*-amidation products via DG-assisted decarboxylation and subsequent amidation.

ACKNOWLEDGEMENTS

I would like to take this wonderful opportunity to thank each and every one for their encouragement and guidance throughout my Ph.D. career at NDSU.

I would like to express my profound gratitude to my advisor Dr. Pinjing Zhao for his guidance and continuous support since my first day at NDSU. Dr. Zhao has encouraged me to pursue various projects without objection and supported me to attend many conferences in the last six years. His immense knowledge and insightful discussions about research enhanced my interest towards organometallic chemistry and catalysis. His patience and optimistic attitude definitely prove that he is an excellent advisor for my Ph.D. study. I would like to express my sincere gratitude for everything he has done for me, and I hope he will continue to encourage and support me in the future.

I am also very grateful to my committee members Prof. Mukund Sibi, Prof. Gregory Cook, and Prof. Dean C. Webster for being on my committee and their thoughtful comments. WOrMS meetings and classes by Dr.Sibi and Dr.Cook helped me in understanding the critical analysis of research in chemistry. I appreciate their precious time and continued support during my time at NDSU.

I would like to thank Dr. Angel Ugrinov for his patience in helping me to work with GC-MS, HRMS, and X-ray crystal structures. I also thank Daniel Wanner and John for training me and helping me with NMR studies whenever needed. I also thank Amy, Wendy, Dionna Martel, Jeff Scholl and Cole Larsen for their constant help and for taking care of all my administrative requirements.

I am intensely grateful to the department of Chemistry and Biochemistry teaching faculty, Greg Oswald and Karla Wohlers for having confidence and belief in my teaching abilities.

V

I am also very grateful to all my fellow labmates, Dr. Jing Zhang, Dr. Rajith. S. Manan, Dr. Junfeng Liu, Sunil Acharya, Kushal Das, Jian Wang and Erik Moran for their continuous support. My special thanks to Sunil and Kushal for their friendship and creating an environment for stimulating discussions.

I would like to thank other research groups at NDSU, Sibi group, Cook group, Sun group, and Siva Group for their useful scientific discussions and lending chemicals whenever needed.

I would also like to thank my friends outside the lab, who have helped me and made these past six years enjoyable in Fargo, Keerthi, Pranavi, Sunitha, Harish, Buddha, Mounisha, Sirisha, Chandra, Vijay, Kishore, Kwaku, Karthik. I thank everyone of you for your friendship, encouraging advice, and positive attitude that played a vital role in affecting my life during these past six years. A special thanks goes to Ramesh Nalajala who encouraged me to pursue or choose research as a career. I appreciate Sridhar, Srinivas, Raghu, Satyanarayana, Santu, and Naresh for their help when I encountered problems.

My heartfelt thanks to my brother-in-laws Krishna Rao and Radha Krishna for their support and encouragement.

I would like to thank my parents and my brother Suresh and my family for their unconditional love and encouragement. Without their support and encouragement, this would not have been possible.

Finally, my most profound appreciation and love are dedicated to my wife Deepthi, and my lovely daughter Harshika. I would like to express my gratitude to my father-in-law Katinni Srinivasa Rao and my mother-in-law Katinni Padma for their understanding and support.

DEDICATION

To my family for their love and continuous support.

ABSTRACT	iii
ACKNOWLEDGEMENTS	v
DEDICATION	vii
LIST OF TABLES	xi
LIST OF FIGURES	xii
LIST OF SCHEMES	xiii
LIST OF ABBREVIATIONS	xix
1. TRANSITION METAL CATALYSED C-H BOND FUNCTIONALISATION	1
1.1. Introduction	1
1.2. References	
2. RUTHENIUM-CATALYSED INTRAMOLECULAR ALKENE HYDROARYLATION FOR OXINDOLE SYNTHESIS	
2.1. Background and Significance	
2.2. Optimization of Reaction Conditions	44
2.3. Substrate Scope	47
2.4. Proposed Mechanism	49
2.5. Conclusion	54
2.6. General Experimental Procedures and Reagent Availability	54
2.6.1. Experimental Procedure for Preparation of Acrylamide Substrates	55
2.6.2. General Procedure for Ru-Catalyzed Intramolecular Alkene Hydroarylatic	on 60
2.6.3. ¹ H NMR-Based Initial-Rate Kinetic Studies	61
2.6.4. Procedure for Synthesis of the Progesterone Receptor Antagonist (2.75)	64
2.7. Spectral Data for Reported Compounds	66
2.8. References	81

TABLE OF CONTENTS

3. REGIOSELECTIVE ALKENYLATION OF HETEROARENES THROUGH RUTHENIUM CATALYSED DECARBOXYLATIVE C-H BOND ACTIVATION	89
3.1. Introduction	89
3.2. Initial Results	104
3.3. Optimization of Reaction Condition	104
3.4. Substrate Scope	106
3.5. Reaction Mechanism Studies	109
3.6. Optimization of Reaction Condition	110
3.7. Substrate Scope	112
3.8. Conclusion	113
3.9. Experimental Procedures	113
3.9.1. General Experimental Procedures and Reagent Availability	113
3.9.2. Experimental Procedure for Preparation of Alkyne Substrates	114
3.9.3. General Procedure for Synthesis of Heteroarene Carboxylic Acids	115
3.9.4. General Procedure for Ru-Catalyzed Alkyne Hydroheteroarylation	116
3.9.5. Intermolecular Competition Experiment	116
3.9.6. Deuterium Labeling Study	116
3.9.7. General Procedure for Ru-Catalyzed Decarboxylative Alkenylation of Heteroarene Carboxylic Acids	117
3.10. Spectral Data of Isolated Products	117
3.11. References	133
4. CHELATION-ASSISTED DECARBOXYLATIVE ARENE AMIDATION	144
4.1. Introduction	144
4.2. Initial Results	162
4.3. Optimization of Reaction Condition	163
4.4. Substrate Scope	166

ix

4.5. Experimental procedures	172
4.5.1. General Experimental Procedures and Reagent Availability	172
4.5.2. Synthesis of Reagents for C-H Amidation	172
4.5.3. General Procedure for Decarboxylative <i>ortho</i> - Amidation of Arenecarboxylic Acids through Chelation Assisted C-H Bond activation	176
4.5.4. General Procedure for Decarboxylative <i>ipso</i> - Amidation of Arenecarboxylic Acids through Chelation Assisted C-C Bond Activation	176
4.6. Spectral Data	177
4.7. References	189
5. CONCLUSION	198

LIST OF TABLES

Table		Page
2.1.	Development of catalytic reaction conditions (Part I)	45
2.2.	Development of catalytic reaction conditions (Part II)	46
3.1.	Optimization of ruthenium catalysed decarboxylative hydroheteroarylation of internal alkynes.	105
3.2.	Optimization of ruthenium catalysed decarboxylative alkenylation of heteroarenes with activated olefins	111
4.1.	Optimization of rhodium catalyzed decarboxylative amidation (Part I)	163
4.2.	Optimization of rhodium catalyzed decarboxylative amidation (Part II)	164
4.3.	Optimization of rhodium catalyzed decarboxylative amidation (Part III)	165

LIST OF FIGURES

<u>Figure</u>	Page
2.1.	Scope of the intramolecular alkene hydroarylation for oxindole synthesis
2.2.	Kinetic isotope effect measurement through ¹ H NMR-based initial-rate kinetic studies
2.3.	Reactivity comparison through ¹ H NMR-based initial-rate kinetic studies
3.1.	Substrate scope of heteroarene 3-carboxylic acids in ruthenium catalyzed decarboxylative alkyne hydroheteroarylation with diarylacetylenes
3.2.	Substrate scope of heteroarene 2-carboxylic acids in ruthenium catalyzed decarboxylative alkyne hydro heteroarylation with diarylacetylenes
3.3.	Substrate scope of alkynes in ruthenium catalyzed decarboxylative alkyne hydro heteroarylation
3.4.	Substrate scope for ruthenium catalyzed decarboxylative alkenylation of heteroarenes
4.1.	Substrate scope for the decarbocxylative C-H amidation through C-H bond activation
4.2.	Rh(III) catalysed decarboxylative <i>ipso</i> amidation of arene carboxylic acids 169

LIST OF SCHEMES

Scher	mes	Page
1.1.	Transition metal catalyzed C-H bond activation.	1
1.2.	The first example of transition metal mediated C-H bond cleavage.	2
1.3.	The first example of ruthenium mediated C-H bond cleavage	2
1.4.	Ruthenium catalyzed alkylation of aromatic ketones with terminal alkenes initiated by carbonyl directed C-H activation.	3
1.5.	Generally accepted reaction mechanism for murai reaction	4
1.6.	Ruthenium-catalyzed alkyne hydroarylation with cyclic α-tetralone	5
1.7.	The proposed reaction mechanism for alkyne hydroarylation with aromatic ketones initiated by C-H bond oxidative addition.	6
1.8.	Iridium-catalyzed alkyne hydroarylation with 1-naphthol.	7
1.9.	Rhodium-catalyzed alkyne hydroarylation with 2-phenyl pyridine.	7
1.10.	Conversion of arene C-H bonds into various functional groups.	8
1.11.	Palladium-catalyzed direct C-H amination of oximes.	9
1.12.	Proposed mechanism for the C-H amidation.	9
1.13.	ortho Bromination and iodination of various arenes.	10
1.14.	Proposed catalytic pathways for bromination of arenes.	11
1.15.	Comparision between classical cross-coupling and direct arylation.	11
1.16.	Proposed mechanism for the palladium-catalyzed direct arylation of 2-phenyl phenol	13
1.17.	Palladium and copper catalyzed C-H trifluoromethylation.	14
1.18.	Rhodium-catalyzed directed C-H amidation using isocyanates.	15
1.19.	A plausible mechanism for the rhodium-catalyzed C-H amidation	16
1.20.	Rhodium-catalyzed directed C-H cyanation of various directing groups	17
1.21.	Proposed reaction mechanism for rhodium-catalyzed directed C-H cyanation	18

1.22.	Traditional vs. Catalytic C(Sp2)-H allylation reactions.	19
1.23.	Ruthenium-catalyzed directed C-H allylation of phenyl pyridine	19
1.24.	Proposed mechanism for ruthenium catalyzed directed C-H allylation.	20
1.25.	Palladium-catalyzed directed the C-O bond formation.	20
1.26.	Proposed mechanism for the oxidative functionalization of C-H bond	21
1.27.	Iridium catalyzed C-H borylation of arenes	22
1.28.	The mechanism for the iridium-catalyzed C-H borylation of arenes.	22
1.29.	Iridium-catalyzed directed borylation of arenes.	23
2.1.	a) Friedel-Crafts alkylation b) Transition metal catalyzed hydroarylation	31
2.2.	Rh(I) catalyzed C-H alkylation of ketimines with olefins	32
2.3.	a) Rh(I) catalysed hydroarylation of activated olefins with perfluoro arenes b) Rh(I) catalysed hydroheteroarylation of activated olefins with heteroarenes	33
2.4.	Iridium catalysed regioselective hydroheteroarylation of styrene derivatives.	34
2.5.	Iridium catalyzed enantioselective hydroheteroarylation of norbornene	35
2.6.	Ir(III) catalyzed hydroarylation of activated olefins	36
2.7.	a) Co(II) catalyzed hydroarylation of olefins b) Co(III) catalyzed hydroarylation of activated olefins.	37
2.8.	Ni catalysed hydroarylation of unactivated olefins with trifluoromethyl-substituted arenes.	38
2.9.	Ni catalysed hydroheteroarylation of unactivated alkenes.	38
2.10.	(a) Electrophilic aromatic substitution (b) Directing group (DG)-assisted tandem C-H activation/cyclometallation.	39
2.11.	Ru(III) catalyzed intramolecular hydroarylation.	40
2.12.	Palladium-catalyzed oxidative aryltrifluoromethylation of activated alkenes	40
2.13.	Annulation of aromatic imines via directed C-H activation.	41
2.14.	Enantioselective synthesis of biologically active molecules possessing dihydropyrroloindole via catalytic C–H bond activation.	42

2.15.	Cobalt-catalyzed intramolecular olefin hydroarylation.	42
2.16.	This study: alkene-tethered directing group for C-H activation.	43
2.17.	Target catalytic process for oxindole synthesis.	44
2.18.	Optimized Ru/NHC catalyst system for current study.	47
2.19.	Proposed catalyst activation process and catalytic cycles.	50
2.20.	Results from reaction mechanism studies. (a) Intramolecular competition experiments. (b) Comparing rate constants at early conversions to determine aromatic substitution effect on reactivity. (c) Deuterium labeling study and measurement of the kinetic isotope effect.	52
2.21.	Studies towards practical synthetic applications. (a) Reactions under scale-up conditions. (b) New synthesis of a progesterone receptor antagonist	53
2.22.	Synthesis of acryl amides (Procedure 1)	55
2.23.	Synthesis of acryloyl chlorides (Procedure 1).	56
2.24.	Synthesis of acryl amides (Procedure 2)	56
2.25.	Synthesis of acryl amides (Procedure 3)	57
2.26.	Synthesis of acryl amides (Procedure 4)	58
2.27.	Synthesis of acryl amides (Procedure 5)	59
2.28.	Synthesis of acryl amides (Procedure 6)	60
2.29.	Synthesis of the progesterone receptor antagonist (Step i and ii)	64
2.30.	Synthesis of the progesterone receptor antagonist (Step iii)	65
2.31.	Synthesis of the progesterone receptor antagonist (Step iv)	66
3.1.	Lewis acid mediated hydroarylation with electron-rich arenes.	89
3.2.	Transition metal catalyzed hydroarlation of alkynes using aryl halides or organometallic reagents	89
3.3.	Hydroaryation via transition metal mediated directed C-H bond activation	90
3.4.	a) Hydroindolation of alkynes with N-(2-pyridyl)-indoles b) Hydroindolation of alkynes using imine as a directing group.	91
3.5.	Palladium catalyzed dehydrogenative coupling of arenes and alkenes.	91

3.6.	Dehydrogenative coupling of olefin and akenes through directed C-H bond activation.	92
3.7.	Dehydrogenative alkenylation using COOH as a directing group	93
3.8.	a) Iridium catalyzed <i>meta</i> borylation of arenes. b) Cu catalyzed <i>meta</i> arylation of amides.	94
3.9.	Template-directed <i>meta</i> C-H bond functionalization of toluene derivatives	95
3.10.	Template-assisted para C-H olefination of toluene derivatives.	96
3.11.	Norbornene as a transient mediator for meta C–H activation	96
3.12.	Transition metal catalyzed C-H bond functionalization using traceless directing group.	97
3.13.	Rhodium-catalyzed dehydrogenative coupling of carboxylic acids with alkynes	98
3.14.	Iridium-catalyzed decarboxylative coupling of carboxylic acids and alkynes	98
3.15.	a) Palladium-catalyzed decarboxylative alkenylation of heteroarene carboxylic acids b) Rhodium-catalyzed decarboxylative <i>ortho</i> -olefination of benzoic acids	98
3.16.	a) Rhodium-catalyzed <i>ortho</i> -olefination and decarboxylation of benzoic acids. b) Traceless directing group relay strategy for <i>meta</i> -selective arylation of phenols	99
3.17.	Rh(III)-Catalyzed decarboxylative <i>ortho</i> -heteroarylation of aromatic carboxylic acids.	100
3.18.	Ru(II)-catalyzed decarboxylative ortho-amidation of aromatic acids	100
3.19.	Decarboxylative alkyne hydroarylation with benzoic acids via double chelation- assistance	101
3.20.	Substrate scope of decarboxylative alkyne hydroarylation with benzoic acids	102
3.21.	Ruthenium catalyzed decarboxylative hydroarylation of alkynes.	103
3.22.	Initial result of ruthenium catalyzed decarboxylative hydro heteroarylation with N-methyl indole-3-acid and diphenylacetylene.	104
3.23.	Results from reaction mechanism studies (a) Intermolecular competition experiments (b) Deuterium labeling study	109
3.24.	Synthesis of internal alkynes.	114
3.25.	Synthesis of indole-3-carboxylic acids.	115

4.1.	a) Ullmann & Goldberg coupling. b) Buchwald & Hartwig coupling	144
4.2.	Classical nitration pathway	145
4.3.	C-H insertion catalysis.	145
4.4.	Intramolecular C-H amination via metal-nitrenoid intermediate	146
4.5.	Rh(II) Catalyzed direct intermolecular C-H amination of arenes	147
4.6.	C-H amination via photoredox catalysis	148
4.7.	Ir-catalysed intramolecular C-H amination via photoredox catalysis	148
4.8.	The mechanism for intramolecular catalyzed C-H amination using a photocatalyst	149
4.9.	Intermolecular C-H amination of heteroarenes via energy transfer mechanism	150
4.10.	C-H amination through transition metal catalyzed C-H activation.	150
4.11.	Copper-catalyzed direct amination of electron-deficient arenes with hydroxylamines.	151
4.12.	The mechanism for copper-catalyzed intermolecular C-H amination	152
4.13.	Palladium-catalyzed intermolecular amination of arenes.	152
4.14.	a) Pd(II)-catalyzed carbazole synthesis through oxidative C–H bond amination b) Palladium-catalyzed intramolecular C–H amination of aromatics with oxime esters.	153
4.15.	Pd catalyzed intramolecular C-H amination for the synthesis of heterocycles	154
4.16.	Amination of palladacycle with an iminoiodinane reagent.	154
4.17.	Intermolecular C-H amination using electrophilic aminating agents.	155
4.18.	Rhodium-catalyzed intermolecular C-H amination using tosyl azide as aminating source.	156
4.19.	a) Cobalt-catalyzed directed C-H amidation of indole derivatives using organic azides b) Iridium-catalyzed directed C-H of amides using organic azides	157
4.20.	Rhodium-catalyzed diarylamine synthesis using aryl azides as an aminating reagent.	158
4.21.	Rhodium-catalyzed direct C-H amidation with dioxazolones	158
4.22.	Rh(I) catalyzed C-C bond activation of ketimines.	159

4.23.	a) Rh-catalyzed alkenylation through directed C-C bond cleavage. b) Rh- catalysed decarbonylative biaryl coupling through directed C-C bond activation 160
4.24.	a) Rh-Catalyzed decarbonylative coupling of alkynes and isatins for the synthesis of 2-quinolinones b) Rh-catalysed C-C bond activation of cyclopentanones using transient directing group
4.25.	Ni-Catalysed decarbonylative biaryl coupling through directed C-C Bond. Cleavage
4.26.	Plausible reaction mechanism for decarboxylative <i>ortho</i> amidation of arene carboxylic acids through chelation assisted C-H bond activation
4.27.	Control experiments for Rh(III) catalysed <i>ipso</i> amidation of arenecarboxylic acids through chelation assisted C-C bond activation
4.28.	Mechanism for Rh(III) catalysed chelation assisted <i>ipso</i> amidation
4.29.	Synthesis of tosyl azide
4.30.	Synthesis of 5,5-dimethyl-3-phenyl-1,4,2-dioxazole
4.31.	Synthesis of 5-phenyl-1,3,2,4-dioxathiazole 2-oxide
4.32.	Synthesis of 3-substituted-1,4,2-dioxazol-5-one
4.33.	Synthesis of <i>tert</i> -Butyloxycarbonylazide175
4.34.	Synthesis of benzoyl azide

LIST OF ABBREVIATIONS

acac	Acetylacetone
AcOH	Acetic acid
Ag	Silver
Bn	Benzyl
Bpin	Bis(pinacolato)diboron
BPMEN	N, N'-bis(6-R-2-pyridylmethyl)-1,2-diaminoethane
bpy	Bipyridine
Bz	Benzoyl
CMD	Concerted metalation and deprotonation
¹³ C-NMR	Nuclear magnetic resonance (detecting carbon isotope ¹³ C)
Coe	Cyclooctene
Co	Cobalt
COD	1,5-Cyclooctadiene
Ср	Cyclopentadienyl
Cp*	Pentamethylcyclopentadienyl
Cu	Copper
DABCO	1,4-diazabicyclo [2.2.2]octane
dba	Dibenzylideneacetone
DBU	1,8-Diazabicyclo[5.4.0]undec-7-ene
DCC	N, N ² -Dicyclohexylcarbodiimide
DCE	1,2-dichloroethane
DCM	Dichloromethane
DG	Directing group

DMAP	4-Dimethylaminopyridine
DME	Dimethoxyethne
DMF	Dimethylformamide
DMSO	Dimethylsulfoxide
dppBz	1,2-Bis(diphenylphosphino)benzene
Dppe	1,2-Bis(diphenylphosphino)ethane
EAS	Electrophic aromatic substitution
EDG	Electron donating group
esp	α , α , α ', α '-tetramethyl-1,3-benzenedipropionic Acid
Et	Ethyl
Fe	Iron
¹⁹ F-NMR	Nuclear magnetic resonance (detecting fluorine)
GC/MS	Gas chromatography/Mass spectrometry
HAA	Hydrogen atom abstraction
HFIP	Hexafluoroisopropanol
¹ H-NMR	Nuclear magnetic resonance (detecting protons)
HRMS	High resolution mass spectrometry
IMes	1,3-bis(2,4,6-trimethylphenyl)-1H-imidazolium chloride
Ipr	2,6-diisopropylphenylimidazolium chloride
Ir	Iridium
К	Potassium
LED	Light emitting diode
LG	Leaving group
Mg	Magnesium

MeCN	Acetonitrile
Mes	Mesityl
MS	Molecular sieves
Na	Sodium
NBS	N-Bromosuccinimide.
NCTS	N-Cyano-N-phenyl-p-toluenesulfonamide
NHC	N-Heterocyclic carbene
Ni	Nickel
NMP	N-Methyl-2-pyrrolidine
OAc	Acetate
OBz	Benzoate
OPiv	Pivalate
PCy ₃	Tricyclohexyl phosphine
P'Bu ₃	Tri-tert-butylphosphine
Pd	Palladium
Phen	Phenanthroline
PivOH	Pivalic acid
PPh3	Triphenylphosphine
Ru	Ruthenium
Rh	Rhodium
RT	Room temperature
Sb	Antimony
SIMes	4,5-Dihydro-1,3-bis(2,4,6-trimethylphenyl)-1H- imidazolium chloride
SIPr	<i>N</i> , <i>N</i> ′-(2,6-Diisopropylphenyl)dihydroimidazolium chloride

ТВНР	tert-Butyl hydroperoxide
Tf	Triflyl
THF	Tetrahydrofuran
TLC	Thin layer chromatography
ТМ	Transition metal
TMTU	Tetramethylthiourea
Ts	Tosyl
Yb	Ytterbium

1. TRANSITION METAL CATALYSED C-H BOND FUNCTIONALISATION 1.1. Introduction

Construction of Carbon-Carbon (C-C) and Carbon-Nitrogen (C-N) bonds are essential transformations in synthetic organic chemistry. Conventional methods such as cross-coupling reactions require pre-functionalized starting materials for the formation of C-C and C-N bonds. Additional steps are required to access these pre-functionalized substrates and therefore limits the substrate scope. The need for these substrates can be bypassed by the direct functionalization of ubiquitous C-H bonds. Hence C-H bond functionalization is considered as an allusive strategy for eco-friendly chemical synthesis with high atom efficiency and reduced waste production.

Transition metal catalysts are widely employed for the functionalization of inert C-H bonds. These catalysts react with C-H bonds to produce a C-M bond by a process known as C-H activation (Scheme 1.1). The resulting C-M bond is more reactive than C-H bond and can be easily converted into functional groups.





Scheme 1.1. Transition metal catalyzed C-H bond activation.

The critical issue in C-H bond activation is selective functionalization of a C-H bond within a complex molecule which has many C-H bonds. The typical strategy employed for the selective C-H functionalization is the usage of substrates that contain coordinating ligands which are termed as directing groups (DGs). The metal center coordinates to the directing group and cleaves the proximal C-H bond to form a cyclometalated intermediate. Formation of this intermediate is the driving force for the selective functionalization of C-H bond at a precisely desired position. This is termed as chelation assisted C-H bond activation.1, 2 The first example for transition metal mediated C-H bond functionalization is demonstrated by Kleiman and Dubeck using azobenzene (**1.01**) and stoichiometric amounts of Cp₂Ni (**1.02**) (Scheme 1.2.).³ The azo functional group acts as a directing group and cleaves the *ortho* C-H bond. Regioselectivity is achieved through the formation of 5-membered nickellacycle (**1.03**). Later in 1965, Chatt and coworkers reported C-H bond functionalization of naphthalene using ruthenium (Scheme 1.3).⁴ They demonstrated that C-H bond functionalization occurs with the formation of naphthyl-ruthenium(ll) hydride complex (**1.05**) via C-H oxidative addition.



Scheme 1.2. The first example of transition metal mediated C-H bond cleavage.



Scheme 1.3. The first example of ruthenium mediated C-H bond cleavage.

After these ground-breaking result, significant efforts were devoted to the catalytic functionalization of unreactive C-H bonds. The first example for catalytic C-H functionalization was reported by Murai and coworkers using ruthenium catalyst (Scheme 1.4)⁵. In their seminal work, they described *ortho*-alkylation of aromatic ketones with terminal alkenes. This report demonstrated the importance of a chelating group to achieve high selectivity in the C-H activation processes.



Scheme 1.4. Ruthenium catalyzed alkylation of aromatic ketones with terminal alkenes initiated by carbonyl directed C-H activation.

The proposed reaction mechanism for *ortho*-alkylation via C-H bond activation is illustrated in Scheme 1.5. First, coordination of transition metal to the chelating group or directing group enables the oxidative addition of C-H bond in the *ortho* position and generates a cyclometalated Ru-H intermediate (**1.08**). Next, olefin coordination and subsequent migratory insertion gives a metal alkyl intermediate (**1.13**). Lastly, reductive elimination produces the *ortho* alkylated product (**1.09**) and regenerates the initial catalyst. In this mechanism, reductive elimination was proposed as the rate-determining step.



Scheme 1.5. Generally accepted reaction mechanism for murai reaction. Following this seminal work, Murai-type C-H alkylation was extended to various functional groups such as ester,⁶ aldehydes,⁷ aldimines,⁸ pyridines,⁹ oxazolines,¹⁰ and nitrile.¹¹ This transformation proceeds with high regioselectivity without any byproducts, and it is considered as a typical example of the atom-, step-, and redox-economic reactions.¹² This pioneering work blazed the trail to an important direction of organic synthesis to advance atom-, step- and redox-economic reactions.

Later, in 1995 Murai et al. reported hydroarylation of internal alkynes using a ruthenium catalyst to generate substituted alkene products.¹³ The hydroarylation of alkynes occurred with site selectivity *ortho* to the directing carbonyl group (Scheme 1.6). α -Tetralone (**1.06b**) was reacted with internal alkynes (**1.14**) with a catalytic amount of RuH₂(CO)(PPh₃)₃ to obtain the alkenylated arenes (**1.15**). When symmetrically substituted internal alkynes such as 4-octyne were employed, the reaction occurred in favor of the *syn*-hydroarylation products. This result

indicated a *syn*-migratory insertion of alkynes. On the other hand, it was observed that the use of unsymmetrically substituted internal alkynes such as 1-phenyl-1-butyne produced all the four regio- and stereoisomers.





The mechanism for alkyne hydroarylation with aromatic ketones is described in the Scheme 1.7, which is analogous to the hydroarylation of olefins. The first step involves the formation of cyclometalated intermediate (1.08) via directed C-H bond oxidative addition. Next subsequent alkyne coordination and migratory insertion into the M-H linkage would generate alkenyl-metal intermediates 1.17 and 1.17'. Finally, reductive elimination of alkenyl-metal intermediates delivers the product and regenerates the catalyst. Electronic and steric effects rationalize the regio- and stereochemistry results. Formation of 1.17 is more favorable than 1.17' because it is electronically stable vinyl-metal intermediate while steric effects might favor the other regioisomer 1.17'. Thus, the combined electronic and steric effects of alkyne substrates make the reaction relatively challenging to achieve high regioselectivity. Due to lack of strong geometry restriction, the formation of both *E* and *Z* isomers is observed.



Scheme 1.7. The proposed reaction mechanism for alkyne hydroarylation with aromatic ketones initiated by C-H bond oxidative addition.

This alkyne hydroarylation represents the most straightforward and economical method to access trisubstituted styrene derivatives, which are omnipresent in biologically active compounds and used extensively as synthetic intermediates for fine chemicals and materials.¹⁴ In 1999, the group of Miura disclosed a new method for hydroarylation of dialkyl alkynes with 1-naphthol using iridium(I)/P'Bu₃ catalytic system in refluxing toluene (Scheme 1.8).^{14a} The transformation proceeds with high regioselectivity and stereoselectivity. Hydroarylation occurred selectively at the *ortho* position of the hydroxyl group and produced only *E*-isomers. The selective formation of *E*-isomers is achieved by using bulky phosphine ligand on the catalyst. Later in 2001, Lim and co-workers discovered another example of alkyne hydroarylation with 2-phenyl pyridines (**1.20**) using rhodium(I)/PPh₃ catalytic system (Scheme 1.9).^{14b} However, the

formation of double alkenylated products (**1.22**) is observed when the other *ortho* C-H bond is not blocked.



Scheme 1.8. Iridium-catalyzed alkyne hydroarylation with 1-naphthol.



Scheme 1.9. Rhodium-catalyzed alkyne hydroarylation with 2-phenyl pyridine.

Following Murai's pioneering study, significant efforts have been devoted to the development of directed C-H bond functionalization. C-H bond functionalization is not limited to the hydroarylation alkene and alkynes, and it has been extended to the construction of various Carbon-Carbon(C-C) and Carbon-Heteroatom (C-X, X= I, Br, F, N, O, P.. etc.) bonds. By employing various transition metal catalysts, arene C-H bonds were converted to a broad range of functional groups through chelation assisted C-H bond activation as shown in the Scheme1.10.



Scheme 1.10. Conversion of arene C-H bonds into various functional groups. Nitrogen-containing molecules are widely present in many natural products, biologically active compounds and functional materials. There has been an increased interest in the construction of C-N bond in these molecules. The most reliable method for of C-N bond formation is Buchwald-Hartwig amination using pre-functionalised aryl halides. Direct C-H amination is more straightforward and presents the advantages of accessing to the broad substrate scope and minimal waste production.¹⁵ Che and coworkers first reported the C-N formation of oximes through chelation directed/assisted C-H bond functionalization using palladium catalyst (Scheme 2.0).¹⁶ This procedure enables the functionalization of both Sp2 and Sp3 C-H bonds and occurs with high regioselectivity.



Scheme 1.11. Palladium-catalyzed direct C-H amination of oximes.

The possible mechanism for the reaction is depicted in Scheme 1.12. Formation of cyclometalated complex **1.35** through the chelation assisted C-H bond functionalization followed by insertion of nitrene (generated from the amide through the oxidation) into C-Pd leads to the formation of Carbon-nitrogen bond. Finally, upon protonation, the desired amination product is obtained (1.20b).



Scheme 1.12. Proposed mechanism for the C-H amidation.

For many decades cross coupling reactions have been used as a valuable tool in synthetic organic chemistry. Aryl halides are widely utilized as reaction partners in the cross-coupling reactions and precursor for the synthesis of various organometallic reagents. Developing new methods for selective bromination and iodination is highly desirable. Glorius and coworkers developed a technique for *ortho* bromination and iodination of arenes by using cationic Rh(III) through chelation assisted mechanism. The current process is compatible with the broad number of directing groups and high functional group tolerance.¹⁷



Scheme 1.13. ortho Bromination and iodination of various arenes.

The plausible mechanism for Rh(III) catalyzed halogenation is illustrated below (Scheme 1.14). The catalytic cycle starts with the formation of rhodacycle (**1.45**) through chelation assistance. In path A, rhodacycle reacts with NBS (**1.37**) to give the final product. Whereas in path B, rhodacycle is oxidized to Rh(V) intermediate (**1.47**). Rh(V) intermediate upon reductive elimination regenerates the catalyst and delivers the desired product.



Scheme 1.14. Proposed catalytic pathways for bromination of arenes.

Biaryls are prevalent in many pharmaceuticals, agrochemicals, and functional materials. Biaryls can be accessed by the cross-coupling reactions between aryl halides and stoichiometric amounts of organometallic reagents.



Scheme 1.15. Comparision between classical cross-coupling and direct arylation.

The synthesis of organometallic nucleophilic reagents involves many steps which produce undesired byproducts. The direct C-H arylation represents an attractive strategy with minimization of waste production.¹⁸

Ames et al. first reported the transition metal catalyzed direct intramolecular arylation with aryl halides using palladium catalyst.¹⁹ This early discovery laid the foundation for the synthesis of numerous important heterocycles through the direct C-H arylation. Later Miura and coworkers disclosed a method for intermolecular direct C-H arylation of 2-aryl phenols with aryl iodides.²⁰ This transformation occurs with high regioselectivity. The mechanism for the protocol is illustrated below as shown in Scheme 1.16. The reaction begins with the oxidative addition of palladium catalyst into the aryl iodide bond. The resulting palladium complex (**1.55**) reacts with 2-aryl phenol (**1.52**) and generates the palladium phenolate complex (**1.56**) which undergoes C-H bond functionalization through chelation assistance to give **1.57**. Finally, reductive elimination of **1.57** delivers the final product and regenerates the active catalyst.



Scheme 1.16. Proposed mechanism for the palladium-catalyzed direct arylation of 2-phenyl phenol.

Further, the direct C-H arylation was extended to various directing groups such as pyridine, amide, ketone, esters, and acids with aryl bromides, aryl triflates using different transition metal catalysts.

Trifluoromethylation of arenes and heteroarenes is an essential transformation for the synthesis of agrochemicals, building blocks for the organic materials and drug candidates.²¹ The incorporation of strong withdrawing trifluoromethyl group into the drug candidates has been known to remarkably improve many properties like catabolic activity, lipophilicity, transport rate and dipole moment.²² The two-step Swarts reaction is the classical method for the synthesis of trifluoromethyl arenes from corresponding methyl arenes.²³ However, it suffers from harsh reaction conditions and limited substrate scope. Because of the high demand for the

trifluoromethylated products, it has drawn the significant attention of synthetic chemists to develop new protocols which are practically applicable for trifluoromethylation.

Amy and coworkers developed copper catalyzed trifluoromethylation of aryl iodides by using Ruppert's Prakash reagent.²⁴ Later Buchwald *et. al*²⁵ made significant advancement by developing palladium-catalyzed trifluoromethylation of aryl chlorides. The advantage of these cross-coupling reactions is regiospecific trifluoromethylation of arenes and heteroarenes. Although considerable progress has been made, these methods require functionalization of starting materials. The direct functionalization of C-H bond bypasses the need for the preactivation of substrates. Hence, the direct trifluoromethylation of C-H bonds is an attractive strategy compared to cross-coupling reactions due to its step and atom economy. Yu et al. first developed a protocol for Pd catalyzed C-H trifluoromethylation of phenyl pyridine using electrophilic Umemoto's reagent.²⁶ Later, Qing et al. established a copper-catalyzed C-H trifluoromethylation of heteroarenes by using Rupert's Prakash regent.²⁷



Scheme 1.17. Palladium and copper catalyzed C-H trifluoromethylation.

Amides are frequently found in many natural products, drug candidates and fundamental linkages of peptides. It is noteworthy to mention that amides were used as a precursor for the synthesis of heterocyclic compounds. In the recent years, isocyanates were utilized for atom and
step economical synthesis of aryl amides employing transition metal catalysts. Hagihara and coworkers independently reported the amide synthesis through the direct addition of benzene C-H bond to isocyanate using Rh₄(CO)₁₂.²⁸ Later Ellman et al. established a protocol for the C-H amidation of phenyl pyridine using Rh(III) catalyst. They demonstrated the synthetical utility of this strategy by synthesizing the N-acyl anthranilamides through the C-H amidation of anilides (Scheme 1.18) which are present in many drug candidates.²⁹ Following this seminal report, C-H amidation was extended to various directing groups such as imines,³⁰ carboxylic acids,³¹ diazo compounds,³² and pyrazoles³³. The synthesis of heterocyclic compounds frequently involved the coupling of C-H amidation with cyclisation.³⁴





The likely mechanism for the transition metal catalyzed C-H amidation is depicted in the Scheme 1.19. The catalytic cycle is commenced by the precordination of directing the group to metal center and then the subsequent formation of cyclometalated intermediate **1.67**. Coordination and then followed by isocyanate insertion generates the metal amido intermediate. Finally, protonation of **1.68** produces the desired product and regenerates the catalyst.



Scheme 1.19. A plausible mechanism for the rhodium-catalyzed C-H amidation.

Cyano group serves as an essential functional group in the organic synthesis, and it can be easily converted into various functional groups such as imines, amines, esters aldehydes and carboxylic acids. Benzonitriles are ubiquitous in many natural products, drugs, and agrochemicals.³⁵ Classical methods for the synthesis of benzonitriles are Sandmayers³⁶ reaction and Rosenmund–von Braun³⁷ reaction. However, these reactions suffer from harsh reaction conditions and the need for stoichiometric amounts of metal cyanides. Buchwald reported the conversion of aryl halides and aryl boronic acids into benzonitriles by employing transition metal catalysts. However, these transformations require pre-functionalised substrates and generates stoichiometric amounts of waste. Introduction of cyano group in arenes through transition metal catalyzed C-H bond activation is an elegant approach to reduce the number of steps and waste production. Recently Fu and coworkers accomplished rhodium catalyzed directed C-H cyanation using less toxic and readily available N-cyano-N-phenyl-p-toluene sulfonamide (NCTS) as the cyanation reagent (Scheme 1.20). This protocol applies to various directing groups such as pyridine, pyrazole, dihydroimidazole, dihydro oxazole, and oxime.³⁸ Later, Ackermann et al. reported the cobalt catalyzed C-H cyanation using NCTS. This reaction proceeds efficiently over a broad range of substrates with excellent functional group tolerance.³⁹



Scheme 1.20. Rhodium-catalyzed directed C-H cyanation of various directing groups.

The plausible mechanism for the directed C-H cyanation is illustrated in Scheme 1.21. In the first step, the reaction of Rh(III) with substrates generate the cyclometalated intermediate **1.72** via concerted metalation-deprotonation. Coordination of NCTS (**1.70**) and then insertion of CN produces rhodacycle **1.73**. Elimination of tosyl anline from intermediate **1.73** delivers the final product and rhodium amido complex **1.74**. Upon protonation of **1.74** regenerates the catalyst.



Scheme 1.21. Proposed reaction mechanism for rhodium-catalyzed directed C-H cyanation. Allylation of arenes is considered as an essential transformation in organic synthesis because allyl benzene is a central structural motif found in many natural products and bioactive compounds.⁴⁰ Allyl group serves as a versatile intermediate in organic synthesis and can be quickly transformed into various functional groups. The most reliable method for the allylation of arenes is transition metal mediated coupling of allyl framework with aryl metal units such as boron, zinc, magnesium and aluminum reagents. However, the main drawback of these methods is the need for stoichiometric metal reagents which are prepared from the corresponding aryl halides. Although Lewis acid promoted Friedel-Crafts allylation is considered as a straightforward approach for allylation, it is limited to electron rich arenes and often produces a mixture of products. In this context, transition metal catalyzed direct C-H allylation has gained significant attention because of it's synthetic and atom efficiency.⁴¹

a) Conventional allylation with aryl metals



Scheme 1.22. Traditional vs. Catalytic C(sp²)-H allylation reactions.

In 2011, Inoue and coworkers demonstrated the ruthenium catalyzed C-H allylation of phenyl pyridine using readily available allyl acetate as a reaction partner.⁴² This reaction proceeds with broad substrate scope and good yields. With linear allyl acetate, a mixture of linear and branched products was obtained, whereas with branched allyl acetate selective linear products were obtained.



Scheme 1.23. Ruthenium-catalyzed directed C-H allylation of phenyl pyridine.

The catalytic cycle involves the first formation of sigma-allylruthenium intermediate through oxidative addition. Sigma allyl ruthenium intermediate upon reaction with phenyl pyridine generates ruthenacycle **1.90** with the elimination of AcOH. Finally, the product is formed through the reductive elimination and regenerates the catalyst.



Scheme 1.24. Proposed mechanism for ruthenium catalyzed directed C-H allylation. Selective hydroxylation of arene or alkane C-H bonds is considered as a challenging transformation in organometallic chemistry. The early example for hydroxylation of arene C-H bond was first revealed by Fujiwara using Pd(OAc)₂.⁴³ In 2005, Rybak-Akimova and Que demonstrated *ortho*-hydroxylation of benzoic acid with hydrogen peroxide using stoichiometric nonheme iron complex [Fe(II)(BPMEN)(CH₃CN)₂](ClO4)₂.⁴⁴



Scheme 1.25. Palladium-catalyzed directed the C-O bond formation.

Sanford et al. reported operationally simple Pd-catalyzed oxidative functionalization of Sp2 and Sp3 C-H bonds through chelation assisted C-H mechanism. This reaction proceeds with

high levels of efficiency and selectivity. This protocol applies to various directing groups such as pyridine, quinoline, imine, pyrazole and azobenzene.⁴⁵

The proposed catalytic cycle is described in the following Scheme 1.26. The catalytic cycle starts with the formation of cyclopalladation intermediate through the chelation assisted C-H activation followed by oxidation Pd(II) to give Pd(IV) intermediate **1.95**. Reductive elimination of carbon-heteroatom gives the final product and regenerates the catalyst. Reductive elimination occurs via intramolecular C-X bond elimination from the metal center or by the attack of an external nucleophile (X) in an S_N 2-like reaction.



Scheme 1.26. Proposed mechanism for the oxidative functionalization of C-H bond.

Arylboran reagents serve as synthetic building blocks in the transition metal catalyzed cross-coupling reactions. Conventional methods for the synthesis of these reagents is an addition of organolithium or magnesium species to borates. Direct borylation of arene C-H bonds is an elegant approach for the synthesis of aryl boron reagents, and it is thermodynamically & kinetically favorable based on the bond energies.⁴⁶ C-H borylation of arenes was first reported using HBpin, in the presence of rhodium and iridium half sandwich complexes such as Cp*Rh(η 4-C₆Me₆), Cp*Ir(PMe₃)(H)(Bpin) and (Cp*RhCl₂)₂. However, these reactions are limited by the harsh reaction conditions and low turn over numbers.^{47, 48, 49, 50}

Recently Ishima, Miyaura, and Hartwig groups disclosed a mild protocol for the C-H borylation using air stable, commercially available Ir(I) precursor in conjunction with bipyridyl ligands.⁵¹ This reaction proceeds with regioselectivity when arene is symmetrically substituted at 1,2 and 1,4 position. Whereas, in the case of 1,3-disubstituted arenes borylation occurred at the common meta position. The steric factors control regioselectivity of this C-H borylation.



Scheme 1.27. Iridium catalyzed C-H borylation of arenes.

Stoichiometric study of this C-H borylation reaction using $[IrCl(coe)_2]_2$, 2 equiv of dtbpy, and 10 equiv of B₂pin₂in mesitylene solvent revealed the formation of tris-boryl complex $[Ir(dtbpy)(coe)(Bpin)_3]$ **2.01** as shown in Scheme 1.28. The reaction of **2.01** with deuterated benzene afforded C₆D₅-Bpin in 80% yield. $[IrCl(coe)_2]_2$ is showed high catalytic activity for C-H borylation of arenes. This is the first example for C-H borylation of arenes at room temperature.



Scheme 1.28. The mechanism for the iridium-catalyzed C-H borylation of arenes.

But this protocol is not suitable for selective borylation of monosubstituted arenes. Later, in 2008 Hartwig et al. reported *ortho* borylation of arenes using hydrosilanes as a directing group. This approach can also be used for *ortho* borylation of phenol and aniline derivatives.⁵²



Scheme 1.29. Iridium-catalyzed directed borylation of arenes.

While working for the development of this thesis, various methods have been developed for transition metal catalyzed C-H bond functionalization of arenes using ruthenium, rhodium, iridium, manganese and cobalt catalysts.¹⁹ Despite the significant advancements in transition metal catalyzed C-H bond functionalization, many catalytic systems require harsh reaction conditions such as high reaction temperatures and heavy salt additives which limits the synthetic utility and functional group compatibility. In most cases, regioselectivity is achieved through the assistance of directing groups. Although chelation assisted C-H bond functionalization occurs with regioselectivity, it is accompanied by several challenges or limitations. First, additional steps are required to install and remove the directing groups. Second, the directing group on the product carried over to the product structure after the reaction which could cause over-reaction leads to a mixture of mono-and difunctionalised products. Third, directed C-H bond activation is well known for *ortho* C-H bond functionalization, but meta and *para* C-H bond functionalization remained as significant challenge.

These challenges have provoked us to develop new catalytic methods for transition metal catalyzed C-H functionalization using directing groups which can be easily removed or incorporated into the product backbone. The following chapters describe our research on the development of new and improved catalytic methods for construction of C-C and C-N bond formations through activation of sp² C-H bonds using ruthenium and rhodium catalysts. Results from the following three major projects are summarized: (1) A tethering directing Group strategy for ruthenium-catalyzed intramolecular alkene hydroarylation (2) Rhodium-catalyzed decarboxylative amination (3) Ruthenium(II)-catalyzed decarboxylative hydroheteroarylation of internal alkynes with heteroarene carboxylic Acids.

1.2. References

- Lyons.T. W.; Sanford, M. S. Palladium-Catalyzed Ligand-Directed C–H Functionalization Reactions. *Chem. Rev.* 2010, *110*, 1147–1169.
- Colby, D. A.; Bergman, R. G.; Ellman, J. A. Rhodium-Catalyzed C–C Bond Formation via Heteroatom-Directed C–H Bond Activation. *Chem. Rev.* 2010, *110*, 624–655.
- Kleiman, J. P.; Dubeck, M. The Preparation of Cyclopentadienyl [*o*-(Phenylazo)phenyl]nickel. *J. Am. Chem. Soc.* **1963**, 85, 1544-1545.
- Chatt, J.; Davidson, J. M. The Tautomerism of Arene and Ditertiary Phosphine Complexes of Ruthenium(0), and the Preparation of New Types of Hydrido-Complexes of Ruthenium(II). *J. Chem. Soc.* **1965**, 843-855.

- Murai, S.; Kakiuchi, F.; Sekine, S.; Tanaka, Y.; Kamatani, A.; Sonoda, M.; Chatani, N. Efficient Catalytic Addition of Aromatic Carbon-Hydrogen Bonds to Olefins. *Nature* 1993, *366*, 529-531.
- Sonoda, M.; Kakiuchi, F.; Kamatani, A.; Chatani, N.; Murai, S. Ruthenium-Catalyzed Addition of Aromatic Esters at the *ortho* C-H Bonds to Olefins. *Chem. Lett.* 1996, 25, 109-110.
- Kakiuchi, F.; Sato, T.; Igi, K.; Chatani, N.; Murai, S. The Ruthenium-Catalyzed Addition of C-H Bonds in Aldehydes to Olefins. *Chem. Lett.* 2001, *30*, 386-387.
- (a) Kakiuchi, F.; Yamauchi, M.; Chatani, N.; Murai, S. Ruthenium-Catalyzed Addition of Aromatic Imines at the *ortho*-C-H Bonds to Olefins. *Chem. Lett.* 1996, 25, 111-112; (b) Kakiuchi, F.; Sato, T.; Tsujimoto, T.; Yamauchi, M.; Chatani, N.; Murai, S. New Protocol for the Site-selective Alkylation and Vinylation of Aromatic Compounds. Catalyst-Specific Reactions. *Chem. Lett.* 1998, 27, 1053-1054.
- Schinkel, M.; Marek, I.; Ackermann, L. Carboxylate-Assisted Ruthenium(II)-Catalyzed Hydroarylation of Unactivated Alkenes through C-H Cleavage. *Angew. Chem. Int. Ed.* 2013, 52, 3977-3980.
- Kakiuchi, F.; Sato, T.; Yamauchi, M.; Chatani, N.; Murai, S. Ruthenium-Catalyzed Coupling of Aromatic Carbon-Hydrogen Bonds in Aromatic Imidates with Olefins. *Chem. Lett.* 1999, 28, 19-20.
- Kakiuchi, F.; Sonoda, M.; Tsujimoto, T.; Chatani, N.; Murai, S. The Ruthenium-Catalyzed Addition of C-H Bonds in Aromatic Nitriles to Olefins. *Chem. Lett.* 1999, 28, 1083-1084.

- Burns, N. Z.; Baran, P. S.; Hoffmann, R. W. Redox Economy in Organic Synthesis.
 Angew. Chem. Int. Ed. 2009, 48, 2854-2867.
- Kakiuchi, F.; Yamamoto, Y.; Chatani, N.; Murai, S. Catalytic Addition of Aromatic C-H Bonds to Acetylenes. *Chem. Lett.* 1995, *24*, 681-682.
- (a) Satoh, T.; Nishinaka, Y.; Miura, M.; Nomura, M. Iridium-Catalyzed Regioselective Reaction of 1-Naphthols with Alkynes at the *peri*-Position. *Chem. Lett.* 1999, 28, 615-616; (b) Lim, Y.-G.; Lee, K.-H.; Koo, B. T.; Kang, J.-B. Rhodium(I)-Catalyzed *ortho-*Alkenylation of 2-Phenylpyridines with Alkynes. *Tetrahedron Lett.* 2001, *42*, 7609-7612.
 (c) Kuninobu, Y.; Kikuchi, K.; Tokunaga, Y.; Nishina, Y.; Takai, K. Hydroarylation of Acetylenes, Acrylates, and Isocyanates with Heteroaromatic Compounds under Rhenium Catalysis. *Tetrahedron.* 2008, *64*, 5974-5981;
- Park, Y.; Kim, Y.; Chang, S. Transition Metal-Catalyzed C–H Amination: Scope, Mechanism, and Applications. *Chem. Rev.*, 2017, *117*, 9247–9301.
- Thu, H. Y.; Yu, W. Y.; Che, C. M. Intermolecular Amidation of Unactivated sp² and sp³ C–H Bonds via Palladium-Catalyzed Cascade C–H Activation/Nitrene Insertion. *J. Am. Chem. Soc.* 2006,128, 9048-9049.
- Schröder, N.; Wencel, J. D.; Glorius, F. High-Yielding, Versatile, and Practical [Rh(III)Cp*]-Catalyzed *ortho*Bromination and Iodination of Arenes. *J. Am. Chem. Soc.* 2012, 134, 8298–8301.
- Ackermann, L.; Vicente, R.; Kapdi. A. R. Transition-Metal-Catalyzed Direct Arylation of (Hetero)Arenes by C-H Bond Cleavage. *Angew. Chem. Int. Ed.* 2009, 48, 9792 – 9826.
- Ames, D. E.; Bull, D. Some Reactions of 3-halogenocinnolines Catalysed by Palladium Compounds. *Tetrahedron*.1982, *38*, 383-387.

- Satoh, T.; Kawamura, Y.; Miura, M.; Nomura, M. Palladium-Catalyzed Regioselective Mono- and Diarylation Reactions of 2-Phenylphenols and Naphthols with Aryl Halides. *Angew. Chem. Int. Ed.* 1997, *36*, 1740-1742.
- Purser, S.; Moore, P. R.; Swallow, S.; Gouverneur, V. Fluorine in Medicinal Chemistry. *Chem. Soc. Rev.* 2008, *37*, 320-330.
- Kirsch, P.; Bremer, M. Nematic Liquid Crystals for Active Matrix Displays: Molecular Design and Synthesis. *Angew. Chem. Int. Ed.* 2000, *39*, 4216-4235.
- 23. Swarts, F. Bull. Acad. R. Med. Belg. 1892, 24, 309.
- Oishi, M.; Kondo, H.; Amii, H. Aromatic Trifluoromethylation Catalytic in Copper. *Chem. Commun.* 2009, 0, 1909-1911.
- Cho, E. J.; Senecal, T. D.; Kinzel, T.; Zhang, Y.; Watson, D. A.; Buchwald, S. L. The Palladium-Catalyzed Trifluoromethylation of Aryl Chlorides. *Science*. 2010, *328*, *1679-1681*.
- 26. Wang, X.; Truesdale, L.; Yu, J. Q. Pd(II)-Catalyzed *ortho*-Trifluoromethylation of Arenes Using TFA as a Promoter. *J. Am. Chem. Soc.* **2010**, *132*, 3648-3649.
- Chu, L.; Qing, F. L.; Copper-Catalyzed Direct C–H Oxidative Trifluoromethylation of Heteroarenes. J. Am. Chem.Soc. 2012, 134, 1298-1304.
- Hong, P.; Yamazaki, H.; Sonogashira, K.; Hagihara, N. Rhodium Carbonyl Cluster Catalyzed Addition of Arenes to Diphenylketene and Aryl Isocyanates *Chem. Lett.* 1978, 7, 535–538.
- Hesp, K. D.; Bergman, R. G.; Ellman, J. A. Expedient Synthesis of *N*-Acyl Anthranilamides and β-Enamine Amides by the Rh(III)-Catalyzed Amidation of Aryl and Vinyl C–H Bonds with Isocyanates. *J. Am. Chem. Soc.* 2011, *133*, 11430–11433.

- a) Takebayashi, S.; Shizuno, T.; Otani, T.; Shibata, T. Rh(III)-Catalyzed Directed C–H Bond Amidation of Ferrocenes with Isocyanates. *Beilstein J. Org. Chem.* 2012, *8*, 1844–1848. b) Kuninobu, Y.; Kikuchi, K.; Tokunaga, Y.; Nishina, Y.; Takai, K. Hydroarylation of Acetylenes, Acrylates, and Isocyanates with Heteroaromatic Compounds Under Rhenium Catalysis. *Tetrahedron.* 2008, *64*, 5974–5981.c) Kuninobu, Y.; Tokunaga, Y.; Takai, K. Rhenium-Catalyzed Amidation of Heteroaromatic Compounds via C–H Bond Activation. *Chem. Lett.* 2007, *36*, 872–873.
- Shi, X. Y.; Liu, K. Y.; Fan, J.; Dong, X. F.; Wei, J. F.; Li, C. J. A Convenient SynthesisofN-Aryl Benzamides by Rhodium-Catalyzed *ortho*-Amidation and Decarboxylation of Benzoic Acids. *Chem. Eur. J.* 2015, *21*, 1900–1903.
- Han, S.; Mishra, N. K.; Sharma, S.; Park, J.; Choi, M.; Lee, S. Y.; Oh, J. S.; Jung, Y. H.;
 Kim, I. S. Synthesis of N-Sulfonylamidated and Amidated Azobenzenes Under Rhodium Catalysis. *J. Org. Chem.* 2015, *80*, 8026–8035.
- a) Hummel, J. R.; Ellman, J. A. Cobalt(III)-Catalyzed C–H Bond Amidation with Isocyanates. *Org. Lett.* 2015, *17*, 2400–2403.
- b) Li, J.; Ackermann, L. Cobalt(III)-Catalyzed Aryl and Alkenyl C–H Aminocarbonylation with Isocyanates and Acyl Azides. *Angew. Chem., Int. Ed.* **2015**, *54*, 8551–8554.
- Hummel, J. R.; Boerth, J. A.; Ellman, J. A. Transition-Metal-Catalyzed C–H Bond
 Addition to Carbonyls, Imines, and Related Polarized *π* Bonds. *Chem. Rev.* 2017, *117*, 9163–9227.
- 35. Larock, R.C. Comprehensive Organic Transformations, *Wiley-VCH, Weinheim*, 1999.
- 36. Sandmeyer, T. Ber. Dtsch. Chem. Ges. 1884, 17, 1633.
- 37. Rosenmund, K. W.; Struck, E. Ber. Dtsch. Chem. Ges., 1919, 2, 1749.

- Gong, T. J.; Xiao, B.; Cheng, W. M.; Su, W.; Xu, J.; Liu, Z. J.; Liu, L.; Fu, Y. Rhodium-Catalyzed Directed C–H Cyanation of Arenes with *N*-Cyano-*N*-phenyl-*p*toluenesulfonamide. *J. Am. Chem. Soc.* 2013, *135*, 10630-10633.
- Li, J.; Ackermann, L. Cobalt-Catalyzed C-H Cyanation of Arenes and Heteroarenes. Angew. Chem. Int. Ed. 2015, 54, 3635-3638.
- 40. a). Ni, G.; Zhang, Q.-J.; Zheng, Z.-F.; Chen, R.-Y.; Yu, D.-Q. 2-Arylbenzofuran Derivatives from Morus Cathayana. *J. Nat. Prod.* 2009, *72*, 966–968. b). Trost, B. M.; Thiel, O. R.; Tsui, H.-C. *J. Am. Chem. Soc.* 2002, *124*, 11616–11617. c). Trost, B. M.; Toste, F. D. Enantioselective Total Synthesis of (–)-Galanthamine. *J. Am. Chem. Soc.* 2000, *122*, 11262–11263.
- Mishra, N. K.; Sharma, S.; Park, J.; Han, S.; Kim, I. S. Recent Advances in Catalytic C(sp²)–H Allylation Reactions. *ACS Catal.* 2017, *7*, 2821–2847.
- 42. Oi, S.; Tanaka, Y.; Inoue. Y. *ortho*-Selective Allylation of 2-Pyridylarenes with Allyl Acetates Catalyzed by Ruthenium Complexes. *Organometallics*. **2006**, *25*, 4773-4778.
- Jintoku, T.; Nishimura, K.; Takaki, K.; Fujiwara, Y. Thermal Activation of Alkane C–H
 Bonds by Palladium Catalysts. Carbonylation of Alkanes with Carbon Monoxide. *Chem. Lett.* 1990, *18*, 1687-1688.
- Taktak, S.; Flook, M.; Foxman, B. M.; Que, L.; Rybak-Akimova, E. V.; Akimova, R. *ortho*-Hydroxylation of Benzoic Acids with Hydrogen Peroxide at a Non-heme Iron Center. *Chem. Commun.* 2005, 0, 5301-5303.
- Desai, L. V.; Hull, K. L.; Sanford, M. S. Palladium-Catalyzed Oxygenation of Unactivated sp³ C–H Bonds. *J. Am. Chem. Soc.* 2004, *126*, 9542-9543.

- 46. Mkhalid, I. A. I.; Barnard, J. H.; Marder, T. B.; Murphy, J. M.; Hartwig, J. F. C-H Activation for the Construction of C-B bonds. *Chem. Rev.* **2010**, *110*, 890-931.
- Chen, H.; Schlecht, S.; Semple, T. C.; Hartwig, J. F. Thermal, Catalytic, Regiospecific Functionalization of Alkanes. *Science*. 2000, 287, 1995-1997.
- 48. Tse, M. K.; Cho, J.-Y.; Smith, M. R. Regioselective Aromatic Borylation in an Inert Solvent. *Org. Lett.* **2001**, *3*, 2831-2833.
- 49. Iverson, C. N.; Smith, M. R. Stoichiometric and Catalytic B–C Bond Formation from Unactivated Hydrocarbons and Boranes. *J. Am. Chem. Soc.* **1999**, *121*, 7696-7697.
- Cho, J.-Y.; Iverson, C. N.; Smith, M. R. Steric and Chelate Directing Effects in Aromatic Borylation. J. Am. Chem. Soc. 2000, 122, 12868-12869.
- Ishiyama, T.; Takagi, J.; Ishida, K.; Miyaura, N.; Anasrasi, N. R.; and J. F. Hartwig. Mild Iridium-Catalyzed Borylation of Arenes. High Turnover Numbers, Room Temperature Reactions, and Isolation of a Potential Intermediate. *J. Am. Chem. Soc.*, 2002, *124*, 390– 391.
- Boebel, T. A.; and Hartwig, J. F. Silyl-Directed, Iridium-Catalyzed *ortho*-Borylation of Arenes. A One-Pot *ortho*-Borylation of Phenols, Arylamines, and Alkylarenes. *J. Am. Chem. Soc.*, 2008, 130, 7534–7535.

2. RUTHENIUM-CATALYSED INTRAMOLECULAR ALKENE HYDROARYLATION FOR OXINDOLE SYNTHESIS

2.1. Background and Significance

An important transformation in the field of synthetic organic chemistry for construction of Carbon-Carbon bonds is alkyaltion of arenes. Conventional method for the alkylation of arenes is electrophilic aromatic substitution (Scheme 2.1 a) which relies on alkylating agents such as alkyl halides and alkyl sulfonates. Though the transformation is highly efficient, it suffers from some drawbacks namely generation of stoichiometric amounts of byproducts and achieving regio and, stereo selectivity.¹



Scheme 2.1. a) Friedel-Crafts alkylation b) Transition metal catalyzed hydroarylation.

With the advances in the field of transition metal chemistry, transition metal catalyzed addition of aromatic C-H bond across the carbon-carbon double bond (Hydroarylation) has become an important strategy for the alkylation of arenes and heteroarenes (Scheme 2.1 b). Hydroarylation proceeds with high regioselectivity and occurs without byproducts. Olefins used in hydroarylation are relatively inexpensive and more readily available unlike alkyl halides and sulfonates which are generally synthesized from olefins.^{2, 3} Murai et al first revealed a protocol for the ruthenium catalyzed C-H alkylation of ketones using vinyl silanes as coupling partners. Alkylation occurs selectively at *ortho* position to the aromatic ketones with high atom efficiency.⁴ After Murai's poinnering work, transition metal-catalyzed hydroarylation of alkenes,

the formal alkene addition by an aromatic C–H bond, has become a powerful strategy for atomefficient C–H alkylation of arenes and heteroarenes. Inspired by Murai's C-H alkylation, many research groups explored broad range of Ru, Rh, Ir, Co and Ni catlysts improving scope and efficiency of the hydroarylation.⁵ In 2002, Jun and coworkers revealed a new protocol for the alkyaltion of ketimines using alkenes in the presence of Rh(I) catalyst (Scheme 2.2). This transformation is suitable for 1-alkenes and dienes wheich were causing problems in ruthenium catalysed Murai's alkylation.⁶



Scheme 2.2. Rh(I) catalyzed C-H alkylation of ketimines with olefins.

The catalytic cycle starts with the dissociation of ligand followed by coordination of substrate gives **2.07**. Subsequent oxidative addition of C-H bond generates Rh-H intermediate **2.08** which coordinates with the olefin followed by insertion into the C-Rh bond producing **2.10**. Finally, reductive elimination of **2.10** affords *ortho* alkylation of ketimine (**2.11**). Authours proposed that the reductive elimination is the rate determining step which was supported by deuterium labelling studies. Later, the Rh(I) catalysed intermolecular C-H hydroarylation was extended to directing groups such as ketone⁷, phenyl pyridine⁸, phenol and aniline⁹ derivatives.



Scheme 2.3. a) Rh(I) catalysed hydroarylation of activated olefins with perfluoro arenes b) Rh(I) catalysed hydroheteroarylation of activated olefins with heteroarenes.

Nakao and Hiyama reported a protocol for the hydroarylation of 2-vinylnaphthalene or buta-1,3-dienylbenzene with perfuoroarenes using Ni-catalyst.¹⁰ In 2008, Zhao and coworkers disclosed a new Rh(I) catalytic system for hydroarylation of activated olefins (**2.13**) with perfluoro arenes (**2.12**) (Scheme 2.3. a). This Rh(I) catalysis occurs with high catalytic efficiency and is compatible with broad range of functional groups.¹¹ The reaction proceeds via formation of C-Rh intermediate **2.15** through C-H bond functionalization. Subsequent olefin insertion into the carbon metal bond leads to the formation of Rh-alkyl intermediate **2.16**. Upon protonolysis of intermediate **2.16** delivers the desired product and regenerates the final product. This catalytic reaction works well without *ortho* directing group and C-H bond functionalization occurs at the most acidic position. In 2013 Chang et al developed a new method for hydroheteroarylation of activated olefins (**2.13**) with heteroarenes. The substrate scope of this protocol is broad and it is suitable for both electron deficient pyridine oxide and electron rich azoles (Scheme 2.3. b).¹²



Scheme 2.4. Iridium catalysed regioselective hydroheteroarylation of styrene derivatives.

In 2008 Shibata and coworkers revealed a protocol for Ir(I) catalysed C-H alkylation of ketones through directed C-H activation mechanism.¹³ This particular reaction is highly effecient in the presence of bidentate phosphine ligand and enanatioselective alkylation was acheived. The same group extended the Iridium catalysed hydroarylation to the N-protected indoles (**2.19 or 2.23**) and regioselectivity was regulated by the combination of protecting group and ligand as

depicted in Scheme 2.4.¹⁴ The Iridium catalysed hydroarylation of alkenes is compatible with various arenes such as ary amides amides, pyridine derivatives, aniline..etc.¹⁵

Another protocol for Ir(I) catalysed hydroheteroarylation of norbornene (**2.28**) was revealed by Hartwig's group which is suitable for various heteroarenes (**2.27**) such as Indole, benzofuran, benzothiophene, pyrrole, furan occuring with high yields and excellent enantiomeric excess (Scheme 2.5).¹⁶ This protocol proceeds without any directing group and alkyaltion occurs at *ortho* to the heteroatom. Deuterium labelling study reveals that the reaction goes through the oxidative addition of C-H bond.



Scheme 2.5. Iridium catalyzed enantioselective hydroheteroarylation of norbornene.

In 2015, Chang and coworkers reported a protocol for C-H hydroarylation of activated olefins using directing group in presence of Ir(III) catalyst (Scheme 2.6).¹⁷ Unlike Murai type alkylation the transformation proceeds via concerted metalation and deprotonation mechanism through the assistance of a directing group. The plausible mechanism for this reaction is depicted in the following Scheme 2.6. The catalytic cycle begins with the formation of cyclometalated intermediate **2.32** through concerted metalation and deprotonation. Coordination of olefin followed by migratory insertion into C-M bond generates seven membered cyclic intermediate **2.34**. Protonolysis of intermediate **2.34** affords final product and regenerates the final product.



Scheme 2.6. Ir(III) catalyzed hydroarylation of activated olefins.

In the recent years significant attention has been diverted towards the development of new and efficient catalytic systems using 3d transition metal catalysts because of their high abundance and low cost compared to 4d and 5d transition metals. Among the first-row metals, cobalt catalysts have gained special interest for the directed C-H bond functionalization.¹⁸ Yoshikai and coworkers developed a cobalt(II)/phenanthroline catalytic system for *ortho* alkylation of aromatic imines (**2.35**) with broad range of olefins (Scheme 2.7 a).¹⁹ The present cobalt catalysis proceeds at room temperature with good yields and high functional group

tolerance, and it is complementary to the rhodium and ruthenium catalysed hydroarylation which occurs at high reaction temperatures. In 2013 Kannai et al demonstrated the hydroarylation of activated olefins with phenyl pyridine (**2.37**) derivatives using a cationic, highvalent Co catalyst (Scheme 2.7 b).²⁰ Authors proposed that C-H bond functionalisation of arenes proceeds via concerted metalation and deprotonation through the assistance of directing group. In 2017, Li and coworkers revealed another protocol for C-2 selective alkylation of indoles with activated olefins using cobalt(III) catalysts.²¹ Inaddition to atom and step economy, cobalt(III) catalytic systems occured with high regioselectivity through the assistance of directing groups pyridine, pyrimidine, pyrazole, ketone and amides.



Scheme 2.7. a) Co(II) catalyzed hydroarylation of olefins b) Co(III) catalyzed hydroarylation of activated olefins.

Nickel catalysts were considered as a replacement for the palladium catalysed cross coupling reactions becuase of high abundance and low cost of the nickel. Recently, nickel complexes have been explored for the C-H bond functionalisation of arenes and heteroarenes. Hartwig et al reported a new protocol for hydroarylations of unactivated olefins (**2.13**) with trifluoromethyl-substituted arenes (**2.39**) without a directing group (Scheme 2.8).²² The reaction is compatible with both terminal and internal alkenes and it exhibited high selectivity to yield

linear alkylarene products. From experimental observations, authours proposed that reductive elimination of an alkylnickel–aryl intermediate to form C-C bond was the rate determining step. The linear selectivity is acheived because of the high difference in energy between the corresponding transition states for C-C bond formation step.



Scheme 2.8. Ni catalysed hydroarylation of unactivated olefins with trifluoromethyl-substituted arenes.

In 2015, the same group disclosed another method for C_2 alkylation of heteroarenes (2.27) with unactivated terminal and internal alkenes using Nickel catalytic system (Scheme 2. 9).²³ The reaction is compatible with various heteroarenes such as Indole, Pyrrole, furan, and benzofuran and it occurs with broad functional group tolerance.



Scheme 2.9. Ni catalysed hydroheteroarylation of unactivated alkenes.

Authors proposed that the reaction proceeds through the direct transfer of hydrogen from arene to alkene without formation of Ni-H intermediate which is termed as ligand-to-ligand hydrogen transfer. Reductive elimination of alkylnickel heteroaryl intermediate leads to the formation of desired alkylated heteroarene.

Recently, applications of intramolecular alkene hydroarylation towards heterocycle synthesis have drawn significant attention.^{24,25} Reports on such cyclization of arenes containing heteroatom-tethered alkene moiety have focused on two general approaches: (1) electrophilic aromatic substitution (EAS) that involves alkene activation by a Lewis-acidic metal catalyst and subsequent intramolecular alkene attack by the arene nucleophile (Scheme 2.10 a);^{24,26} (2) aromatic C–H activation/cyclometalation assisted by an *ortho*-directing group (DG) and subsequent intramolecular alkene insertion into the metal-aryl linkage (Scheme 2.10 b).^{25,27,28}



Scheme 2.10. (a) Electrophilic aromatic substitution (b) Directing group (DG)-assisted tandem C-H activation/cyclometallation.

Sames and coworkers reported a Ru(III)/Ag(I) catalytic system for an intramolecular hydroarylation of arene-ene substrates (Scheme 2.11).²⁹ This protocol works under mild reaction conditions with good yields and it enables the easy access of various heterocycles and carbocycles such as chromane, tetralin, terpenoid, dihydrocoumarin, tetrahydroquinoline, and indolocyclohexane and cyclopentane systems. Authors proposed that the reaction occurs via an electrophilic C-H activation pathway as discussed earlier in the Scheme 2.10 a. In 2016,

Shigehisa et al reported a new protocol for intramolecular hydroarylation of unactivated olefins using Co(salen) complex.³⁰



Scheme 2.11. Ru(III) catalyzed intramolecular hydroarylation.

Guosheng Liu et al used the olefin activation strategy for the aryl trifluormethylation of activated olefins using Pd(II) catalyst as shown in Scheme 2.12.³¹ This protocol provides a straightforward access to the trifluormethylated oxindole derivatives. Based on preliminary mechanistic studies authors proposed that the reaction is initiated by the activation of olefin through the coordination with palladium catalyst. Nucleophilic attack of arene on olefin produces Pd intermediate **2.56**.



Scheme 2.12. Palladium-catalyzed oxidative aryltrifluoromethylation of activated alkenes.

The intermediate **2.56** undergoes oxidation in presence of I(III) reagent and leads to formation of Pd(IV) intermediate **2.57** which upon reductive elimination delivers the final product trifluoromethylated oxindole (**2.54**) and regenerates catalyst.

Notably, these specific hydroarylation pathways have led to inherent limitations on substrate scopes. For example, the EAS approach suffers significant loss of reactivity with electron-poor arenes and prefers electron-donating aromatic substituents.²⁶ In comparison, the cyclometalation approach reduces reactivity dependence on electronic properties of arenes and has more success with asymmetric catalysis.^{25c,25d,25h}





In 2001, Ellman et al reported a method for intramolecular hydroarylation of aroatic imines in which alkene is tethered *meta* to the imine (Scheme 2.13).^{25a} Annulation occurs selectively to the more hindered *ortho* site to furnish functionalized bicyclic ring systems. This

protocol is useful for synthesis of functionalized indane, tetralane, dihydrobenzofuran, and dihydroindole derivatives from simple starting materials.

Later in 2004, Ellman et al reported the a highly enantioselective and efficient protocol for the intramolecular imine-directed C-H/olefin coupling reaction using Rh(I) catalytst in presence of chiral phosphoramidite ligands as shown in Scheme 2.14.^{25c} This methodolgy was used for the synthesis of biologically active molecules possessing dihydropyrroloindole cores.^{25d}



Scheme 2.14. Enantioselective synthesis of biologically active molecules possessing dihydropyrroloindole via catalytic C–H bond activation.

In 2013, Ding and cowrkers disclosed a cobalt-NHC-catalytic system for intramolecular olefin hydroarylation of N-allyl indole derivatives (Scheme 2.15).³² The reaction proceeds via rate limiting C-H bond activation through chelation assistance of imine directing group. This reaction provides staightforward approach for synthesis of dihydropyrroloindole and tetrahydropyridoindole derivatives under mild conditions. The regioselectivity of intramolecular



Scheme 2.15. Cobalt-catalyzed intramolecular olefin hydroarylation.

hydroarylation was acheived by the steric nature of the NHC ligand as depicted in Scheme 2.15.

However, in cyclometalation approach the substrate structures are limited by the requirement of having carbonyl derivative-based DGs *meta* to the tethered alkene moiety. We hypothesize that incorporation of DGs into the tethered alkene moiety would eliminate the need for additional DGs and achieve broader substrate scopes (Scheme 2.16). This modified cyclometalation approach requires that the metal coordination by tethered DG to be strong enough to promote *ortho*-aromatic C–H activation, yet weak enough to allow the resulting metallacycle (**A**) to undergo facile ligand substitution via DG dissociation and alkene coordination ($\mathbf{A} \rightarrow \mathbf{B}$). Subsequent ring-closure via intramolecular alkene arylmetalation ($\mathbf{B} \rightarrow \mathbf{C}$) is envisioned to occur without DG assistance.



Scheme 2.16. This study: alkene-tethered directing group for C-H activation. We herein report the first demonstration of this catalysis design with intramolecular alkene hydroarylation of N-aryl acrylamides (2.68) to form oxindoles (2.70), which are important synthetic targets due to their biological activities (Scheme 2.17).³³ Utilizing the amide carbonyl as a tethered DG for aromatic C–H activation (2.68→2.69), this Ru-catalyzed cyclization process does not require additional DG assistance and tolerates a broad range of aromatic and vinyl substituents. With the redox-neutral nature and high atom-efficiency of alkene hydroarylation, the current method for oxindole synthesis complements existing strategies such as N-aryl acrylamide cyclizations via free radical processes an Heck-type oxidative alkene functionalizations.^{25e, 33-35}



Scheme 2.17. Target catalytic process for oxindole synthesis.

2.2. Optimization of Reaction Conditions

Current catalyst development was guided by results from our previous report on Ru/Nheterocyclic carbene (NHC)-catalyzed [3+2] annulation between N-H aromatic ketimines and alkynes, which was promoted by [Ru(cod)(η^3 -methallyl)₂] pre-catalyst and IPr ligand.³⁶ The proposed mechanism shared key common features with the envisioned pathway for target oxindole synthesis, which included DG assistance for aromatic C-H activation (imine C=N vs. amide C=O) and *5-exo-trig* ring-closure by intramolecular 1,2-insertion (imine into M-alkenyl vs. alkene into M-aryl). Thus, we focused our attention on Ruthenium catalysts to evaluate conditions for a model reaction with N-methyl-N-phenyl methacrylamide (**2.68a**).

	2.68a	N O Metal Additi Tempe	catalyst, Ligand ve, Solvent, erature	2.70a	
Catalyst		Additive	Solvent	Temp (°C)	Yield (%)
[RuH ₂ (CO)(PPh ₃) ₃]		-	Dioxane	120	0
[Ru ₃ (CO) ₁₂]		-	Dioxane	120	0
$[RuCl_2(p-cymene)]_2$		-	Dioxane	120	0
[Ru(COD)(Methallyl) ₂]		NaOAc	Dioxane	120	0
$[\operatorname{RuCl}_2(\operatorname{cod})]_2$		NaOAc	Dioxane	120	43 ^a
$[\operatorname{RuCl}_2(\operatorname{cod})]_2$		AgOAc	Dioxane	120	20 ^b
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^{c}$		NaOAc	Dioxane	120	61 ^a
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^c$		AgOAc	Dioxane	120	55 ^a
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^c$		K ₂ CO ₃	Dioxane	120	20 ^b
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^c$		AgSbF ₆ + Pivalic aicd	Dioxane	120	10 ^b
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^c$		NaOAc	Toluene	120	60 ^a
[RuCl ₂ (<i>p</i> -cym	ene)] ₂ ^c	NaOAc	THF	120	30 ^b
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^{c}$		NaOAc	DCE	120	<20 ^b
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^c$		NaOAc	DME	120	<20 ^b
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^c$		NaOAc	DMF	120	<10 ^b
$[\operatorname{RuCl}_2(p\text{-cymene})]_2^{c}$		NaOAc	CH ₃ CN	120	<5 ^b

 Table 2.1. Development of catalytic reaction conditions (Part I).

0.17 mmol of Substarte, 10 mol% Catalyst, 20 mol% additive, 1 mL of Solvent.

a = Isolated yieldb = estimated based on TLC. c = 5 mol% of the catalyst

	2.68a	N O Additive Temp	talyst, Liga , Solvent,	and () 2.7	V0a	
Catalyst		Additive	Ligand	Solvent	Temp (° C)	Yield (%)
[RuCl ₂ (<i>p</i> -cym	ene)] ₂	NaOAc(20mol%)	-	Dioxane	140	73 ^{a,c}
[RuCl ₂ (<i>p</i> -cym	ene)] ₂	NaOAc(20mol%)	Phen	Dioxane	140	82 ^{a,c}
[Ru(COD)(Me	ethallyl) ₂]			Dioxane	120	0^{b}
[Ru(COD)(Me	ethallyl) ₂]		PPh ₃	Dioxane	120	<5 ^b
[Ru(COD)(Me	ethallyl) ₂]		PCy ₃	Dioxane	120	<10
[Ru(COD)(Me	ethallyl) ₂]		PMe ₃	Dioxane	120	0^{b}
[Ru(COD)(Me	ethallyl) ₂]		Phen	Dioxane	120	0^{b}
[Ru(COD)(Me	ethallyl) ₂]		IPr	Dioxane	120	28 ^a
[Ru(COD)(Me	ethallyl) ₂]		SIPr	Dioxane	120	25 ^a
[Ru(COD)(Me	ethallyl) ₂]	AcOH (1equiv)	IPr	Dioxane	120	46 ^a
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	IPr	Dioxane	120	81 ^a
-		AcOH(2equiv))	-	Dioxane	120	0
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	SIPr	Dioxane	120	80 ^a
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	SIMes	Dioxane	120	78 ^a
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	IMes	Dioxane	120	75 ^a
[Ru(COD)(Me	ethallyl) ₂]	Pivalic acid(2equiv)	IMes	Dioxane	120	<50 ^b
[Ru(COD)(Me	ethallyl) ₂]	CF ₃ COOH(2equiv)	IMes	Dioxane	120	<70 ^b
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	IPr	Dioxane	120	54 ^c
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	IPr	Toluene	120	79 ^a
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	IPr	DCE	120	<60 ^b
[Ru(COD)(Me	ethallyl) ₂]	AcOH(2equiv)	IPr	THF	120	<60 ^b

 Table 2.2. Development of catalytic reaction conditions (Part II).

0.17mmol of Substarte, 10mol% Catalyst, 20 mol% additive, 1mL of Solvent.

a = Isolated yield b = estimated based on TLC c = 5mol% catalyst and Ligand

We found that intramolecular alkene hydroarylation with **2.68a** was promoted by 10 mol% [Ru(cod)(η^3 -C₄H₇)₂]/IPr and 2 equivalents of acetic acid at 120 °C and in dioxane solvent. Under these conditions, the oxindole product **2.70a** was formed in 81% yield over 24 hours and no other byproducts were detected.³⁷



Scheme 2.18. Optimized Ru/NHC catalyst system for current study.

2.3. Substrate Scope

With the standard reaction conditions established, various N-aryl acrylamides (2.68) were studied for Ru(II)-catalyzed intramolecular hydroarylation (Scheme 2.19). In general, oxindole products were formed without the detection of 2-quinolone byproducts.³⁷ Scope of the N-aryl moiety was studied with N-aryl-N-methylmethacrylates to generate products 2.70a-I'. Good yields were acquired for substrates with *para*-electron-withdrawing groups including halogens, CF₃, NO₂, acyl, and ester groups (products 2.70d-j), although *para*-iodine and cyano functionality led to no product formation. In comparison, reduced reactivity was observed for substrates with *para*-electron-donating groups. The *p*-anisyl substrate led to product 2.70c in a moderate yield of 65%, while the *p*-tolyl substrate required the use of CF₃CO₂H additive in place of acetic acid to form 2.70b in 61% yield. With *ortho*-fluoro substituent, harsher reaction conditions of 140 °C over 48 h was necessary to give product 2.70k in 73% yield. With *meta*-fluoro substituent, mixed regioisomers 2.70I/2.70I' corresponding to C–H activation *para*- vs. *ortho*-to-F were formed in 84% combined yield and 9:1 selectivity favoring the *ortho*-isomer 2.70I'. Replacing N-methyl with another N-phenyl or N-benzyl moiety led to products 2.70m

and **2.70n** in satisfactory yields, with the latter providing a convenient synthetic handle for oxindole N-deprotection.



General reaction conditions: 2.68 (0.17 mmol, 1.0 equiv), 2.71 (0.10 equiv), 2.72 (0.10 equiv), AcOH (2.0 equiv), dioxane (1.0 mL), 120 °C, 24h; averaged isolated yields of two runs. [a] Using CF_3CO_2H in place of AcOH. [b] Reaction at 140 °C for 48 h. [c] Using 0.20 equiv 3a and 4a; reaction at 155 °C for 72 h. [d] Using 10 mol % [Ru(*p*-cymene)Cl₂] as Ru catalyst precursor and 2 0 mol% 1,10-phenanthroline ligand.

Figure 2.1. Scope of the intramolecular alkene hydroarylation for oxindole synthesis.

Scope of the alkene moiety was studied by modification of α - and β -substituents in Naryl-N-methyl substrates to form products **2.700-v**. Replacing α -methyl with ethyl and acetoxymethyl groups in methacrylamide backbone did not cause major loss of reactivity and gave products **2.700** and **2.70p** in 71% and 74% yield respectively. In contrast, the cis-2,3dimethylacrylamide analog was much less reactive and required the use of CF₃CO₂H additive, 20 mol% Ru/NHC catalyst loading, as well as 155 °C over 72 h to give 2.700 in 59% yield. Such harsher conditions were also required for the α -CF₃ analog of **2.68a** to form product **2.70q** in 61% yield, giving a rare example of synthesizing 3-CF₃-substituted oxindoles that were not accessible via established methods such as free radical cyclization.³⁴ The scope of α -substituents was further explored with N-4-nitrophenyl-N-methyl acrylamides to afford products **2.70r-t** in 69-82% yields containing 3-butyl, -methoxymethyl, and -benzyl groups. Considering the high values of spirooxindoles in drug development,³³ we attempted the synthesis of product **2.70u** with a 1-cyclopentenecarboxylic amide substrate. However, only trace amount of **2.70u** was formed under standard conditions. A modified catalyst-ligand combination of 10 mol% [Ru(pcymene)Cl₂] and 20 mol% 1,10-phenanthroline ligand allowed **2.70u** to be synthesized in 38% yield over 48 h of heating at 140 °C. Lastly, removing α -methyl from substrate **2.68a** led to total loss of reactivity and failure to form corresponding oxindole product **2.70v**.³⁸

2.4. Proposed Mechanism

Current results on structure-reactivity correlations provide several important mechanistic insights as follows: (1) The general higher reactivity for electron-poor arene substrates than electron-rich arenes (products **2.70a-j**) suggests that the hydroarylation process does not occur via electrophilic aromatic substitution. Thus, we propose an amide-directed C–H activation that occurs by a proton abstraction pathway such as concerted metalation-deprotonation (CMD),³⁹ and is either the rate-limiting step or a pre-rate-limiting equilibrium. This hypothesis resonantes with the proposed C–H activation mechanism in our prior report on Ru/NHC-catalyzed imine/alkyne [3+2] annulation.³⁶ (2) The significantly lower reactivity for substrates with

49



Scheme 2.19. Proposed catalyst activation process and catalytic cycles.

ortho-F substitution or larger alkene substituents (products **2.70k**, **2.70o**, **2.70q** and **2.70u**) suggests that the cyclization step by intramolecular alkene arylmetalation sterically inhibited by bulky alkene or arene moieties. (3) The exclusive formation of 5-membered oxindoles over 6-membered 2-quinolones further supports a pronounced steric effect on the cyclization process,
with the former favored by formation of a less crowded alkyl C–Ru bond and the later favored by electronic stabilization of the resulting enolate C–Ru bond (see Scheme 2.20 for a more detailed mechanism description).

With a major focus of current study on new method development for C-H functionalization, we carried out a series of kinetic experiments to explore details of the proposed deprotonation pathway for C–H activation (Scheme 2.21). Firstly, intramolecular competition with unsymmetrical N,N-diaryl substrates 2.68w and 2.68x under standard catalytic conditions showed slightly higher C–H alkylation reactivity for more electron-deficient aryls based on yield comparison (1.2:1 for phenyl vs. p-anisyl, and 1.1:1 for 4-fluorophenyl vs. phenyl moiety) (Scheme 2.21a). Such reactivity differences were smaller than expected and may be cuased by intramolecular electronic effects with the non-transformed aryl on reacted aryl. Secondly, initial-rate ¹H NMR kinetics with electronically differentiated substrates **2.68a**, **2.68c** and **2.68g** showed a clear reactivity disparity that favors more electron-deficient aryls, with roughly 8:5:1 relative rate for 4-trifluoromethylphenyl vs. phenyl vs. p-anisyl moiety (Scheme 2.21b). Lastly, reaction rate comparison between **2.68a** and its C_6D_5 -analog (d_5 -**2.68a**) led to the determination of a normal isotope effect ($k_{\rm H}/k_{\rm D}=2.1$) (Scheme 2.21c). Notably, the initial-rate kinetic experiment with d_5 -2.68a led to a 1.2:1 product mixture of d_4 -2.70a and d_3 -2.70a. Neither product incorporated deuterium at 3,3-dimethyl substituents, which supports the involvement of acetic acid additive as an external proton source in the hydroarylation process.⁴⁰ Compared to d_4 -2.70a, d_3 -2.70a contained one less aromatic deuterium, which can be attibuted to ortho-H/D exchange of d_5 -2.68a via an equilibrium of Ru-mediated C–H activation/cyclometalation and protonation of Ru-aryl linkage (likely by acetic acid). Overall, results from these studies are

consistent with a reversible, amide-directed aromatic C–H activation via proton abstraction, which precedes the rate-limiting alkene arylmetalation via insertion into the Ru-aryl linkage.⁴¹



Scheme 2.20. Results from reaction mechanism studies. (a) Intramolecular competition experiments. (b) Comparing rate constants at early conversions to determine aromatic substitution effect on reactivity. (c) Deuterium labeling study and measurement of the kinetic isotope effect.

To explore the potential of current method towards practical synthetic applications, we first studied hydroarylation of **2.68a** under scale-up conditions (Scheme 2.22a). Gratifyingly, solvent volume reduction enabled scale-up reactions to proceed smoothly at reduced catalyst loadings. Thus, gram-scale synthesis of **2.68a** was successfully carried out with 0.70 mM substrate concentration and 2.5 or 5 mol% catalyst loading to achieve 80% and 92% isolated

yield respectively. Next, we incorporated the hydroarylation method into a new synthetic route for an oxindole-based progesterone receptor antagonist (**2.75**) (Scheme 2.22b).⁴²



(i) Stage 1: 1.5 equiv Et_2NH , AcOEt solvent, 0 C, 5 min; then 1.5 equiv (HCHO)n, reflux for 2 h. Stage 2: SOCl₂, N₂, 50 C, 5 h. (ii) 2 equiv Et_3N , CH_2Cl_2 , 0 °C to r.t., 6 h. (iii) 10 mol% 2.71/2.72, 2 equiv AcOH, dioxane, 120 °C, 24 h. (iv) 5 mol% PdCl₂, 1 equiv 4-FC₆H₄B(OH)₂, 2 equiv K₂CO₃, 50% EtOH/H₂O, r.t., 24 h.

Scheme 2.21. Studies towards practical synthetic applications. (a) Reactions under scale-up conditions. (b) New synthesis of a progesterone receptor antagonist.

The synthesis started with turning ethylmalonic acid into 2-ethylacryloyl chloride via an established procedure.⁴³ The crude product was subjected to Et₃N-mediated amidation with 4bromo-N-methylaniline to afford N-4-bromophenyl-N-methyl-2-ethylacrylamide (**2.68y**) in 79% yield over two steps. Intramolecular hydroarylation with **2.68y** was carried out using 10 mol% Ru/NHC catalyst to form oxindole product **2.70y** in 71% yield. Lastly, Pd-catalyzed Suzuki coupling between **2.70y** and 4-fluorophenylboronic acid gave desired product **2.75** in 78% yield. Although **2.75** was acquired as a racemic mixture, the current method has the potential to be further developed into an enantioselective catalysis based on the proposed catalyst-enabled construction of quaternary chiral center at the oxindole 3-position.

2.5. Conclusion

In summary, we have developed a Ru-catalyzed intramolecular alkene hydroarylation with N-aryl acrylamides for oxindole synthesis. This redox-neutral, 5-exo-trig cyclization occurs via a proposed tandem sequence of aromatic C–H activation and intramolecular alkene arylmetalation. The amide-directed C–H activation is promoted by electron-withdrawing aromatic substituents and does not require the assistance of additional directing groups.

2.6. General Experimental Procedures and Reagent Availability

Unless otherwise noted, all manipulations were carried out under a nitrogen atmosphere using standard Schlenk-line or glove box techniques. All glassware was oven-dried for at least 1 h prior to use. Toluene and hexane solvents were degassed by purging with nitrogen for 45 min and dried with a solvent purification system (MBraun MB-SPS). THF and dichloromethane were dried over activated 3Å molecular sieves and degassed by purging with nitrogen. Other reagents and starting materials for substrate synthesis were purchased from commercial vendors and used as received. TLC plates were visualized by exposure to ultraviolet light.

Organic solutions were concentrated by rotary evaporation at ~10 torr. Flash column chromatography was performed with 40–63 microns silica gel. ¹H NMR and ¹³C NMR spectra were obtained on a 400 MHz spectrometer, and chemical shifts were recorded relative to residual protiated solvent. Both ¹H and ¹³C NMR chemical shifts were reported in parts per million downfield from tetramethylsilane ($\delta = 0$ ppm). ¹⁹F NMR spectra were obtained at 282.4

54

MHz, and all chemical shifts were reported in parts per million upfield of CF₃COOH (δ = -78.5

ppm). High resolution mass spectra were obtained at a Waters HRMS spectrometer.

2.6.1. Experimental Procedure for Preparation of Acrylamide Substrates

2.6.1.1. General Procedure for Preparation of Substrates 2.68a- 2.68g and 2.68i- 2.68n^{44, 45}



Scheme 2.22. Synthesis of acryl amides (Procedure 1).

Into a 250 mL round-bottom flask equipped with a magnetic stir-bar was added solution of aniline **2.80** (1.0 g, 1 equiv) in DCM (60 mL) and triethylamine (2 equiv). The mixture was stirred at 0 $^{\circ}$ C, and methacryloyl chloride **2.81** (1.5 equiv) was added under nitrogen atmosphere. The resulting solution was allowed to warm up to room temperature and stirred for 6 hours, followed by the addition of H₂O (150 mL) to quench excess acyl chloride. The mixture was settled in a separation funnel, and the organic layer was extracted, washed with brine (3 x 100 mL), and dried over Na₂SO₄. The solvent was removed under reduced pressure and the crude product was purified by column chromatography (0 to 6% ethyl acetate in hexanes).

2.6.1.2. Preparation of Substrates 2.680 and 2.68u



Scheme 2.23. Synthesis of acryloyl chlorides (Procedure 1).

Into a 50 mL round-bottom flask equipped with a magnetic stir-bar was added solution of acrylic acid **2.82** (1.5 g, 1 equiv) in dichloromethane (DCM, 20 mL), followed by dropwise addition of oxalyl chloride (2 equiv) and 2 drops of DMF under nitrogen atmosphere. The mixture was stirred at room temperature for 6 hours before removing all volatiles under reduced pressure. The crude product was used for the next step without purification.

The general procedure for acrylamide synthesis was applied for the reaction between corresponding N-methylaniline and crude acryloyl amide compounds.

Compound 2.680 is isolated in 76% yield as a white solid.

Compound **2.68u** is isolated in 71% yield as a white solid.

2.6.1.3. Preparation of Substrates 2.68h and 2.68t⁴⁵



Scheme 2.24. Synthesis of acryl amides (Procedure 2).

Into a 250 mL round-bottom flask equipped with Liebig condenser and a magnetic stirbar was added a solution of N-methyl- 4-nitroaniline **S1a** (1.0 g, 1 equiv) in benzene (100 mL) and potassium carbonate (1.82 g, 2 equiv). Under N₂ atmosphere, the reaction mixture was added with acryloyl chloride **2.84** (1.5 equiv), transferred to an oil bath, and refluxed under stirring for 24h. After cooling to room temperature, the reaction mixture was added with water (150 mL) and continued to with stirring for 30 min. The reaction mixture was then extracted with ethyl acetate (3 x 100 mL), and the combined organic phases were evaporated under reduced pressure to remove all volatiles. The resulting crude product was purified by column chromatography (0 to 8% ethyl acetate in hexanes).

The compound **2.68h** is isolated in 69% yield as a yellow solid.

The compound **2.68t** is isolated in 63% yield as a yellow solid.

2.6.1.4. Preparation of Substrate 2.68s⁴⁶



Scheme 2.25. Synthesis of acryl amides (Procedure 3).

i) Synthesis of intermediate 2.85: the synthetic procedure for substrate 2.68h
2.68h and 2.68ht was applied for the reaction between 2.80 and parent acryloyl chloride
(2.84) to obtain 2.85.

ii) Synthesis of intermediate **2.86**: Into a 10 mL Schlenk tube equipped with a magnetic stir-bar was added paraformaldehyde (5 equiv), DABCO (1 equiv), phenol (0.25 equiv) and 4 mL of mixed solvent (3:7 *t*-BuOH/H₂O). The mixture was stirred at 55 °C until all reactants were dissolved. Intermediate **2.85** (1.0 g, 1 equiv) was then added slowly for 5 min, and the reaction

mixture was stirred at 55 °C for three days. After cooling to room temperature, the mixture was evaporated under reduced pressure to remove *t*-BuOH and extracted with ethyl acetate (3 x 100 mL). The combined organic phases was dried over anhydrous sodium sulfate and concentrated over rotary evaporator to remove all volatiles. The resulting crude mixture was subjected to column chromatography (0 to 17% ethyl acetate in hexanes) to obtain the desired intermediate **2.86** in 45% yield as light yellow oil.

iii) Methylation of **2.86** to synthesize substrate **2.68s**: Into a 100 mL round-bottom flask equipped with a magnetic stir-bar was added a solution of intermediate **2.86** (1.5 g, 1 equiv) in DCM (40 mL), Ag₂O (2 equiv), and iodomethane (3 equiv). The resulting mixture was stirred at 40 °C and monitored by TLC to ensure the complete conversion of **2.86**. The reaction mixture was cooled to room temperature and filtered. The filtrate was evaporated under reduced pressure to remove all volatiles. The resulting crude mixture was purified by column chromatography (0 to 10% ethyl acetate in hexanes) to afford compound **2.68s** in 71% yield as a yellow solid.





Scheme 2.26. Synthesis of acryl amides (Procedure 4).

Into a 100 mL round-bottom flask equipped with a magnetic stir-bar was added a solution of butyl malonic acid **2.87** (2.0 g, 1 equiv) in dry ethyl acetate (50 mL). The solution was cooled to 5 °C and added with diethyl amine (1.5 equiv) under nitrogen atmosphere. After stirring the mixture for 5 min, paraformaldehyde (1.5 equiv) was added slowly and the resulting mixture was refluxed for 2 h. The reaction mixture was cooled to room temperature and quenched with water (100 mL). The p^H value of the solution was brought to ~1 by adding concentrated HCl before being extracted with ethyl acetate (3 x 75 mL). The combined organic phases were dried over anhydrous sodium sulfate and concentrated at rotary evaporator to remove all volatiles. The crude 2-butyl acryloyl chloride was acquired as a pale-yellow liquid and used for the next step without purification.

Synthesis of substrate **2.68r**: the synthetic procedure for substrate **2.68h** and **2.68t** was applied for the reaction between **2.80** and **2.84** to obtain **2.68r**. The compound **2.68r** is purified by column chromatography (0 to 8% ethyl acetate in hexanes) in 65% yield as yellow solid. **2.6.1.6.** *Preparation of Substrate 2.68p*⁴⁶



Scheme 2.27. Synthesis of acryl amides (Procedure 5).

- Synthesis of intermediate 2.87: the synthetic procedure for intermediate S4 was applied for the reaction with N-methyl-N-phenyl acrylamide (2.68) to afford intermediate 2.87.
- Acetylation of 2.87 to synthesize substrate 2.68p

Into a 100 mL round-bottom flask equipped with a magnetic stir-bar was added a solution of **2.87** (1.0 g, 1 equiv) in DCM (30 mL) and triethylamine (2 equiv). The mixture was cooled to 0 °C and added with acetyl chloride (1.5 equiv) under nitrogen atmosphere, then stirred at room temperature for 12h. The reaction mixture was then added with water (100 mL) to quench the excess acetyl chloride and settled in a separation funnel. The organic layer was collected, washed with brine (3 x 50 mL), and dried over anhydrous sodium sulfate. All volatiles were removed under reduced pressure and the resulting crude product was purified by column chromatography (0 to 10% ethyl acetate in hexanes) to afford the compound **2.68p** in yield 55% as a white solid.

2.6.1.7. Preparation of Substrate 2.68q



Scheme 2.28. Synthesis of acryl amides (Procedure 6).

Into a 100 mL round-bottom flask equipped with a magnetic stir-bar was added N-methyl aniline **2.80** (2 equiv), acrylic acid **2.82** (1.0 g, 1 equiv) and dichloromethane (40 mL). The mixture was cooled to -20 °C before the addition of a solution of DCC (1.3 equiv) and DMAP (0.2 equiv) in dichloromethane (10 mL). The resulting mixture was stirred overnight at room temperature, and then evaporated under reduced pressure to remove all volatiles. The crude product was subjected to column chromatography (6% ethyl acetate in hexanes) to obtain compound **2.68q** in 62% yield as a white solid.

2.6.2. General Procedure for Ru-Catalyzed Intramolecular Alkene Hydroarylation

Into a 4 mL scintillation vial equipped with a magnetic stir bar was placed [Ru(cod)(η^3 -C₄H₇)₂] (0.017 mmol), IPr (0.017 mmol), and N-aryl acrylamide substrate (0.17 mmol). Dioxane

(1.0 mL) and acetic acid (0.34 mmol) were subsequently added. The reaction mixture was stirred at room temperature for 5 minutes and the vial was sealed with a silicone-lined screw-cap and electrical tape before being transferred out of the glovebox and stirred at 120 °C for 24 hours. The reaction mixture was cooled to room temperature and evaporated under reduced pressure to remove all volatiles. Further purification was achieved by flash-column chromatography on silica.

2.6.3. ¹H NMR-Based Initial-Rate Kinetic Studies

General Procedure: Into a 4 mL scintillation vial equipped with a magnetic stir bar was placed [Ru(cod)(η^3 -C₄H₇)₂] (0.017 mmol), IPr (0.017 mmol), acrylamide substrate (0.17 mmol), and triphenylmethane (internal standard, 41.5 mg). Deuterated toluene (C₇D₈, 99.94% D, 1.0 mL) and acetic acid (2 equiv) was subsequently added, and the mixture was stirred at room temperature for 5-10 minutes to get a homogeneous solution. The reaction mixture was transferred into a screw-cap NMR tube and analyzed by ¹H NMR to determine initial substrate concentration. The NMR tube was placed in a 120 °C oil bath and the reaction progress was monitored by ¹H NMR by taking the NMR tube out of the oil bath every 20 minutes, immediately transferring to a cold water bath to cool to room temperature, and analyzed by ¹H NMR to get relative integration values of substrate and product signals against the internal standard. The resulting product concentrations were plotted against time to determine the reaction rates at early conversions (<25% conversion).



Figure 2.2. Kinetic isotope effect measurement through ¹H NMR-based initial-rate kinetic studies.



Figure 2.3. Reactivity comparison through ¹H NMR-based initial-rate kinetic studies.

2.6.4. Procedure for Synthesis of the Progesterone Receptor Antagonist (2.75)

- Synthesis of N-Aryl Acrylamide Compound 2.68y
- Synthesis of acyl chloride intermediate 2.77

In a 250 mL round bottom flask equipped with a magnetic stir bar was placed ethylmalonic acid **2.76** (2.0 g, 1 equiv) and dry ethyl acetate (50 mL). Diethylamine (1.5 equiv) was subsequently added at 5 °C under nitrogen atmosphere. After stirring the mixture at 5 °C for 5 min, paraformaldehyde (1.5 equiv) was slowly added over 5 min and then the mixture was refluxed for 2 h. The reaction mixture was cooled to room temperature and quenched with water (150 mL). The P^H value of the solution was brought to ~1 by adding concentrated HCl, and the resulting mixture was settled in a separation funnel and extracted with ethyl acetate (3 x 75 mL). The combined organic phases were dried over anhydrous sodium sulfate and evaporated under reduced pressure to remove all volatiles. The crude compound **2.77** was acquired as a paleyellow liquid and used for the next step without further purification.



Scheme 2.29. Synthesis of the progesterone receptor antagonist (Step i and ii).2.6.4.1. Synthesis of 2.68y by N-acylation with 2.77

In a 250 mL round bottom flask equipped with a magnetic stir bar was placed 4–bromo-N-methylaniline **2.78** (1.0 g, 1 equiv), dichloromethane (60 mL) and triethylamine (2 equiv). A solution of crude 2-ethylacryloyl chloride **2.77** (~2 equiv) in dry dichloromethane (10 mL) was subsequently added dropwise at 0 °C under nitrogen atmosphere. The resulting solution was stirred at room temperature for 6 hours, followed by addition of water (150 mL) to quench the excess acryl chloride. The mixture was settled in a separation funnel and extracted with dichloromethane (3 x 50 mL). The combined organic layers was washed with brine (3 x 100 mL) and then dried over anhydrous Na₂SO₄. Szolvent was removed under reduced pressure and the crude product was purified by column chromatography (0 to 5% ethyl acetate in hexanes) to give **2.68y** as white solid in 79% yield.

2.6.4.2. Synthesis of Oxindole 2.70y by Intramolecular Hydroarylation with 2.68y



Scheme 2.30. Synthesis of the progesterone receptor antagonist (Step iii).

In a 20 mL scintillation vial equipped with a magnetic stir-bar was placed [Ru(COD)(methallyl)₂] (0.10 equiv), IPr (0.10 equiv) and **2.68y** (300 mg, 1.0 equiv, 1.12 mmol). Dioxane (5 mL) was added, followed by the addition of acetic acid (2 equiv). The reaction mixture was stirred at room temperature for five minutes and the vial was sealed with a silicone-lined screw-cap before transferring out of the glove box. The mixture was stirred in a 120° C oil bath for 24 hours. After the reaction mixture was cooled to room temperature, all volatiles were removed under reduced pressure. Further purification was achieved by column chromatography (0 to 4% ethyl acetate in hexanes) to give **2.70y** as colorless oil in 71% yield.



Scheme 2.31. Synthesis of the progesterone receptor antagonist (Step iv).

In a 10 mL round bottom flask equipped with a magnetic stir bar was placed **2.70y** (213 mg, 1.0 equiv), 50% ethanol (4 mL), and PdCl₂ (0.05 equiv). The mixture was stirred at room temperature for 5 min to give a homogeneous solution, and subsequently added with potassium carbonate (2.0 equiv) and 4-FC₆H₅B(OH)₂ **2.79** (1.0 equiv). The mixture was stirred at room temperature under air for 24 hours, mixed with brine (50 mL), and settled in a separation funnel. After extraction with dichloromethane (3 x 50 mL), the combined organic layers were dried over sodium sulfate and evaporated under reduced pressure to remove all volatiles. Further purification by column chromatography (0 to 3% ethyl acetate in hexanes) gave **2.75** as colorless oil in 78% yield.

2.7. Spectral Data for Reported Compounds



1, 3, 3-trimethylindolin-2-one (2.70a): The compound **2.70a** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 81% yield as a light-yellow oil. 1H NMR (400 MHz, CDCl3): δ 7.24 – 7.16 (m, 2H), 7.02 (td, J = 7.5, 1.0 Hz, 1H), 6.82 (d, J = 7.8 Hz, 1H), 3.18 (s, 3H), 1.34 (s, 6H).

13C NMR (100 MHz, CDCl3): δ 181.2, 142.5, 135.7, 127.6, 122.4, 122.2, 108.0, 77.6, 77.2, 76.9, 44.0, 26.1, 24.3. HRMS: m/z calcd for C₁₁H₁₃NO: 176.1075; found: 176.1076.



1,3,3,5-tetramethylindolin-2-one (2.70b) : The compound 2.70b was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 61% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃): δ 7.09 – 7.05 (m, 2H), 6.77 (d, J = 7.8 Hz, 1H), 3.22 (s, 3H), 2.36 (s, 3H), 1.38 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 181.7, 140.0, 135.8, 132.2, 127.9, 123.1, 107.9, 77.4, 77.1, 76.8, 44.4, 26.3, 24.3, 21.1. HRMS: m/z calcd for C₁₂H₁₅NO: 190.1232; found: 190.1239.



5-methoxy-1, 3, 3-trimethylindolin-2-one (**2.70c**): The compound **2.70c** was prepared according to the general method described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 65% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃): δ 6.82 – 6.81 (m, 1H), 6.78 – 6.72 (m, 2H), 3.79 (s, 3H), 3.18 (s, 3H), 1.35 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 180.9, 156.07, 137.2, 136.1, 111.5, 110.0, 108.2, 77.4, 77.1, 76.8, 55.7, 44.5, 26.2, 24.3. HRMS: m/z calcd for C₁₂H₁₆NO₂: 206.1181; found: 206.1181.



5-fluoro-1, 3, 3-trimethylindolin-2-one (**2.70d**): The compound **2.70d** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 88% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 6.95 - 6.90 (m, 2H), 6.74 (dd, J = 9.1, 4.2 Hz, 1H), 3.18 (s, 3H), 1.34 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 180.8 δ 159.3 (d, J = 240.3 Hz), 138.5, 137.4 (d, J = 7.8 Hz), 113.7 (d, J = 23.4 Hz), 110.5 (d, J = 24.5 Hz), 108.4 (d, J = 8.1 Hz). 77.4, 77.1, 76.8, 44.6, 26.2, 24.2. ¹⁹F NMR (376 MHz, CDCl₃): δ -120.9. HRMS: m/z calcd for C₁₁H₁₂FNO: 194.0981; found: 194.0989.



5-chloro-1, 3, 3-trimethylindolin-2-one (**2.70e**): The compound **2.70e** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 86% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.21 – 7.15 (m, 2H), 6.75 (d, J = 8.2 Hz, 1H), 3.18 (s, 3H), 1.34 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 180.7, 141.1, 137.4, 127.7, 127.5, 122.8, 108.9, 77.4, 77.1, 76.8, 44.3, 26.2, 24.2. HRMS: m/z calcd for C₁₁H₁₂ClNO: 210.0686; found: 210.0690.



5-bromo-1, 3, 3-trimethylindolin-2-one (2.70f): The compound **2.70f** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 79% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.35 (dd, *J* = 8.2, 2.0 Hz, 1H), 7.29 (d, *J* = 1.9 Hz, 1H), 6.71 (d, *J* = 8.2 Hz, 1H), 3.17 (s, 3H), 1.34 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 180.5, 141.6, 137.8, 130.4, 125.6, 115.1, 109.4, 77.4, 77.1, 76.8, 44.3, 26.2, 24.2. HRMS: m/z calcd for C₁₁H₁₂BrNO: 254.0181; found: 254.0180.



1, **3**, **3-trimethyl-5-(trifluoromethyl) indolin-2-one (2.70g):** The compound **2.70g** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 89% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl3): δ 7.53 (ddd, J = 8.2, 1.7, 0.8 Hz, 1H), 7.43 – 7.32 (m, 1H), 6.91 (d, J = 8.2 Hz, 1H), 3.24 (s, 3H), 1.38 (s, 6H). ¹³C NMR (100 MHz, CDCl3): δ 181.1, 145.6, 136.2, 125.4 (q, J = 4 Hz), 124.6 (q, J = 32.4 Hz), 124.4 (q, J = 270 Hz), 119.2 (q, J = 3.7 Hz) 107.6, 77.4, 77.0, 76.7, 44.1, 26.3, 24.1. ¹⁹F NMR (376 MHz, CDCl3) δ -61.3. HRMS: m/z calcd for C₁₂H₁₂F₃NO: 244.0949; found: 244.0953.



1,3,3-trimethyl-5-nitroindolin-2-one (2.70h): The compound **2.70h** was prepared according to the general method described above and purified by flash column chromatography (0 to 6% ethyl acetate in hexanes) in 91% yield as a yellow solid. ¹H NMR (400 MHz, CDCl₃): δ 8.21 (dd, *J* = 8.6, 2.3 Hz, 1H), 8.07 (d, *J* = 2.2 Hz, 1H), 6.93 (d, *J* = 8.6 Hz, 1H), 3.27 (s, 3H), 1.40 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 181.2, 148.4, 143.4, 136.4, 125.1, 118.2, 107.6, 77.4, 77.1, 76.8, 44.1, 26.6, 24.0. HRMS: m/z calcd for C₁₁H₁₂N₂O₃: 221.0926; found: 221.0929.



5-acetyl-1, 3, 3-trimethylindolin-2-one (2.70i): The compound 2.70i was prepared according to the general method described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 83% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.83 (dd, J = 8.2, 1.7 Hz, 1H), 7.75 (d, J = 1.5 Hz, 1H), 6.82 (d, J = 8.2 Hz, 1H), 3.15 (s, 3H), 2.47 (s, 3H), 1.28 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 196.7, 181.4, 146.9, 135.8, 132.0, 129.8, 122.0, 107.4, 77.5, 77.2, 76.9, 43.8, 26.3, 26.3, 24.1. HRMS: m/z calcd for C₁₃H₁₅NO₂: 218.1181; found: 218.1190.



Ethyl 1, 3, 3-trimethyl-2-oxoindoline-5-carboxylate (2.70j): The compound 2.70j was prepared according to the general method described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 84% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.95 (dd, J = 8.2, 1.7 Hz, 1H), 7.82 (d, J = 1.4 Hz, 1H), 6.82 (d, J = 8.2 Hz, 1H), 4.30 (q, J = 7.1 Hz, 2H), 3.18 (s, 3H), 1.37 – 1.29 (m, 9H). ¹³C NMR (100 MHz, CDCl₃): δ 181.4, 166.3, 146.6, 135.5, 130.3, 124.6, 123.4, 107.4, 77.5, 77.1, 76.8, 60.7, 43.9, 26.3, 24.1, 14.3. HRMS: m/z calcd for C₁₄H₁₇NO₃: 248.1287; found: 248.1292.



7-fluoro-1,3,3-trimethylindolin-2-one (**2.70k**): The compound **2.70k** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 73% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 6.97 – 6.90 (m, 3H), 3.39 (d, J = 2.7 Hz, 3H), 1.34 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 180.7, 147.7 (d, J = 243.2 Hz), 138.7 (d, J = 3.1 Hz), 129.1 (d, J = 8.0 Hz), 122.9 (d, J = 6.3 Hz), 118.0 (d, J = 3.2 Hz), 115.5 (d, J = 19.3 Hz). 77.4, 77.1, 76.8, 44.4, 28.5, 24.4. ¹⁹F NMR (376 MHz, CDCl₃): δ -136.9. HRMS: m/z calcd for C₁₁H₁₂FNO: 194.0981; found: 194.0978.



6-fluoro-1,3,3-trimethylindolin-2-one + 4-fluoro-1,3,3-trimethylindolin-2-one (2.70) and 2.70l'): The product mixture of 2.70l/2.70l' was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 84% yield as a white solid. The product ratio was determined by ¹H NMR. ¹H NMR (400 MHz, CDCl₃): δ 7.21 – 7.16 (m, 0.09H) (2.70l³) 7.08 (dd, *J* = 8.1, 5.3 Hz, 0.91H) (2.70l), 6.70 – 6.61 (m, 1.09H), (2.70I + 2.70l³) 6.55 (dd, *J* = 8.9, 2.3 Hz, 0.90H) (2.70l), 3.16 (d, *J* = 6.1 Hz, 3H) (2.70I+2.70l³), 1.42 (s, 0.53H) (2.70l), 1.31 (s, 5.49H) (2.70l). ¹³C NMR (100 MHz, CDCl₃) (2.70l³): δ 181.4 , 162.7 (d, *J* = 243.8 Hz), 144.0 (d, *J* = 11.5 Hz), 131.0 (d, *J* = 2.9 Hz), 123.0 (d, *J* = 9.7 Hz), 108.2 (d, *J* = 22.3 Hz), 96.8 (d, *J* = 27.4 Hz). 43.7, 26.2, 24.3. ¹³C NMR (100 MHz, CDCl₃)) (2.70l): δ 180.5, 158.9 (d, *J* = 246.9 Hz), 136.7 (d, *J* = 54.8 Hz), 129.2 (d, *J* = 8.6 Hz), 120.6 (d, *J* = 19.7 Hz), 110.0 (d, *J* = 21.0 Hz), 104.1 (d, *J* = 3.0 Hz), 44.0, 26.5, 22.7 . ¹⁹F NMR (376 MHz, CDCl₃): δ -113.2, -121.9. HRMS: m/z calcd for C₁₁H₁₂FNO: 194.0981; found: 194.0986.



3,3-dimethyl-1-phenylindolin-2-one (**2.70m**): The compound **2.70m** was prepared according to the general method described above and purified by flash column chromatography

(0 to 2% ethyl acetate in hexanes) in 90% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.62 – 7.09 (m, 8H), 6.91 (d, *J* = 7.8 Hz, 1H), 1.56 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 180.6, 142.5, 135.6, 134.7, 129.5, 127.9, 127.6, 126.5, 123.0, 122.7, 109.4, 77.6, 77.3, 77.0, 44.3, 24.8. HRMS: m/z calcd for C₁₆H₁₃NONa: 260.1051; found: 260.1058.



1-benzyl-3,3-dimethylindolin-2-one (2.70n): The compound 2.70n was prepared according to the general method described above and purified by flash column chromatography (0 to 2% ethyl acetate in hexanes) in 84% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.37 – 7.23 (m, 6H), 7.17 (td, J = 7.7, 1.3 Hz, 1H), 7.07 (td, J = 7.6, 1.0 Hz, 1H), 6.78 (d, J = 7.7Hz, 1H), 4.97 (s, 2H), 1.50 (s, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 181.4, 141.7, 136.2, 135.8, 128.8, 127.6, 127.5, 127.2, 122.5, 122.3, 109.1, 77.6, 77.2, 76.9, 44.2, 43.5, 24.6. HRMS: m/z calcd for C₁₇H₁₇NONa: 274.1208; found: 274.1221.



3-ethyl-1,3-dimethylindolin-2-one (2.70o): The compound **2.70o** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 59% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃): δ 7.26 (t, *J* = 7.6 Hz, 1H), 7.17 (d, *J* = 7.1 Hz, 1H), 7.07 (t, *J* = 7.3 Hz, 1H), 6.84 (d, *J* = 7.6 Hz, 1H), 3.22 (s, 3H), 1.93 (dd, *J* = 13.6, 7.1 Hz, 1H), 1.77 (dd, *J* = 13.6, 7.2 Hz, 1H), 1.35 (s, 3H),

0.59 (t, J = 7.3 Hz, 3H). ¹³C NMR (100 MHz, CDCl₃): δ 180.6, 143.4, 133.9, 127.6, 122.4, 122.3, 107.8, 77.4, 77.1, 76.8, 48.9, 31.4, 26.0, 23.3, 8.8. HRMS: m/z calcd for C₁₂H₁₅NO: 190.1232; found: 190.1234.



(1,3-dimethyl-2-oxoindolin-3-yl) methyl acetate (2.70p): The compound 2.70p was prepared according to the general method described above and purified by flash column chromatography (0 to 6% ethyl acetate in hexanes) in 74% yield as a light yellow oil. ¹H NMR (400 MHz, CDCl₃): δ 7.28 – 7.20 (m, 2H), 7.03 (t, *J* = 7.4 Hz, 1H), 6.84 (d, *J* = 7.8 Hz, 1H), 4.46 (d, *J* = 10.8 Hz, 1H), 4.14 (d, *J* = 10.8 Hz, 1H), 3.21 (s, 3H), 1.83 (s, 3H), 1.36 (s, 3H). ¹³C NMR (100 MHz, CDCl₃): δ 178.1, 170.3, 143.3, 131.2, 128.3, 123.0, 122.5, 108.1, 77.5, 77.1, 76.8, 67.4, 47.9, 26.2, 20.5, 19.6. HRMS: m/z calcd for C₁₃H₁₃NO₃Na: 256.0950; found: 256.0962.



1,3-dimethyl-3-(trifluoromethyl) indolin-2-one (2.70q): The compound **2.70q** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 61% yield as a light yellow oil. ¹H NMR (400 MHz, CDCl3): δ 7.37 – 7.36 (m, 2H), 7.11 (t, *J* = 7.3 Hz, 1H), 6.89 (d, *J* = 7.7 Hz, 1H), 3.22 (s, 3H), 1.63 (s, 3H). ¹³C NMR (100 MHz, CDCl₃): δ 172.2, 143.6, 129.9, 126.1, δ 125.0 (q, *J* = 281.8 Hz), 124.4, 123.1, 108.6, 77.5, 77.2, 76.8, δ 52.0 (q, *J* = 27.6 Hz), 26.4, 17.6.

¹⁹F NMR (376 MHz, CDCl₃): δ -73.6. HRMS: m/z calcd for C₁₁H₁₀F₃NO: 230.0793; found: 230.0803



3-butyl-1,3-dimethyl-5-nitroindolin-2-one (2.70r): The compound **2.70r** was prepared according to the general method described above and purified by flash column chromatography (0 to 6% ethyl acetate in hexanes) in 82% yield as a yellow oil. ¹H NMR (400 MHz, CDCl₃): δ 8.21 (dd, *J* = 8.6, 2.3 Hz, 1H), 8.02 (d, *J* = 2.3 Hz, 1H), 6.92 (d, *J* = 8.6 Hz, 1H), 3.26 (s, 3H), 1.90 (td, *J* = 12.9, 12.4, 4.7 Hz, 1H), 1.76 (td, *J* = 13.4, 12.9, 4.5 Hz, 1H), 1.36 (s, 3H), 1.15 (ddd, *J* = 14.8, 7.6, 3.0 Hz, 2H), 0.89 (ddd, *J* = 10.7, 7.5, 5.4 Hz, 1H), 0.81 – 0.69 (m, 4H). ¹³C NMR (100 MHz, CDCl₃): δ 180.7, 149.0, 143.3, 135.0, 125.1, 118.2, 107.4, 77.4, 77.1, 76.8, 48.4, 38.0, 26.4, 26.4, 23.4, 22.6, 13.7. HRMS: m/z calcd for C₁₄H₁₈N₂O₃: 263.1396; found: 263.1408.



3-(methoxymethyl)-1, 3-dimethyl-5-nitroindolin-2-one (2.70s): The compound 2.70s was prepared according to the general method described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 78% yield as a yellow oil. ¹H NMR (400 MHz, CDCl₃): δ 8.21 (dd, *J* = 8.6, 2.3 Hz, 1H), 8.11 (d, *J* = 2.3 Hz, 1H), 6.92 (d, *J* = 8.6 Hz, 1H), 3.72 – 3.57 (m, 2H), 3.25 (s, 3H), 3.18(s, 3H), 1.34 (s, 3H). ¹³C NMR (100 MHz, CDCl₃): 179.1,

149.3, 143.3, 133.5, 125.4, 118.7, 107.5, 77.5, 77.2, 76.8, 76.5, 59.3, 49.3, 26.6, 19.3. HRMS: m/z calcd for C₁₂H₁₄N₂O₄: 251.1032; found: 251.1036.



3-benzyl-1, 3-dimethyl-5-nitroindolin-2-one (2.70t): The compound **2.70t** was prepared according to the general method described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 69% yield as a yellow solid. ¹H NMR (400 MHz, CDCl₃): δ 8.14 (dd, *J* = 8.6, 2.3 Hz, 1H), 8.06 (d, *J* = 2.2 Hz, 1H), 7.07 – 7.01 (m, 3H), 6.81 (dd, *J* = 7.5, 1.7 Hz, 2H), 6.67 (d, *J* = 8.6 Hz, 1H), 3.20 (d, *J* = 13.1 Hz, 1H), 3.05 (d, *J* = 15.7 Hz, 4H), 1.54 (s, 3H). ¹³C NMR (100 MHz, CDCl₃): δ 179.9, 148.8, 143.0, 135.1, 133.8, 129.5, 127.8, 126.9, 125.2, 119.0, 107.3, 77.4, 77.1, 76.8, 50.1, 44.5, 26.2, 22.4. HRMS: m/z calcd for C₁₇H₁₆N₂O₃: 297.1239; found: 297.1238.



1-methylspiro(cyclopentane-1')-indolin-2-one (2.70u): The compound **2.70u** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 38% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃): δ 7.28 – 7.23 (m, 2H), 7.08 – 7.04 (m, 1H), 6.84 (d, *J* = 7.7 Hz, 1H), 3.23 (s, 3H), 2.20 – 1.86 (m, 6H), 1.85 (ddd, *J* = 12.1, 7.0, 4.2 Hz, 2H). ¹³C NMR (100 MHz, CDCl₃): δ 181.9, 142.9, 136.8, 127.3, 122.5, 122.2, 107.6, 77.4, 77.0, 76.7, 53.9, 38.3, 26.6, 26.2. HRMS: m/z calcd for C₁₁H₁₃NO: 202.1232; found: 202.1242.



5-methoxy-3,3-dimethyl-1-phenylindolin-2-one + 1-(4-methoxyphenyl)-3,3dimethylindolin-2-one (2.70w and 2.70w'): The product mixture of 2.70w/2.70w' was prepared according to the general method described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 79% yield as a white solid. The product ratio was determined by ¹H NMR. ¹H NMR (400 MHz, CDCl₃): δ 7.54 – 7.46 (m, 3.42H), 7.40 – 7.28 (m, 4.48H), 7.21 – 7.17 (m, 1.18H), 7.04 – 7.12 (m, 3.51H), 6.94 (d, *J* = 2.4 Hz, 0.82H) (H₁), 6.82 (d, *J* = 8.3 Hz, 2H) (H₃), 6.74 (dd, *J* = 8.6, 2.5 Hz, 0.82H) (H₂), 3.84 (s, 3.44H), 3.81 (s, 2.42H), 1.52 (s, 12.28H). ¹³C NMR (100 MHz, CDCl₃): δ 180.8, 180.3, 159.0, 156.4, 142.9, 137.0, 135.8, 135.57, 135.00, 129.50, 127.94, 127.64, 127.33, 126.29, 122.86, 122.59, 114.87, 111.85, 109.9, 109.8, 109.2, 77.6, 77.3, 77.0, 55.7, 55.5, 44.7, 44.2, 24.8, 24.7.



1-(4-fluorophenyl)-3,3-dimethylindolin-2-one + 5-fluoro-3,3-dimethyl-1-

phenylindolin-2-one (2.70x and 2.70x'): The product mixture of 2x/2x' was prepared according to the general method described above and purified by flash column chromatography (0 to 3%

ethyl acetate in hexanes) in 87% combined yield as a white solid. The product ratio was determined by ¹H NMR. ¹H NMR (400 MHz, CDCl₃): δ 7.55 – 7.41 (m, 6.52H), 7.31 (d, *J* = 7.2 Hz, 1H) (H₁), 7.24 – 7.19 (m, 2.73H), 7.13 (t, *J* = 7.4 Hz,0.931H), 7.05 (dd, *J* = 7.8, 2.3 Hz, 0.94H), 6.90 – 6.79 (m, 2.75H), 1.56 – 1.47 (d, 10.68H). ¹³C NMR (101 MHz, CDCl₃) δ 180.7, 180.2, 162.9, 160.7, 160.5, 158.3, 142.3, 138.3, 137.3 (d, *J* = 7.8 Hz), 135.5, 134.6, 130.6 (d, *J* = 3.1 Hz), 129.6, 128.4 (d, *J* = 8.6 Hz), 127.8 (d, *J* = 27.8 Hz), 126.4 , 122.9 (d, *J* = 40.7 Hz), 116.5 (d, *J* = 22.8 Hz), 113.8 (d, *J* = 23.4 Hz), 110.6 (d, *J* = 24.4 Hz), 110.0 (d, *J* = 8.0 Hz), 109.2, 44.7, 44.3, 24.7 (d, *J* = 10.6 Hz).



N-(4-bromophenyl)-N-methyl-2-methylenebutanamide (2.68y): The compound **2.68y** was prepared according to the general method described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 79% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.44 (d, *J* = 8 Hz 2H), 7.01 (d, *J* = 8 Hz, 2H), 5.01 (d, *J* = 30.9 Hz, 2H), 3.31 (s, 3H), 2.12 (q, *J* = 7.4 Hz, 2H), 0.97 (t, *J* = 7.4 Hz, 3H).¹³C NMR (100 MHz, CDCl₃): δ 171.6, 146.1, 143.6, 132.3, 128.2, 120.3, 117.3, 77.4, 77.1, 76.8, 37.6, 26.5, 11.7.



5-bromo-3-ethyl-1,3-dimethylindolin-2-one (2.70y): The compound 2.70y was

prepared according to the general method described above and purified by flash column chromatography (0 to 4% ethyl acetate in hexanes) in 71% yield as a colorless oil. ¹H NMR (400

MHz, CDCl₃): δ 7.35 (dd, *J* = 8.2, 1.9 Hz, 1H), 7.25 (d, *J* = 1.9 Hz, 1H), 6.70 (d, *J* = 8.2 Hz, 1H), 3.17 (s, 3H), 1.90 (dq, *J* = 14.8, 7.4 Hz, 1H), 1.72 (dq, *J* = 14.8, 7.4 Hz, 1H), 1.31 (s, 3H), 0.56 (t, *J* = 7.4 Hz, 3H). ¹³C NMR (100 MHz, CDCl₃): δ 179.94, 142.53, 136.04, 130.44, 125.78, 115.11, 109.28, 77.47, 77.15, 76.83, 49.18, 31.39, 26.14, 23.23, 8.82.



3-ethyl-5-(4-fluorophenyl)-1,3-dimethylindolin-2-one (2.75): The compound **2.75** was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 78% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃): δ 7.50 – 7.34 (m, 4H), 7.08 (t, *J* = 7.6 Hz, 2H), 6.87 (d, *J* = 7.6 Hz, 1H), 3.21 (s, 3H), 2.06 – 31.70 (m, 2H), 1.38 (s, 3H), 0.62 (t, *J* = 6.3 Hz, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 180.5, 163.4, 160.9, 142.9, 137.3 (d, *J* = 3.1 Hz), 134.7 (d, *J* = 21.1 Hz), 128.3 (d, *J* = 7.9 Hz), 126.4, 121.2, 115.6 (d, *J* = 21.4 Hz), 108.1, 49.0, 31.5, 26.0, 23.3, 8.9. ¹⁹F NMR (376 MHz, CDCl₃): δ -116.0.



N-methyl-N-phenyl-2-(*n*-butyl) Acrylamide (2.68r): The compound 2.68r was prepared according to the general method described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 65% yield as a yellow solid. ¹H NMR (400 MHz, CDCl₃): δ 8.22 – 8.15 (m, 2H), 7.36 – 7.29 (m, 2H), 5.20 – 5.11 (m, 1H), 5.08 – 5.00 (m, 1H), 3.40 (s, 3H), 2.19 – 2.08 (m, 2H), 1.44 – 1.35 (m, 2H), 1.28 (dq, *J* = 14.3, 7.1 Hz, 2H), 0.86 (t, *J* = 7.3 Hz, 3H). ¹³C NMR (100 MHz, CDCl₃): δ 171.6, 150.2, 145.3, 144.6, 126.3, 124.5, 119.1, 77.4, 77.1, 76.8, 37.5, 33.2, 29.6, 22.2, 13.8. HRMS: m/z calcd for C₁₄H₁₈N₂O₃Na: 285.1215; found: 285.1227.



N-methyl-N-phenyl-2-(methoxy methyl) Acrylamide (2.68s): The compound **2.68s** was prepared according to the general method described above and purified by flash column chromatography (0 to 10% ethyl acetate in hexanes) in 71% yield as a yellow solid. ¹H NMR (400 MHz, CDCl₃): δ 8.23 – 8.11 (m, 2H), 7.47 – 7.36 (m, 2H), 5.31 (s, 1H), 4.98 (s, 1H), 4.10 – 4.00 (m, 2H), 3.46 – 3.29 (m, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 169.62, 150.24, 145.44, 141.08, 126.72, 124.50, 120.82, 77.48, 77.16, 76.84, 73.23, 58.84, 37.38. HRMS: m/z calcd for C₁₂H₁₄N₂O₄Na: 273.0851; found: 273.0863.



N-methyl-N-phenyl-2-(benzyl) Acrylamide (2.68t): The compound 2.68t was prepared according to the general method described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 63% yield as a yellow solid. ¹H NMR (400 MHz, CDCl₃): 8.11 - 8.01 (m, 2H), 7.39 - 7.24 (m, 3H), 7.22 - 7.12 (m, 2H), 6.92 - 6.82 (m, 2H), 5.20 - 5.14 (m, 1H), 4.98 - 4.90 (m, 1H), 3.60 (s, 2H), 3.32 (s, 3H). ¹³C NMR (100 MHz, CDCl₃): δ 170.6, 150.1, 145.5, 143.6, 137.4, 129.3, 128.7, 126.9, 126.5, 124.4, 120.2, 77.4, 77.1, 76.8, 40.5, 37.6. HRMS: m/z calcd for C₁₇H₁₆N₂O₃Na: 319.1059; found: 319.1066.

2.8. References

- Poulsen, T. B.; Jørgensen, K. A. Catalytic Asymmetric Friedel–Crafts Alkylation Reactions Copper Showed the Way. *Chem. Rev.* 2008, *108*, 2903–2915.
- a) Trost, B. The Atom Economy-a Search for Synthetic Efficiency. *Science*. 1991, 254, 1471–1477. (b) Newhouse, T.; Baran, P. S.; Hoffmann, R. W. The Economies of Synthesis. *Chem. Soc. Rev.* 2009, *38*, 3010–3021. (c) Ren, T.; Patel, M.; Blok, K. Olefins from Conventional and Heavy Feedstocks: Energy Use in Steam Cracking and Alternative Processes. *Energy*. 2006, *31*, 425–451.
- Sevov, C. S.; Hartwig, J. F. Iridium-Catalyzed Intermolecular Asymmetric Hydroheteroarylation of Bicycloalkenes. J. Am. Chem. Soc. 2013, 135, 2116-2120.
- Murai, S.; Kakiuchi, F.; Sekine, S.; Tanaka, Y.; Kamatani, A.; Sonoda, M.; Chatani, N. Efficient Catalytic Addition of Aromatic Carbon-Hydrogen Bonds to Olefins. *Nature* 1993, *366*, 529-531.
- a) Dong, Z.; Ren, Z.; Thompson, S. J.; Xu, Y.; Dong, G.; *Chem. Rev.* 2017, 9333-9403.
 b) J.-Q. Yu, Z. Shi, Editors, *C-H Activation. [In: Top. Curr. Chem., 2010; 292]*, 2010.
- a) Jun, C.-H.; Hong, J.-B.; Kim, Y.-H.; Chung, K.-Y. The Catalytic Alkylation of Aromatic Imines by Wilkinson's Complex: The Domino Reaction of Hydroacylation and *ortho*-Alkylation. *Angew. Chem., Int. Ed.* 2000, *39*, 3440–3442. (b) Jun, C.-H.; Moon, C.
 W.; Hong, J.-B.; Lim, S.-G.; Chung, K.-Y.; Kim, Y.-H. Chelation-Assisted Rh(I)-Catalyzed *ortho*-Alkylation of Aromatic Ketimines or Ketones with Olefins. *Chem. - Eur. J.* 2002, *8*, 485–492.
- Lenges, C. P.; Brookhart, M. Addition of Olefins to Aromatic Ketones Catalyzed by Rh(I) Olefin Complexes. *J. Am. Chem. Soc.* 1999, *121*, 6616–6623.

81

- Lim, Y.-G.; Kim, Y. H.; Kang, J.-B. Rhodium-Catalysed Regioselective Alkylation of the Phenyl Ring of 2-Phenylpyridines with Olefins. J. *Chem. Soc., Chem. Commun.* 1994, 2267–2268.
- Carrion, M. C.; Cole-Hamilton, D. J. Halide-Free Ethylation of Phenol by Multifunctional Catalysis Using Phosphinite Ligands. *Chem. Commun.* 2006, 4527–4529.
- Nakao, Y.; Kashihara, N.; Kanyiva, K. S.; Hiyama, T. Nickel-Catalyzed Alkenylation and Alkylation of Fluoroarenes via Activation of C–H Bond over C–F Bond. *J. Am. Chem. Soc.* 2008, *130*, 16170-16171.
- 11. Sun, Z. M.; Zhang, J.; Manan, R. S.; Zhao, P. Rh(I)-Catalyzed Olefin Hydroarylation with Electron-Deficient Perfluoroarenes. *J. Am. Chem. Soc.* **2010**, *132*, 6935–6937.
- Ryu, J.; Hwan Cho, S. H.; Chang, S.; A Versatile Rhodium(I) Catalyst System for the Addition of Heteroarenes to both Alkenes and Alkynes by a C-H Bond Activation.
 Angew. Chem. Int. Ed. 2012, *51*, 3677 –3681.
- Tsuchikama, K.; Kasagawa, M.; Hashimoto, Y.-K.; Endo, K.; Shibata, T. Cationic Iridium-BINAP Complex-Catalyzed Addition of Aryl Ketones to Alkynes and Alkenes via Directed C-H Bond Cleavage. J. Organomet. Chem. 2008, 693, 3939–3942.
- Pan, S.; Ryu, N.; Shibata, T. Ir(I)-Catalyzed C-H Bond Alkylation of C₂-Position of Indole with Alkenes: Selective Synthesis of Linear or Branched 2-Alkylindoles. *J. Am. Chem. Soc.* 2012, *134*, 17474–17477.
- a) Crisenza, G. E. M.; McCreanor, N. G.; Bower, J. F. Branch Selective, Iridium-Catalyzed Hydroarylation of Monosubstituted Alkenes via a Cooperative Destabilization Strategy. *J. Am. Chem. Soc.* 2014, *136*, 10258–10261. b) Crisenza, G. E. M.; Sokolova,

O. O.; Bower, J. F. BranchSelective Alkene Hydroarylation by Cooperative
Destabilization Iridium-Catalyzed *ortho*-Alkylation of Acetanilides. *Angew. Chem., Int. Ed.* 2015, *54*, 14866–14870. c) Pan, S.; Ryu, N.; Shibata, T. Iridium(I)-Catalyzed Direct
C-H Bond Alkylation of the C-7 Position of Indolines with Alkenes. *Adv. Synth. Catal.*2014, *356*, 929–933. d) Ebe, Y.; Nishimura, T. Iridium-Catalyzed Branch-Selective
Hydroarylation of Vinyl Ethers via C-H Bond Activation.J. Am. Chem. Soc.2015, 137, 5899–5902.

- Sevov, C. S.; Hartwig, J. F. Iridium-Catalyzed Intermolecular Asymmetric
 Hydroheteroarylation of Bicycloalkenes. *J. Am. Chem. Soc.* 2013, *135*, 2116–2119.
- 17. Kim, J.; Park, S.-W.; Baik, M.-H.; Chang, S. Complete Switch of Selectivity in the C-H
 Alkenylation and Hydroarylation Catalyzed by Iridium: The Role of Directing Groups. *J. Am. Chem. Soc.* 2015, *137*, 13448–13451.
- Gao, K.; Yoshikai, N.; Low-Valent Cobalt Catalysis: New Opportunities for C–H Functionalization. *Acc. Chem. Res.* 2014, 47, 1208–1219.
- Gao, K.; Yoshikai, N. Cobalt–Phenanthroline Catalysts for the *ortho* Alkylation of Aromatic Imines under Mild Reaction Conditions. *Angew. Chem., Int. Ed.* 2011, *50*, 6888–6892.
- Yoshino, T.; Ikemoto, H.; Matsunaga,S.; Kanai, M. A Cationic High-Valent Cp*Co(III) Complex for the Catalytic Generation of Nucleophilic Organometallic Species: Directed C-H Bond Activation. *Angew. Chem. Int. Ed.* 2013, *52*, 2207 –2211.
- Li, J.; Zhang, Z.; Ma, W.; Tang, M.; Wang, D.; Zou, L-H.; Mild Cobalt(III)-Catalyzed C– H Hydroarylation of Conjugated C = C/C = OBonds. *Adv. Synth. Catal.* 2017, *359*,1717 – 1724.

- Bair, J. S.; Schramm, Y.; Sergeev, A. G.; Clot, E.; Eisenstein, O.; Hartwig, J. F. Linear-Selective Hydroarylation of Unactivated Terminal and Internal Olefins with Trifluoromethyl-Substituted Arenes. *J. Am. Chem. Soc.* 2014, *136*, 13098–13101.
- Schramm, Y.; Takeuchi, M.; Semba, K.; Nakao, Y.; Hartwig, J. F. Anti-Markovnikov Hydroheteroarylation of Unactivated Alkenes with Indoles, Pyrroles, Benzofurans, and Furans Catalyzed by a Nickel-N Heterocyclic Carbene System. *J. Am. Chem. Soc.* 2015, *137*, 12215–12218.
- a) Youn, S. W.; Pastine, S. J.; Sames, D. Ru(III)-Catalyzed Cyclization of Arene-Alkene Substrates via Intramolecular Electrophilic Hydroarylation. *Org. Lett.* 2004, *6*, 581-584;
 b) Ueda, S.; Okada, T.; Nagasawa, H. Oxindole synthesis by palladium-catalysed aromatic C–H alkenylation. *Chem. Commun.* 2010, 2462-2464. c) Xie, K.; Wang, S.; Li, P.; Li, X.; Yang, Z.; An, X.; Guo, C.-C.; Tan, Z. Synthesis of tetralin and chromane derivatives via In-catalyzed intramolecular hydroarylation. *Tetrahedron Lett.* 2010, *51*, 4466-4469. (d) Jean, M.; van de Weghe, P. Gold-catalyzed intramolecular hydroarylation of olefins. Scope evaluation and preliminary mechanistic studies. *Tetrahedron Lett.* 2011, *52*, 3509-3513. (e) Mu, X.; Wu, T.; Wang, H.-y.; Guo, Y.-I.; Liu, G. Palladium-Catalyzed Oxidative Aryltrifluoromethylation of Activated Alkenes at Room Temperature. *J. Am. Chem. Soc.* 2012, *134*, 878-881. (f) Shigehisa, H.; Ano, T.; Honma, H.; Ebisawa, K.; Hiroya, K. Co-Catalyzed Hydroarylation of Unactivated Olefins. *Org. Lett.* 2016, *18*, 3622-3625.
- a) Thalji, R. K.; Ahrendt, K. A.; Bergman, R. G.; Ellman, J. A. Annulation of Aromatic Imines via Directed C–H Activation with Wilkinson's Catalyst. *J. Am. Chem. Soc.* 2001, *123*, 9692-9693. (b) Ahrendt, K. A.; Bergman, R. G.; Ellman, J. A. Synthesis of a

Tricyclic Mescaline Analog by Catalytic C-H Bond Activation . Org. Lett. 2003, 5, 1301-1303. (c) Thalji, R. K.; Ellman, J. A.; Bergman, R. G. Highly Efficient and Enantioselective Cyclization of Aromatic Imines via Directed C-H Bond Activation. J. Am. Chem. Soc. 2004, 126, 7192-7193. (d) Wilson, R. M.; Thalji, R. K.; Bergman, R. G.; Ellman, J. A. Enantioselective Synthesis of a PKC Inhibitor via Catalytic C-H Bond Activation. Org. Lett. 2006, 8, 1745-1747. (e) Davis, T. A.; Hyster, T. K.; Rovis, T. Rh(III)-Catalyzed Intramolecular Hydroarylation, Amidoarylation, and Heck-type Reaction: Three Distinct Pathways Determined by Amide Directing Group. Angew. Chem. Int. Ed. 2013, 52, 14181-14185. (f) Ding, Z.; Yoshikai, N. Cobalt-Catalyzed Intramolecular Olefin Hydroarylation Leading to Dihydropyrroloindoles and Tetrahydropyridoindoles. Angew. Chem. Int. Ed. 2013, 52, 8574-8578. (g) Shi, Z.; Boultadakis-Arapinis, M.; Koester, D. C.; Glorius, F. Rh(III) - Catalyzed intramolecular redox-neutral cyclization of alkenes via C-H activation. Chem. Commun. 2014, 50, 2650-2652. (h) Ye, B.; Do-nets, P. A.; Cramer, N. Chiral Cp-Rhodium(III)-Catalyzed Asymmetric Hydroarylations of 1,1-Disubstituted Alkenes. Angew. Chem. Int. Ed. 2014, 53, 507-511. (i) Ghosh, K.; Rit, R. K.; Ramesh, E.; Sahoo, A. K. Ruthenium-Catalyzed Hydroarylation and One-Pot Twofold Unsymmetrical C-H Functionalization of Arenes. Angew. Chem. Int. Ed. 2016, 55, 7821-7825.

- Jia, C.; Piao, D.; Oyamada, J.; Lu, W.; Kitamura, T.; Fuji-wara, Y. Efficient Activation of Aromatic C-H Bonds for Addition to C-C Multiple Bonds. *Science*. 2000, 287, 1992-1995.
- a) Lewis, L. N.; Smith, J. F. Catalytic carbon-carbon bond formation via *ortho*-metalated complexes. *J. Am. Chem. Soc.* 1986, *108*, 2728-2735. (b) Murai, S.; Kakiuchi, F.; Sekine,

S.; Tanaka, Y.; Kamatani, A.; Sonoda, M.; Chatani, N. Efficient catalytic addition of aromatic carbon-hydrogen bonds to olefins. *Nature*. **1993**, *366*, 529-531.

- 28. Albrecht, M. Cyclometalation Using d-Block Transition Metals: Fundamental Aspects and Recent Trends. *Chem. Rev.* **2010**, *110*, 576-623.
- S. W. Youn, S. J. Pastine, D. Sames. Ru(III)-Catalyzed Cyclization of Arene-Alkene Substrates via Intramolecular Electrophilic Hydroarylation. *Org. Lett.* 2004, *6*, 581-584.
- Shigehisa, H.; Ano, T.; Honma, H.; Ebisawa, K.; Hiroya, K. Co-Catalyzed Hydroarylation of Unactivated Olefins. *Org. Lett.* 2016, *18*, 3622–3625.
- X. Mu, T. Wu, H.-y. Wang, Y.-l. Guo, G. Liu. Palladium-Catalyzed Oxidative Aryltrifluoromethylation of Activated Alkenes at Room Temperature *J. Am. Chem. Soc.* 2012, *134*, 878-881.
- Ding, Z.; Yoshikai, N. Cobalt-Catalyzed Intramolecular Olefin Hydroarylation Leading to Dihydropyrroloindoles and Tetrahydropyridoindoles. *Angew. Chem. Int. Ed.* 2013, 52, 8574–8578
- 33. Santos, M. M. M. Recent Advances in the Synthesis of Biologically active Spirooxindoles. *Tetrahedron* 2014, 70, 9735-9757. (b) Singh, G. S.; Desta, Z. Y. Isatins As Privileged Molecules in Design and Synthesis of Spiro-Fused Cyclic Frameworks. *Chem. Rev.* 2012, 112, 6104-6155.
- 34. Chen, J.-R.; Yu, X.-Y.; Xiao, W.-J. Tandem Radical Cyclization of *N*-Arylacrylamides: An Emerging Platform for the Construction of 3,3-Disubstituted Oxindoles. *Synthesis*.
 2015, 47, 604-629.
- 35. (a) Jaegli, S.; Dufour, J.; Wei, H.-l.; Piou, T.; Duan, X.-H.; Vors, J.-P.; Neuville, L.; Zhu,J. Palladium-Catalyzed Carbo-Heterofunctionalization of Alkenes for the Synthesis of
Oxindoles and Spirooxindoles. *Org. Lett.* 2010, *12*, 4498-4501. (b) Ueda, S.; Okada, T.;
Nagasawa, H. Oxindole synthesis by palladium-catalysed aromatic C–H alkenylation. *Chem. Commun.* 2010, 2462-2464. (c) Schiffner, J. A.; Oestreich, M. All-CarbonSubstituted Quaternary Carbon Atoms in Oxindoles by an Aerobic Palladium(II)Catalyzed Ring Closure onto Tri- and Tetrasubstituted Double Bonds. *Eur. J. Org. Chem.*2011, 1148-1154. (d) Wu, T.; Mu, X.; Liu, G. Palladium-Catalyzed Oxidative
Arylalkylation of Activated Alkenes: Dual C-H Bond Cleavage of an Arene and
Acetonitrile. *Angew. Chem. Int. Ed.* 2011, 50, 12578-12581.

- Zhang, J.; Ugrinov, A.; Zhao, P. Ruthenium(II)/N-Heterocyclic Carbene Catalyzed [3+2]
 Carbocyclization with Aromatic N-H Ketimines and Internal Alkynes. *Angew. Chem. Int. Ed.* 2013, 52, 6681-6684.
- 37. See refs 3f and 4f for heterocycle synthesis via Co-catalyzed 6-endo-trig intramolecular alkene hydroarylation.
- Flegeau, E. F.; Bruneau, C.; Dixneuf, P. H.; Jutand, A. Autocatalysis for C–H Bond Activation by Ruthenium(II) Complexes in Catalytic Arylation of Functional Arenes. J. Am. Chem. Soc. 2011, 133, 10161-10170.
- 39. We did not detect formation of the corresponding 3-methylene oxindole, the possible by-product from intramolecular dehydrogenative Heck reaction (Fujiwara–Moritani reaction) as a competing process. It is possible that the lack of α-substitution led to a highly stabilized cyclometalation intermediate (A1 in Scheme 1c, with R1 = R2 =H), which failed to undergo amide ligand substitution by alkene. Thus, the steric crowding between N- and α-substituents (R/R1 in A1) might be a key factor that promotes desired hydroarylation reactivity.

- Lapointe, D.; Fagnou, K. Overview of the Mechanistic Work on the Concerted Metallation–Deprotonation Pathway. *Chem. Lett.* **2010**, 39, 1119-1126.
- Gui, Q.; Hu, L.; Chen, X.; Liu, J.; Tan, Z. Synthesis of Oxindoles via Iron-Mediated Hydrometallation-Cyclization of *N*-Aryl acrylamides. *Asian J. Org. Chem.* 2015, *4*, 870-874.
- Alluri, S.; Feng, H.; Livings, M.; Samp, L.; Biswas, D.; Lam, T. W.; Lobkovsky, E.;
 Ganguly, A. K. *Tetrahedron Lett.* 2011, *52*, 3945-3948.
- 43. Kuang, Y-Y.; Chen, F-E. a Convenient and Efficient Asymmetric Synthesis Of (S)-A-Arylthiomethyl-A-Hydroxybutyric Acid Esters. *Org. Prep. Proc. Int.* **2005**, *37*, 184-188.
- Pinto, A.; Jia, Y.; Neuville, L.; Zhu, J. Palladium-Catalyzed Enantioselective Domino Heck–Cyanation Sequence: Development and Application to the Total Synthesis of Esermethole and Physostigmine. *Chem. Eur. J.* 2007, *13*, 961-967.
- 45. Raghunathan, R.; Kumarasamy, E.; Akila, I.; Ugrinov, A.; Sivaguru, J.
 Intramolecular Paternò–Büchi reaction of atropisomeric α-oxoamides in solution and in the solid-state. *Chem. Commun.* 2013, 49, 8713-8715.
- 46. a) Ayitou, A. J.; J. Sivaguru; Light-Induced Transfer of Molecular Chirality in Solution: Enantiospecific Photocyclization of Molecularly Chiral Acrylanilides. *J. Am. Chem. Soc.* 2009, *131*, 5036-5037. b) Mu, X.; Wu, T.; Wang, H.; Guo, Y.; Liu, G. Palladium-Catalyzed Oxidative Aryltrifluoromethylation of Activated Alkenes at Room Temperature. *J. Am. Chem. Soc.* 2012, *134*, 878-881.

3. REGIOSELECTIVE ALKENYLATION OF HETEROARENES THROUGH RUTHENIUM CATALYSED DECARBOXYLATIVE C-H BOND ACTIVATION 3.1. Introduction

Alkenyl (hetero)arenes are widely present in many useful organic compounds such as fluorescent dyes, natural products, and pharmaceuticals. Hydro(hetro)arylation of alkynes emerged as a valuable tool for the synthesis of alkenyl (Hetero)arenes.¹ Classical method for the Hydro(hetro)arylation is Lewis acid catalyzed Friedel-Crafts reaction (Scheme 3.1.). This reaction is suitable only for arenes with electron donating substituents, and it suffers from achieving regio and stereoselectivity.²



Scheme 3.1. Lewis acid mediated hydroarylation with electron-rich arenes.

Transition metal catalysts have been employed for alkyne hydroarylation (hydro heteroarylation) using aryl halides (**3.05**) and organometallic reagents (**3.08**) as coupling partners (Scheme 3.2.). The main drawback of these reactions is the generation of stoichiometric amounts of salt waste and the need for pre-functionalized substrates.^[3]



Scheme 3.2. Transition metal catalyzed hydroarlation of alkynes using aryl halides or organometallic reagents.

On the other hand, direct addition of (hetero)arenes across the alkynes through transition metal catalyzed C-H bond activation occurs with high atom and step economy.

hydro(hetero)arylation of alkynes through transition metal catalyzed C-H bond functionalization represents the most straightforward method to access trisubstituted olefins which are an integral part of many biologically active compounds and used extensively as synthetic intermediates for fine chemicals and materials. Achieving regioselectivity is crucial for the practical application of transition metal catalyzed Hydro(hetero)arylation. The most reliable method for the regioselective C-H bond functionalization is the usage of directing group (Scheme 3.3.).⁴ Murai and co-workers reported first example for the Hydroarylation of alkenes using ketone as a directing group.⁵ Later, in 1995 they disclosed another method for Hydroarylation of alkynes using ruthenium catalyst.⁶ Following Murai's pioneering work, hydroarylation was extended to many directing groups such as pyridine, imine, amides, ester, etc.⁷⁻¹²



Scheme 3.3. Hydroaryation via transition metal mediated directed C-H bond activation.

Chelation assisted C-H bond activation also employed for the functionalization of heteroarenes. The group of Zeng developed a method for the ruthenium-catalyzed C-2

hydroindolation of alkynes with N-(2-pyridyl)-indoles (**3.16**), which provides access to 2-alkenyl substituted indoles (Scheme 3.4 a).¹³ This approach tolerates various indole derivatives and alkynes bearing electron donating and withdrawing substituents. Recently, Yoshikai and co-workers established another protocol for alkenylation of indole via imine directed as C-H bond activation using an earth-abundant iron catalyst (Scheme 3.4 b).¹⁴ This reaction proceeds through oxidative addition rather than concerted metalation and deprotonation.



Scheme 3.4. a) Hydroindolation of alkynes with N-(2-pyridyl)-indoles b) Hydroindolation of alkynes using imine as a directing group.



Scheme 3.5. Palladium catalyzed dehydrogenative coupling of arenes and alkenes.

In addition to hydroarylation, dehydrogenative coupling of (hetero)arene and alkene is also emerged as a powerful tool for alkenylation of (hetero)arenes. Fujiwara's group first revealed a palladium catalyzed dehydrogenative coupling of electron-rich aromatics with styrene which proceeds through electrophilic metalation pathway.¹⁵ Next, enormous efforts were devoted for the dehydrogenative alkenylation of arenes and heteroarenes.

In 1997, Miura et al disclosed a protocol for oxidative coupling of phenol derivatives with activated olefins using palladium catalyst in presence of stoichiometric amounts of oxidant.¹⁶ In this example, regioselective alkenylation is achieved by the coordination of Pd catalyst to hydroxy group. Later, chelation assisted Pd-catalyzed oxidative coupling was extended to benzoic acids, aryl sulfonamides and anilides by Miura, Yu and Van Leeuwen.¹⁷ The plausible mechanism for the dehydrogenative Heck reaction is depicted in Scheme 3.5.¹⁸ Miura, Fagnou, Satoh and Glorius expanded the scope of dehydrogenative coupling using rhodium catalysts in presence of stoichiometric amounts of oxidant.¹⁹ In the recent years, less expensive Ruthenium catalysts have been explored for the directed C-H alkenylation and significant advancement has been made by Milstein, Ackermann, Jagenmohan and many others.



Scheme 3.6. Dehydrogenative coupling of olefin and akenes through directed C-H bond activation.

Ackermann's group established a oxidative coupling of aromatic acids and alkenes using ruthenium catalyst to produce cyclic phthalides.²⁰ Inspired by this work Jagenmohan and coworkers extended the ruthenium catalysed dehydrogenative coupling with broad range of arenes and olefins.²¹ In their initial reports, stoichiometric amounts of copeer and silver oxidants were used for dehydrogenative heck reaction. In 2017, a new protocol was established for C-H alkenylation using Ruthenium catalyst without any external oxidant. This reaction proceeds at room temperature and suitable for *ortho* alkenylation of aromatic amides, ketoximes and anilides (Scheme 3.6.).²²

In 2011, Miura et al developed a protocol for C-H alkenylation of various heteroarenes such as indole, pyrrole, thiophene, benzothiophene and furan through the coupling of heteroarene carboxylic acids (**3.27**) and activated olefins (**3.28**) using Ru(II)/Cu(II) catalytic system.²³ This transformation occurs via chelation assisted C-H bond activation of carboxylic acid. Recently, the group of Baidya established another method for dehydrogenative alkenylation by coupling benzoic acids (**3.30**) and styrene (**3.31**) using ruthenium catalyst in presence of copper oxidant.²⁴ This method enables the easy synthesis of 2-styrylbenzoic acids and it occurs with broad scope, and good yields.



Scheme 3.7. Dehydrogenative alkenylation using COOH as a directing group.

Directed C-H bond activation is limited to functionalization of *ortho* C-H bond to a directing group and functionalization at *meta*, and *para* positions remained a challenge. Recently many methods have been developed for the *meta* C-H bond functionalization. Hartwig et al. reported a protocol for the *meta* borylation of arenes using Iridium catalyst (Scheme 3.8 a). The regioselectivity was controlled through the steric effects of the substrate. This transformation is suitable only for disubstituted arenes to achieve *meta* regioselectivity, whereas mono substituted arenes produced the mixture of isomers.²⁵



Scheme 3.8. a) Iridium catalyzed *meta* borylation of arenes. b) Cu catalyzed *meta* arylation of amides.

In the year 2009, Gaunt and coworkers discovered a new method for *meta* arylation of benzanilides (**3.36**) using copper(II) catalyst. This method is complementary to both chelation

assisted C-H activation and Friedel-Crafts reaction. The reaction proceeds via *meta*-C–H bond cupration via dearomatizing oxy-cupration as described in Scheme 3.8b. Upon rearomatization followed by reductive elimination delivers the desired *meta* arylated product.²⁶

Yu and coworkers described a new protocol for *meta* functionalization by cleverly designed a template as shown in Scheme 3.9. In their first report, a nitrile-based template was designed for *meta* alkenylation of toluene and hydrocinnamic acid derivatives using the catalytic amount of Pd(OPiv)₂. Later, template-directed C-H bond activation was extended to *meta* arylation, acetoxylation and alkylation of toluene, and amine derivatives.²⁷



Scheme 3.9. Template-directed *meta* C-H bond functionalization of toluene derivatives. Inspired by the success story of template-directed *meta* C-H bond activation, Maiti and coworkers devised a new template for the exclusive *para* C-H bond activation as depicted in Scheme 3.10. Authors reported an easily recyclable, biphenyl-silyl-tethered D-shaped assembly for *para* functionalization of toluene derivatives. Incorporation of biphenyl in the template regulates the chain length of the assembly and increases the strain for *ortho* and *meta* C-H bond activation, which facilitates the selective distal *para* C-H bond activation. The utility of Dshaped assembly is demonstrated by the alkenylation and acetoxylation of toluene derivatives in the presence of Pd(II) catalytic system.²⁸



Scheme 3.10. Template-assisted para C-H olefination of toluene derivatives.

Although these protocols are applicable for *meta* and *para* C-H bond functionalization, covalent linkage of complex templates remained as a drawback. Recently the groups of Yu and Dong reported an elegant approach for the *meta* C-H bond functionalization using norbornene as a transient mediator (Scheme 3.11).



Scheme 3.11. Norbornene as a transient mediator for *meta*-C–H activation.

Yu and coworkers reported metal alkylation and arylation of phenylacetic amides, Whereas Dong et al. disclosed *meta* arylation of tertiary amine using Pd/norbornene catalysis. These protocols imply the possibility of obtaining *meta* or *ortho* selectivity through catalyst control. ²⁹ However, transition metal catalyzed directed C-H bond activation requires extra steps for installation and removal of directing group or template. The ideal directing group could be commercially available and readily removable after C-H bond functionalization. In this context, the traceless directing group is considered as an elusive strategy for the site-selective C-H bond functionalization. This strategy provides access to *meta* or *para* functionalized arenes with atom and step economy (Scheme 3.12).



Scheme 3.12. Transition metal catalyzed C-H bond functionalization using traceless directing group.

In the recent years, -COOH has been employed as a traceless directing group because carboxylic acids are readily available, cheap, air stable and easy to remove.^{30, 31} In this regard, decarboxylative C-H bond functionalization is considered as an elegant strategy for the construction of C-C bonds. Miura and coworkers reported oxidative coupling of benzoic acid with alkynes for the synthesis of coumarins (**3.54**) using Rh(III) as a catalyst and Cu(OAc)₂ oxidant (Scheme 3.13). This oxidative cyclization occurs via a tandem reaction of carboxylic acid directed C-H bond activation, alkyne insertion, and oxidative cyclization. Iridium-catalyzed [2+2+2] decarboxylative cyclization of benzoic acid and alkyne was also realized at elevated temperature in the presence of the Ag₂CO₃ oxidant to afford naphthalene derivatives (**3.55**) (Scheme 3.14). The reaction was proposed to proceed *via* formation of the cyclometalated complex (iridacycle), followed by alkyne insertion and decarboxylation. Second alkyne insertion and then reductive elimination furnished the cyclization product.³²

97



Scheme 3.13. Rhodium-catalyzed dehydrogenative coupling of carboxylic acids with alkynes.



Scheme 3.14. Iridium-catalyzed decarboxylative coupling of carboxylic acids and alkynes.

Recently, Miura et al. first revealed the potential of benzoic acid as a traceless directing group in the year 2008 (Scheme 3.15 a). In the initial report, they disclosed palladium catalyzed decarboxylative alkenylation of heteroarene carboxylic acids in the presence of the stoichiometric amount of copper. The reaction occurs via carboxylic acid directed C-H bond functionalization followed by decarboxylation to produce corresponding products.³³ They subsequently developed a one-pot procedure for the decarboxylative alkenylation of styrenes with aromatic carboxylic acids using Rh(III)/Ag(I) catalytic system to obtain stilbene derivatives (Scheme 3.15 b).



Scheme 3.15. a) Palladium-catalyzed decarboxylative alkenylation of heteroarene carboxylic acids b) Rhodium-catalyzed decarboxylative *ortho*-olefination of benzoic acids.

The present reaction proceeds in two steps. The first step is rhodium-catalyzed C-H bond functionalization through directed C-H bond activation and the second step is silver mediated decarboxylation.³⁴ Larrosa et al. applied traceless directing group strategy to furnish biaryl compounds through decarboxylative coupling of *ortho*-substituted benzoic acids (**3.57**) and aryl iodides (**3.59**) using Pd catalyst (Scheme 3.16 a).³⁵ This reaction was performed in the presence of acetic acid and the stoichiometric amount of silver. Silver catalyzed protodecarboxylation was avoided using acetic acid. Authours demonstrated that Pd catalyst performed the protodecarboxylation of arylated product. Later in 2014, a new methodology was described using traceless directing group relay strategy for *meta* functionalization of phenol (**3.61**) (Scheme 3.16 b).³⁶ This transformation uses carbon dioxide as a transient directing group for *meta* arylation using aryl iodide as coupling partner. The reaction proceeds via carboxylation followed by tandem arylation and protodecarboxylation. This protocol occurs with high regioselectivity and compatible with many functional groups.



Scheme 3.16. a) Rhodium-catalyzed *ortho*-olefination and decarboxylation of benzoic acids. b) Traceless directing group relay strategy for *meta*-selective arylation of phenols.

The group of You discovered a new catalytic method for decarboxylative cross-coupling of carboxylic acids (**3.57**) and heteroarenes (**3.63**) (Scheme 3.17). This reaction proceeds through tandem Rh(III) catalyzed carboxylic acid directed C-H arylation/Ag-catalyzed protodecarboxylation and it occurres with broad substrate scope, and functional group tolerance. This transformation enables the easy synthesis of *meta*-substituted biaryl scaffolds with heteroarenes.³⁷



Scheme 3.17. Rh(III)-Catalyzed decarboxylative *ortho*-heteroarylation of aromatic carboxylic acids.

Recently, Li and co-workers disclosed another protocol for ruthenium catalyzed regiospecific C-H bond functionalization of carboxylic acids using isocyanate (**3.65**) as a coupling partner (Scheme 3. 18).³⁸ This transformation occurs via carboxylic acid directed C-H bond functionalization followed by decarboxylation. This method offers access to *meta*-substituted benzamides (**3.66**) which are difficult to synthesize in a single step.



Scheme 3.18. Ru(II)-catalyzed decarboxylative ortho-amidation of aromatic acids.

Although numerous methods have been developed for decarboxylative functionalization of arenes using transition metal catalysts, most of them need an *ortho*-inductive electronwithdrawing aromatic substituent and stoichiometric amounts of additives, and oxidants to facilitate the transformations which limit the substrate scope and functional group tolerance. Zhao group et al. established a new protocol for alkyne hydroarylation using arene carboxylic acids as coupling partner.³⁹ Authors envisioned that the reaction proceeds in a tandem sequence of carboxylic acid directed alkenylation followed by chelation(alkenyl) assisted decarboxylation.



Scheme 3.19. Decarboxylative alkyne hydroarylation with benzoic acids via double chelationassistance

Authors tested the proposed hypothesis by taking benzoic acid and diphenylacetylene as reaction partners. After the extensive screening of ligands and solvents decarboxylative hydroarylation is achieved in 90% yield using 10 mol% [Ru(p-cymene) (OAc)₂] as a catalyst in the mixed solvent system (2:2:1 dioxane, mesitylene, and heptane) at 80 °C. In addition, formation of byproducts such as alkyne hydrocarboxylation,⁴⁰ oxidative [4+2] heterocyclization,⁴¹ and oxidative [2+2+2] carbocyclization via decarboxylation was observed. The present reaction occurs with high regio and stereoselectivity, and high functional group tolerance.

Substrate scope of the arene carboxylic acids was studied by taking diphenyl acetylene as coupling partner. *para*-Substituted arene carboxylic acids worked well under optimized conditions and furnished *meta*-substituted alkenyl arenes. The reaction of *meta* substituted arene carboxylic acids under established conditions lead to formation *para* alkenylation products. Later alkyne scope was investigated with the model reaction between the *para* methoxy benzoic acid

and various alkynes. All internal alkynes proceeded smoothly with high yields and regio and stereoselectivity.



Scheme 3.20. Substrate scope of decarboxylative alkyne hydroarylation with benzoic acids. Based on "double chelation assistance" hypothesis, authors proposed a plausible mechanism for the decarboxylative hydroarylation as depicted in Scheme 3.21. It was proposed that the formation of cyclometalated intermediate II via carboxylate directed C-H bond

activation followed by insertion of alkyne into the metal-aryl bond generates cyclometalated alkenylruthenium (**III**). Protonation of metal alkenyl intermediate produces alkenyl-chelated Ru(II) carboxylate intermediate (**IV**). chelation-assisted decarboxylation of intermediate **IV** leads to the formation of aryl-ruthenium intermediate **V**. Upon protonation of **V** affords the decarboxylative hydroarylation product.



Scheme 3.21. Ruthenium catalyzed decarboxylative hydroarylation of alkynes.

The newly developed ruthenium catalyzed decarboxylative alkyne hydroarylation occurs under redox neutral conditions with broad range of functional group tolerance and substrate scope. This method also obviates the need for *ortho* substituted arene carboxylic acids and stoichiometric amounts of bases or oxidants. This protocol provides an easy route for the synthesis of various *meta*, *para*-substituted alkenyl arenes which are difficult to achieve via conventional Hydroarylation. This strategy demonstrates the synthetic utility of arene carboxylic acids as readily available synthons for transition metal catalyzed C-H bond functionalization.

In this chapter we describe ruthenium-based decarboxylative alkenylation of heteroarenes through directed C-H bond functionalization. The decarboxylative functionalization of heteroarenes occurs with high regioselectivity and broad range of functional group tolerance.

This decarboxylation proceeds without stoichiometric amounts of bases or Ag/Cu salts and it is applicable for functionalization of various hetero arenes such as indole, pyrrole, thiophene, benzothiophene, and benzofuran at both C-2 and C-3 positions. This protocol provides a straightforward approach for the synthesis of trisubstituted olefins with heteroarenes.

3.2. Initial Results

Recently the groups of Zhao, Gooßen and Ackermann reported decarboxylative hydroarylation of alkynes with aromatic carboxylic acids using ruthenium catalyst. Inspired by these reports and in continuation of our efforts towards development transition metal catalyzed C-H bond functionalization, we investigated decarboxylative hydroheteroarylation of alkynes with heteroarene carboxylic acids using ruthenium catalysts. After preliminary testing, we observed the formation of desired product in 38% yield using [RuCl₂(p-cymene)]₂ in the presence of AgSbF₆ and NaOAc in toluene at 100 °C.



Scheme 3.22. Initial result of ruthenium catalyzed decarboxylative hydro heteroarylation with Nmethyl indole-3-acid and diphenylacetylene.

3.3. Optimization of Reaction Condition

Encouraged by the initial result, we screened various ruthenium catalysts as shown in Table 1. Best result was obtained with $[Ru(OAc)_2(p-cymene)]_2$ in toluene at 100 °C. After testing the several solvents, dioxane was considered as the optimal solvent. Increasing the temperature to 120 °C improved the reaction efficiency and the desired product was obtained in 84% yield. The optimized condition for decarboxylative hydro heteroarylation of alkynes with heteroarene carboxylic acids was found to be $[Ru(OAc)_2(p-cymene)]_2$ (10 mol%) in Dioxane at 120 °C.

COOH , + Ph- 3.69a	—— Ph 3.02a	Catalyst, A Temperatur Solven	$\frac{\text{dditive}}{\text{re}(^{\circ}\text{C})}$	3.70a	Ph
Catalyst (mol%)	A	dditive	Temp	Solvent	yield(%) ^[a]
$[RuCl_2(p-cymene)]_2$	Ag	$gSbF_6 + NaO$	Ac 100	Toluene	38
$[RuCl_2(p-cymene)]_2$	Ag	$gSbF_6 + Ag_2C$	O ₃ 100	Toluene	23
$[RuCl_2(p-cymene)]_2$	Ν	aOAc	120	Toluene	53
[Ru(OAc) ₂ (<i>p</i> -cymene)]		-	120	Toluene	56
[Ru(OBz) ₂ (<i>p</i> -cymene)]	-		120	Toluene	48
[Ru(MesCO ₂) ₂ (<i>p</i> -cymene	e)] -		120	Toluene	53
[Ru(OPiv) ₂ (<i>p</i> -cymene)]	-		120	Toluene	45
[Ru(OAc) ₂ (<i>p</i> -cymene)]	Pi	valic acid	120	Toluene	52
[Ru(OAc) ₂ (<i>p</i> -cymene)]	A	cetic acid	120	Toluene	55
[Ru(OAc) ₂ (<i>p</i> -cymene)]	Tr	ifluoroacetic	acid120	Toluene	0
[Ru(OAc) ₂ (<i>p</i> -cymene)]	K ₂	CO ₃	120	Toluene	34
[Ru(OAc) ₂ (<i>p</i> -cymene)]	K ₂	HPO ₄	100	Toluene	32
[Ru(OAc) ₂ (<i>p</i> -cymene)]	KI	H_2PO_4	120	Toluene	32
[Ru(OAc) ₂ (<i>p</i> -cymene)]	KI	H ₂ PO ₄	120	Toluene	32
$[Ru(OAc)_2(p-cymene)]$	-		120	DCE	52
[Ru(OAc) ₂ (<i>p</i> -cymene)]	-		120	DME	44
$[Ru(OAc)_2(p-cymene)]$	-		120	DMF	52
$[Ru(OAc)_2(p-cymene)]$	-		120	CH ₃ CN	44
$[Ru(OAc)_2(p-cymene)]$	-		120	Xylene	52
$[Ru(OAc)_2(p-cymene)]$	-		120	Benzene	44
$[Ru(OAc)_2(p-cymene)]$	-		120	THF	59
$[Ru(OAc)_2(p-cymene)]$	-		120	Mixed	54
$[Ru(OAc)_2(p-cymene)]$	-		120	Dioxane	68
$[Ru(OAc)_2(p-cymene)]$	-		120	Dioxane	87 ^[b]

Table 3.1. Optimization of ruthenium catalysed decarboxylative hydroheteroarylation of internal alkynes.

[a] General reaction conditions: 3.69a (0.30 mmol, 1.3 equiv),

3.02a (0.15 mmol, 1.0 equiv), [Ru] (0.015mmol, 0.10 equiv), dioxane (1.0 mL), 120 °C, 48 h; NMR yields using 1,3,5-trimehoxy bezene as internal standard.

[b]: 2 equiv of 3.69a instaed of 1.3 equiv

3.4. Substrate Scope



General reaction conditions: 3.69 (0.30 mmol, 2.0 equiv), 3.02a (0.15 mmol, 1.0 equiv), [Ru] (0.015mmol, 0.10 equiv), dioxane (1.0 mL), 120 °C, 48 h; averaged isolated yields of two runs.

Figure 3.1. Substrate scope of heteroarene 3-carboxylic acids in ruthenium catalyzed decarboxylative alkyne hydroheteroarylation with diarylacetylenes.

With the establishment of optimized conditions, the scope of heteroarene-3-carboxylic acids (**3.69**) was investigated using diphenylacetylene (**3.02a**) as a reaction partner. The reaction of Indole-3-carboxylic acid (**3.69a**) with diphenylacetylene (**3.02a**) proceeded smoothly, and the desired product was isolated in 84% yield. Later, various indole 3-carboxylic acids (**3.69g-j**) bearing electron donating and withdrawing groups were subjected to ruthenium catalyzed decarboxylative hydro heteroarylation. All indole-3-carboxylic acids were coupled efficiently irrespective of electronics, and corresponding products (**3.70g-j**) were obtained in 72-81% yield. All the products were afforded as a mixture of E and Z isomers. Next, established ruthenium

catalytic system was extended to other heteroarene 3-carboxylic acids such as thiophene, furan, pyrrole, benzothiophene, benzofuran and the desired products (**3.69b-f**) were afforded in 46-83% yield. Subsequently, we also examined the reactivity of heteroarene 2-carboxylic acids (**3.72**) by taking diphenylacetylene as a coupling partner and the corresponding decarboxylative hydro heteroarylated products (**3.73a-f**) were furnished in 25-81% yield.



General reaction conditions: 3.72 (0.30 mmol, 2.0 equiv), 3.02 (0.15 mmol, 1.0 equiv), [Ru] (0.015mmol, 0.10 equiv), dioxane (1.0 mL), 120 °C, 48 h; averaged isolated yields of two runs.

Figure 3.2. Substrate scope of heteroarene 2-carboxylic acids in ruthenium catalyzed decarboxylative alkyne hydro heteroarylation with diarylacetylenes.

Next, the scope of diarylacetylenes was examined by taking indole-3-carboxylic acid

(3.69a) as a coupling partner. Decarboxylative hydro heteroarylation (hydroindolation) of

symmetrical diarylacetylenes proceeded efficiently, and the corresponding products were

afforded in good yields 62-79%. There was no loss of reactivity with the change of electronics at

the para position. Although reactivity is good with para Br and para CF3 substrates, to obviate

isolation problem thiophene-2-carboxylic acid (**3.72e**) was chosen as coupling partner and desired products are furnished in 67% and 83% yields. Dimethyl acetylene dicarboxylate also coupled efficiently with thiophene-2- carboxylic acid to give the product **3.73i** in 61% yield. Upon reaction of unsymmetrical alkyne 1-phenyl-1-propyne with thiophene 2-carboxylic acids using stabilized ruthenium catalytic condition provided the desired product (**3.73j**) in 63% yield. The present catalytic system tolerates various functional groups such as methoxy, methyl, bromo, chloro, fluoro, and ester and occurred with high regioselectivity, and good yields.



Scope of Alkynes. General reaction conditions: 3.69a or 3.72e (0.30 mmol, 2.0 equiv), 3.02 (0.15 mmol, 1.0 equiv), [Ru] (0.015 mmol, 0.10 equiv), dioxane (1.0 mL), 120 °C, 48 h; averaged isolated yields of two runs



3.5. Reaction Mechanism Studies

To further understand the mechanism, series of experiments were performed as shown in Scheme 3.23 a. First, intermolecular competition reaction was performed with an equimolar mixture of **3.69g** and **3.69j** by reacting with diphenylacetylene and the corresponding products were obtained in (**3.70g : 3.70j**) 1:1.3 in ratio. High reactivity of electron-deficient indole acid (**3.69j**) suggests that electrophilic C-H activation mechanism is less likely. Second, decarboxylative hydroindolation was carried out in the presence of deuterated acetic acid and incorporation of deuterium in **3.73a** demonstrates the intermolecular protonation. Overall these experiments prove the present reaction mechanism is consistent with the previously reported mechanism for decarboxylative hydroarylation which occurs through the Ruthenium catalysed chelation assisted C-H bond alkenylation followed by chelation assisted decarboxylation.



Scheme 3.23. Results from reaction mechanism studies (a) Intermolecular competition experiments (b) Deuterium labeling study.

3.6. Optimization of Reaction Condition

In addition to C-H hydro(hetero)arylation of alkynes, cross dehydrogenative coupling of arene or heteroarene C-H bonds with alkenes is also emerged as an essential tool for the synthesis of vinyl (hetero)arenes. As disucussed earlier, various ruthenium catalysts were explored for oxidative coupling of (hetero)arenes and alkenes. In the year 2011, Miura et al. reported regioselective oxidative vinylation heteroarene carboxylic acids with alkenes using ruthenium catalyst. We attempted to develop a new method for the decarboxylative oxidative coupling heteroarene carboxylic acids with activated olefins using ruthenium catalyst. We began our study by taking indole-2-carboxylic acid (3.72) and tertiary butyl acrylate as model substrates. After screening various oxidants in the presence of $[Ru(OAc)_2(p-cymene)]$ catalyst best result was obtained with V₂O₅. Dioxane proved to be optimal solvent for this reaction.

N + N He	OR Catalyst	t, Additive ature (°C)		OR
3.69	3.74		Me	3.75
Catalyst (mol%)	Additive	Temp (°C) Solvent	yield(%) ^{[a], [b]}
[Ru(OAc) ₂ (<i>p</i> -cymene)]	Ag ₂ CO ₃	100	Dioxane	31
[Ru(OAc) ₂ (<i>p</i> -cymene)]	AgOAc	100	Dioxane	35
[Ru(OAc) ₂ (<i>p</i> -cymene)]	Ag_2O	100	Dioxane	24
[Ru(OAc) ₂ (<i>p</i> -cymene)]	$K_2S_2O_8$	100	Dioxane	41
[Ru(OAc) ₂ (<i>p</i> -cymene)]	Benzoquinone	100	Dioxane	55
[Ru(OAc) ₂ (<i>p</i> -cymene)]	PhI(OAc) ₂	100	Dioxane	55
[Ru(OAc) ₂ (<i>p</i> -cymene)]	CuO	100	Dioxane	34
[Ru(OAc) ₂ (<i>p</i> -cymene)]	Cu ₂ O	100	Dioxane	32
[Ru(OAc) ₂ (<i>p</i> -cymene)]	$Cu(OAc)_2$	100	Dioxane	52
[Ru(OAc) ₂ (<i>p</i> -cymene)]	$Cu(OAc)_2$	100	Toluene	33
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V_2O_5	100	Dioxane	49
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V_2O_5	120	Dioxane	87
[Ru(OAc) ₂ (<i>p</i> -cymene)]	$Cu(OAc)_2$	120	Toluene	85
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V_2O_5	100	Xylene	44
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V_2O_5	120	DCE	68
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V_2O_5	120	Acetonitril	e 37
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V ₂ O ₅	120	DME	59
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V_2O_5	120	DMF	41
[Ru(OAc) ₂ (<i>p</i> -cymene)]	V_2O_5	120	THF	59

Table 3.2. Optimization of ruthenium catalysed decarboxylative alkenylation of heteroarenes with activated olefins.

[a] General reaction conditions: 3.69 (0.30 mmol, 2.0 equiv),

3.74 (0.15 mmol, 1.0 equiv), [Ru] (0.015mmol, 0.10 equiv), oxidant (2 equiv), dioxane (1.0 mL), 120 °C, 48 h; [b] GC-MS yields.

3.7. Substrate Scope

With best reaction conditions in hand, we examined the scope of heteroarene-2carboxylic acids such as indole, thiophene, pyrrole, benzothiophene and benzofuran in ruthenium catalyzed decarboxylative C-H alkenylation and the corresponding products (**4.75a-4.75f**) are furnished in 35 to 82% yield. Indole-3- carboxylic acid(**4.69a**) is also proceeded smoothly under optimized conditions to afford 2-alkenylated indole (**4.75f**). However, other heteroarene-3carboxylic acids didn't react to give the desired products. This reaction proceeds in a tandem sequence of ruthenium catalyzed carboxylic acid directed C-H alkenylation followed by decarboxylation.



General reaction conditions: 3.69 or 3.72 (0.30 mmol, 2.0 equiv), 3.74 (0.15 mmol, 1.0 equiv), [Ru] (0.015 mmol, 0.10 equiv), V_2O_5 (2.2 equiv), dioxane (1.0 mL), 120 °C, 48 h; averaged isolated yields of two runs. a: 5 equiv of acid is used

Figure 3.4. Substrate scope for ruthenium catalyzed decarboxylative alkenylation of heteroarenes.

3.8. Conclusion

In summary, we disclosed a new protocol for ruthenium catalyzed decarboxylative alkenylation of various heteroarenes through C-H bond activation. This transformation proceeds with high regioselectivity and enables alkenylation of heteroarenes at both second and third positions. Moreover, the current study demonstrates the potential of carboxylic acid as a traceless removal directing the group to achieve regioselectivity in transition metal-mediated directed C-H bond activation.

3.9. Experimental Procedures

3.9.1. General Experimental Procedures and Reagent Availability

Unless otherwise noted, all manipulations were carried out under a nitrogen atmosphere using standard Schlenk-line or glove box techniques. All glassware was oven-dried for at least one hour before use. Toluene, THF, benzene, and hexane solvents were degassed by purging with nitrogen for 45 min and dried with a solvent purification system (MBraun MB-SPS). Other reagents and starting materials for substrate synthesis were purchased from Sigma-Aldrich, VWR, Strem, Oakwood Chemical or Ark-Pharm and were used as received. TLC plates were visualized by exposure to ultraviolet light. Organic solutions were concentrated by rotary evaporation at ~10 Torr. Flash column chromatography was performed with 40–63 microns silica gel. 1H and 13C NMR spectra were obtained on a 400 MHZ spectrometer, and chemical shifts were recorded relative residual protiated solvent. Both 1H and ¹³C NMR chemical shifts were reported in parts per million downfield from tetramethylsilane ($\delta = 0$ ppm). ¹⁹F NMR spectra were obtained at 282.4 MHz, and all chemical shifts were reported in parts per million upfield of CF₃COOH ($\delta = -78.5$ ppm). GC analyses were performed on a Shimadzu GC-2010 system. GC-MS data were obtained on an Agilent 7890AGC system and an Agilent 5975C mass selective detector. High-resolution mass spectra were obtained at a Waters HRMS spectrometer.

3.9.2. Experimental Procedure for Preparation of Alkyne Substrates

Diaryl alkynes were prepared by following the reported literature.⁴²

$$\begin{array}{c|c} & & \\ \hline \\ R \end{array} + \end{array} = -SiMe_3 \end{array} \xrightarrow{PdCl_2(PPh_3)_2, Cul} \\ \hline \\ DBU, H_2O, Benzene \\ r.t or 60-80 \ C, 18 \ h \end{array} \xrightarrow{R} \xrightarrow{R} \\ \end{array}$$

Scheme 3.24. Synthesis of internal alkynes.

i) General procedure for sonogashira coupling at room temperature (Symmetrical): Into a flame-dried 100 mL flask equipped with a magnetic stir bar was charged with $PdCl_2(PPh3)_2(16.8 mg, 6 mol\%)$, CuI (15.2 mg, 10 mol%) and starting material iodide. The flask was purged with nitrogen for 20 min, and then dry benzene was added. While stirring, DBU (6 equiv) was added, followed by Ice-chilled trimethylsilyl ethynylene (57 μ L, 0.50 equiv) and distilled water (5.8 μ L, 40 mol%) were added. The reaction mixture was purged with nitrogen, and the flask was covered in aluminum foil while stirring the mixture at room temperature for 18 hours. After that, the reaction mixture is partitioned into ethyl ether and distilled water (50 mL each). The organic layer is washed with 10% HCl (3X 75 mL), saturated aqueous NaCl (1X75 mL) and dried over MgSO4. The solution was filtered, and the solvent was removed in vacuum to afford crude product. The crude product was purified by silica gel column chromatography.

ii) General procedure for sonogashira coupling at elevated temperature (Symmetrical): Oven-dried 100 mL flask was purged with nitrogen and then charged with $PdCl_2(PPh_3)_2$ (27.4 mg, 6 mol%), CuI (24.8 mg, 10 mol%), and starting material iodide (1 equiv, 1.3 mmol). While stirring, dry benzene (6.5 mL, starting material is 0.20 M in dry benzene) was added followed by dropwise addition of DBU (1.17 mL, 6 equiv) under a nitrogen atmosphere. To the resulting mixture, Distilled water (5.6 µL, 40 mol%) and ice-chilled trimethylsilyl ethynylene (92 µL, 0.50 equiv) were added. The reaction mixture was transferred to a preheated oil bath at 60 °C and blocked from incidental light. After stirring the flask for 18 h, the reaction mixture was cooled to room temperature and partitioned in ethyl ether and distilled water (50 mL each). The organic layer was washed with 10% HCl (3X 75 mL) followed by saturated aqueous NaCl (1X 75 mL). The combined organic layers were dried over MgSO4 and the solvent removed in vacuum to obtain the crude product. The crude product is purified by silica gel column chromatography.

3.9.3. General Procedure for Synthesis of Heteroarene Carboxylic Acids^{43, 44}



Scheme 3.25. Synthesis of indole-3-carboxylic acids.

A flame-dried vial equipped with a magnetic stir bar was charged with indole (4.8 mmol, 1.2 equiv) and N-cyano-4- methyl-N-phenyl benzene sulfonamide (NCTS, 1) (4 mmol, 1.0 equiv). After addition of anhydrous DCE (4 mL) and $BF_3 \cdot OEt_2$ (0.4 mmol) vial was sealed with a silicone-lined screw-cap and stirred at 80 °C for 12 h. The reaction mixture was concentrated under vacuum and purified by column chromatography to furnish the desired cyanated product.

Into a 10 mL vial equipped with a stir bar was charged with cyano compound (3 mmol, 1 equiv) and 4N aqueous KOH (10 equiv). The resulting mixture was moved to the preheated oil bath and stirred at 105 °C for 12 h. After completion of the reaction, the reaction mixture was neutralized by 2N HCl aq. (2.8 mL) and p^{H} of the solution was adjusted to 2-3. The reaction mixture was extracted with DCM (600 mL), and the combined organic layers were dried over MgSO4. The solvent was removed under reduced pressure, and the solid was washed with Et₂O (3 × 10 mL) to afford the product as a colorless solid.

3.9.4. General Procedure for Ru-Catalyzed Alkyne Hydroheteroarylation

Into a 4 mL scintillation vial equipped with a magnetic stir bar was charged with $[Ru(OAc)_2(p-cymene)]$ (0.017 mmol), Alkyne (0.17 mmol), and heteroarene carboxylic acid (0.34 mmol). Subsequently, Dioxane (1 mL) was added, and the vial was sealed with a silicone-lined screw-cap and electrical tape. After stirring the reaction mixture at room temperature for 5 minutes, the vial was transferred out of the glovebox and stirred at 120 °C for 48 hours. The reaction mixture was cooled to room temperature, and the solvent was removed under reduced pressure to obtain the crude product. Further purification was achieved by flash-column chromatography on silica.

3.9.5. Intermolecular Competition Experiment

An oven dried 4 mL scintillation vial was charged with [Ru(OAc)₂(*p*-cymene)] (0.017 mmol), Alkyne (0.17 mmol), and heteroarene carboxylic acids 4.69g (0.34 mmol) and 4.69(0.34 mmol). After addition of Dioxane (1 mL) solvent, the vial was sealed with a silicone-lined screw-cap and electrical tape and the reaction mixture was stirred at room temperature for 5 minutes. The vial was transferred out of the glovebox and stirred at 120 °C for 30 minutes. After completion, the reaction mixture was cooled to room temperature, and the solvent was removed under reduced pressure to obtain the crude product. The crude product was purified by flash-column chromatography on silica.

3.9.6. Deuterium Labeling Study

Alkyne (0.17 mmol), heteroarene carboxylic acid (0.34 mmol) and $[Ru(OAc)_2(p-Cymene)]$ (0.017 mmol) were added into a 4 mL scintillation vial. To the above mixture, Dioxane (1 mL) and CH₃COOD (0.34 mmol) were added, and the vial was sealed with a silicone-lined screw-cap and electrical tape. After stirring the reaction mixture at room

temperature for 5 minutes, the vial was transferred out of the glovebox and stirred at 120 °C for 24 hours. The reaction mixture was cooled to room temperature, and volatiles were removed under reduced pressure to obtain the crude product. The crude product was subjected to further purification by flash-column chromatography on silica.

3.9.7. General Procedure for Ru-Catalyzed Decarboxylative Alkenylation of Heteroarene Carboxylic Acids

An oven dried 4 mL was charged with $[Ru(OAc)_2(p-Cymene)]$ (0.017 mmol), heteroarene carboxylic acid (0.34 mmol), and V₂O₅ (2 equiv). Subsequently, Dioxane (1 mL) and alkene (0.17 mmol) were added, and the vial was sealed with a silicone-lined screw-cap and electrical tape. The vial was transferred out of the glovebox and stirred at 120 °C for 36 hours. The reaction mixture was brought to room temperature, and the solvent was removed in vacuo to afford the crude product. The crude mixture was purified by flash-column chromatography on silica.

3.10. Spectral Data of Isolated Products



2-(1,2-diphenylvinyl)-1-methyl-1H-indole (3.70a): The compound **3.70a** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 84% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃): δ 7.80 – 7.72 (m, 2H), 7.55 – 7.22 (m, 25H), 7.18 – 7.12 (m, 2H), 7.09 (s, 1H), 6.72 (s, 0.7H), 6.59 (d, *J* = 0.8 Hz, 0.8H), 3.51 (d, *J* = 1.2 Hz, 6H). ¹³C NMR (100 MHz, CDCl₃): δ 143.6, 142.0, 139.1, 138.8, 138.5, 137.5, 136.9, 136.8, 134.3, 132.5, 132.0, 131.4, 130.2, 129.7,

129.1, 128.6, 128.5, 128.5, 128.4, 128.2, 128.0, 127.8, 127.7, 127.7, 127.3, 127.0, 122.0, 121.5, 120.9, 120.7, 119.8, 119.7, 109.7, 109.4, 103.8, 102.7, 77.5, 77.2, 76.8, 31.3, 30.31.HRMS: m/z calcd for C₂₃H₁₉N: 310.1596; found: 310.1588;



2-(1,2-diphenylvinyl)benzo[b]thiophene (3.70b): The compound **3.70b** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 55% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃) δ 7.90 – 7.80 (m, 1H), 7.76 (dd, *J* = 6.6, 2.0 Hz, 0.3H), 7.65 (dd, *J* = 6.1, 2.9 Hz, 1H), 7.58 – 7.32 (m, 10H), 7.30 – 7.03 (m, 8H), 6.93 (s, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 147.6, 142.8, 142.4, 140.9, 140.3, 139.9, 139.1, 138.9, 136.8, 136.7, 136.4, 135.2, 131.5, 130.0, 129.6, 129.6, 129.0, 128.9, 128.3, 128.2, 128.1, 128.0, 128.0, 127.7, 127.5, 127.3, 125.3, 124.8, 124.5, 124.3, 124.2, 123.7, 123.6, 123.5, 122.4, 122.1, 77.4, 77.1, 76.8. HRMS: m/z calcd for C₂₂H₁₆S: 313.1051; found: 313.1035;



2-(1,2-diphenylvinyl)benzofuran (3.70c): The compound **3.70c** was prepared according to the general method described above and purified by flash column chromatography (In hexanes) in 46% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃)) δ 7.80 – 6.99 (m, 15H), 6.33 (s, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 158.01, 155.11, 137.61, 136.32, 131.60,

130.10, 129.83, 129.26, 129.05, 128.18, 128.14, 127.42, 124.79, 122.94, 121.01, 111.02, 106.25. HRMS: m/z calcd for C₂₂H₁₆O: 297.1279; found: 297.1279.



2-(1,2-diphenylvinyl)-1-methyl-1H-pyrrole (3.70d): The compound **3.70d** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.5% ethyl acetate in hexanes) in 74% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.59 – 7.11 (m, 15H), 7.07 – 6.98 (m, 2H), 6.92 – 6.64 (m, 2H), 6.40 – 6.19 (m, 2H), 6.18 – 6.09 (m, 1H), 3.34 (s, 2H), 3.27 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 156.0, 142.3, 137.8, 136.6, 131.7, 129.9, 129.5, 128.8, 128.0, 127.9, 126.8, 124.5, 111.6, 109.5, 77.4, 77.1, 76.8. HRMS: m/z calcd for C₁₉H₁₇N: 260.1439; found: 260.1438;



2-(1,2-diphenylvinyl)thiophene (3.70e): The compound **3.70e** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 72% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃) δ 7.56 – 7.34 (m, 5H), 7.29 (dd, *J* = 5.2, 1.2 Hz, 1H), 7.25 – 7.15 (m, 4H), 7.06 (ddd, *J* = 22.4, 5.5, 3.0 Hz, 3H), 6.84 (dd, *J* = 3.6, 1.2 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 148.0, 139.5, 136.7, 136.3, 130.0, 129.5, 128.8, 128.1, 128.0, 127.9, 127.5, 126.9, 126.4, 126.1, 124.8, 77.4, 77.1, 76.8. HRMS: m/z calcd for C₁₈H₁₄S: 263.0894; found: 263.0903;



2-(1,2-diphenylvinyl)furan(3.70f): The compound **3.70f** was prepared according to the general method described above and purified by flash column chromatography (In hexanes) in 61% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.56 – 7.36 (m, 6H), 7.30 (s, 1H), 7.22 – 7.14 (m, 3H), 7.11 – 7.04 (m, 2H), 6.44 (dd, *J* = 3.4, 1.8 Hz, 1H), 5.99 (d, *J* = 3.3 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 156.0, 142.3, 137.8, 136.6, 131.7, 129.9, 129.5, 128.8, 128.0, 127.9, 126.8, 124.5, 111.6, 109.5, 77.4, 77.1, 76.8. HRMS: m/z calcd for C₁₈H₁₄O: 247.1123; found: 247.1131;



2-(1,2-diphenylvinyl)-5-methoxy-1-methyl-1H-indole (3.70g): The compound **3.70g** was prepared according to the general method described above and purified by flash column chromatography (0 to 1% ethyl acetate in hexanes) in 72% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.47 – 7.17 (m, 17H), 7.13 (dd, *J* = 4.1, 2.4 Hz, 2H), 7.08 – 7.03 (m, 2H), 6.99 – 6.97 (m, 1H), 6.96 – 6.91 (m, 1H), 6.54 (d, *J* = 0.8 Hz, 1H), 6.41 (d, *J* = 0.8 Hz, 1H), 3.92 (d, *J* = 5.5 Hz, 5H), 3.41 (d, *J* = 9.5 Hz, 5H). ¹³C NMR (101 MHz, CDCl₃) δ 154.2, 154.2, 144.1, 141.9, 139.1, 138.9, 136.8, 136.7, 134.2, 134.1, 132.8, 132.4, 131.7, 131.1, 130.1, 129.6, 129.0, 128.5, 128.4, 128.4, 128.1, 127.9, 127.7, 127.5, 127.1, 126.9, 112.2, 111.8, 110.3, 110.1, 103.3, 102.3, 102.1, 77.3, 77.0, 76.7, 55.9, 55.8, 31.4, 30.3. HRMS: m/z calcd for C₂₄H₂₁NO: 340.1701; found: 340.1698;



2-(1,2-diphenylvinyl)-1,5-dimethyl-1H-indole (3.70h): The compound **3.70h** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 73% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃) δ 7.48 – 7.12 (m, 19H), 7.12 – 7.03 (m, 3H), 6.9 (s, 1H), 6.53 (s, 0.5H), 6.41 (s, 1H), 3.40 (d, *J* = 3.3 Hz, 5H), 2.52 (d, *J* = 12.3 Hz, 5H). ¹³C NMR (101 MHz, CDCl₃) δ 143.6, 142.1, 139.2, 138.5, 137.2, 136.9, 136.8, 135.9, 134.3, 132.5, 131.7, 131.0, 130.1, 129.7, 129.0, 128.9, 128.9, 128.5, 128.5, 128.4, 128.2, 128.1, 127.9, 127.7, 127.6, 127.2, 127.0, 123.6, 123.1, 120.5, 120.2, 109.4, 109.1, 103.3, 102.1, 77.4, 77.3, 77.1, 76.8, 31.3, 30.2, 21.5, 21.5. HRMS: m/z calcd for C₂₄H₂₁N: 324.1752; found:324.1742;



5-chloro-2-(1,2-diphenylvinyl)-1-methyl-1H-indole (3.70i): The compound **3.70i** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 65% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃) δ 7.62 (dt, *J* = 2.6, 1.6 Hz, 1H), 7.47 – 7.11 (m, 18H), 7.04 – 6.98 (m, 2H), 6.56 (s, 1H), 6.44 (s, 0.5H), 3.41 (d, *J* = 2.2 Hz, 5H). ¹³C NMR (101 MHz, CDCl₃) δ 144.8, 141.5, 139.8, 138.7, 137.0, 136.6, 136.5, 135.8, 133.7, 132.2, 131.9, 131.9, 130.0, 129.6, 129.1, 128.9, 128.6, 128.5, 128.4, 128.1, 128.0, 127.9, 127.7, 127.4, 126.8, 125.4, 125.3, 122.0, 121.7,

120.0, 119.8, 110.6, 110.3, 103.1, 102.2, 77.3, 77.0, 76.7, 31.4, 30.3. HRMS: m/z calcd for C₂₃H₁₈NCl: 344.1206; found: 344.1200;



2-(1,2-diphenylvinyl)-5-fluoro-1-methyl-1H-indole (3.70j): The compound **3.70j** was prepared according to the general method described above and purified by flash column chromatography (In hexanes) in 81% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.54 – 7.20 (m, 20H), 7.16 – 7.02 (m, 4H), 6.65 (s, 1H), 6.53 (s, 0.44H), 3.49 (d, *J* = 4.5 Hz, 5H). ¹³C NMR (101 MHz, CDCl₃) δ 159.3 , 159.2 , 157.0 , 156.9 , 145.1 , 141.7 , 140.1 , 138.9 , 136.8 , 136.6 , 135.4 , 134.1 , 134.0 , 132.2 (d, *J* = 2.5 Hz), 131.8 , 130.1 , 129.7 , 129.0 , 128.8 – 127.8 (m), 127.8 (d, *J* = 1.7 Hz), 127.4 , 126.9 , 110.4 – 109.9 (m), 109.7 , 105.6 , 105.4 , 105.2 , 103.7 (d, *J* = 4.6 Hz), 102.6 (d, *J* = 4.7 Hz), 31.5 , 30.4. ¹⁹F NMR (376 MHz, CDCl₃) δ -124.6. HRMS: m/z calcd for C₂₃H₁₈NF:328.1501; found: 328.1500.



2-(1,2-di-p-anisylvinyl)-1-methyl-1H-indole (370k): The compound 3.70k was prepared according to the general method described above and purified by flash column chromatography (0 to 1% ethyl acetate in hexanes) in 81% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃) δ 7.72 (dd, *J* = 11.2, 7.8 Hz, 2H), 7.46 – 7.17 (m, 11H), 7.04 – 6.72 (m, 9H), 6.66 (s, 1H), 6.54 (s, 0.5H), 3.93 – 3.83 (m, 8H), 3.79 (s, 2H), 3.48 (d, *J* = 9.7 Hz, 5H). ¹³C NMR
(101 MHz, CDCl₃) δ 159.3, 159.0, 158.8, 158.6, 144.1, 138.9, 138.6, 137.4, 134.8, 131.8, 131.6, 131.3, 130.8, 130.2, 130.1, 129.8, 129.7, 129.6, 128.4, 127.9, 127.7, 121.6, 121.2, 120.7, 120.4, 119.6, 119.5, 113.9, 113.8, 113.6, 109.6, 109.3, 103.3, 102.2, 77.4, 77.1, 76.7, 55.3, 55.2, 55.1, 31.2, 30.1. HRMS: m/z calcd for C₂₅H₂₃NO₂: 370.1807; found: 370.1802;



2-(1,2-di-p-tolylvinyl)-1-methyl-1H-indole (3.70I): The compound **3.70I** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 65% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.82 (t, *J* = 7.9 Hz, 2H), 7.59 – 6.98 (m, 25H), 6.78 (s, 1H), 6.65 (s, 1H), 3.58 (d, *J* = 6.4 Hz, 6H), 2.64 – 2.32 (m, 12H). ¹³C NMR (101 MHz, CDCl₃) δ 144.1, 139.4, 138.9, 138.8, 137.7, 137.5, 137.5, 137.1, 136.4, 134.3, 134.2, 133.4, 131.4, 131.2, 131.0, 130.1, 129.7, 129.4, 129.3, 129.0, 129.0, 128.5, 127.9, 126.9, 121.9, 121.5, 120.9, 120.7, 119.8, 119.7, 109.7, 109.4, 103.7, 102.5, 77.6, 77.2, 76.9, 31.4, 30.3, 21.5, 21.4, 21.4, 21.3. HRMS: m/z calcd for C₂₅H₂₃N: 338.1909; found: 338.1905;



2-(1,2-bis(4-chlorophenyl)vinyl)-1-methyl-1H-indole (3.70m): The compound **3.70m** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 65% yield as a colorless oil. ¹H NMR

(400 MHz, CDCl₃) δ 7.68 (dd, J = 7.8, 6.1 Hz, 1H), 7.45 – 7.09 (m, 16H), 6.97 (d, J = 8.6 Hz, 2H), 6.62 (s, 1H), 6.49 (s, 0H), 3.44 (d, J = 9.1 Hz, 4H). ¹³C NMR (101 MHz, CDCl₃) δ 142.6, 140.1, 138.8, 137.6, 137.4, 137.2, 135.1, 134.9, 134.0, 133.9, 133.7, 133.5, 133.1, 132.0, 131.5, 130.9, 130.7, 130.3, 130.2, 129.0, 128.7, 128.7, 128.5, 128.2, 128.2, 127.6, 122.3, 121.9, 121.0, 120.7, 120.0, 119.9, 109.8, 109.5, 104.3, 103.0, 31.4, 30.3. HRMS: m/z calcd for C₂₃H₁₇NCl₂: 378.0816; found: 378.0791;



2-(1,2-bis(4-fluorophenyl)vinyl)-1-methyl-1H-indole (3.70n): The compound **3.70n** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 65% yield as a yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.84 – 7.73 (m, 2H), 7.54 – 7.23 (m, 14H), 7.20 – 7.00 (m, 9H), 6.96 (t, *J* = 8.7 Hz, 2H), 6.74 (s, 1H), 6.60 (s, 1H), 3.54 (d, *J* = 5.7 Hz, 6H). ¹³C NMR (101 MHz, CDCl₃) δ 163.8 (d, *J* = 31.2 Hz), 163.2 (d, *J* = 12.4 Hz), 161.3 (d, *J* = 31.3 Hz), 160.8 (d, *J* = 11.8 Hz), 143.1 , 138.9 , 138.0 (d, *J* = 3.2 Hz), 138.0, 137.6, 134.9 (d, *J* = 3.5 Hz), 133.1 (d, *J* = 1.5 Hz), 133.0 (d, *J* = 3.4 Hz), 131.9 (d, *J* = 7.9 Hz), 131.4 (d, *J* = 7.9 Hz), 131.3 (d, *J* = 2.0 Hz), 130.8 (d, *J* = 7.9 Hz), 130.4 , 130.2 , 128.7 (d, *J* = 8.0 Hz), 128.3 , 127.7 , 122.3 , 121.9 , 120.9 (d, *J* = 23.4 Hz), 120.0 (d, *J* = 2.4 Hz), 116.0 , 115.7, 115.6 , 115.5 , 115.4 , 115.2 , 109.8 , 109.5 , 104.1 , 102.9 , 31.4 , 30.3. ¹⁹F NMR (376 MHz, CDCl₃) δ -112.7, -113.0, -113.4, -113.7. HRMS: m/z calcd for C₂₃H₁₇NF₂: 346.1407; found: 346.1392;



2-(1,2-di-m-tolylvinyl)-1-methyl-1H-indole (3.700): The compound **3.700** was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 71% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.80 (dd, *J* = 7.7, 5.0 Hz, 2H), 7.53 – 7.06 (m, 22H), 7.02 (s, 1H), 6.90 – 6.82 (m, 1H), 6.75 (s, 1H), 6.62 (s, 1H), 3.56 (d, *J* = 7.9 Hz, 6H), 2.50 (s, 3H), 2.42 (d, *J* = 9.2 Hz, 6H), 2.33 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 143.9, 142.1, 139.2, 138.7, 138.7, 138.2, 138.1, 137.8, 137.7, 137.5, 136.9, 136.8, 134.2, 132.4, 132.0, 131.5, 130.6, 130.6, 130.5, 128.8, 128.6, 128.5, 128.4, 128.4, 128.4, 128.1, 128.0, 127.8, 127.6, 127.3, 126.7, 125.6, 124.3, 121.9, 121.5, 120.9, 120.7, 119.8, 119.7, 109.7, 109.5, 103.7, 102.6, 77.5, 77.2, 76.9, 31.4, 30.3, 21.6, 21.5, 21.5. HRMS: m/z calcd for C₂₅H₂₃N: 338.1909; found: 338.1906;



3-(1,2-diphenylvinyl)-1-methyl-1H-indole (3.73a): The compound 3.73a was prepared according to the general method described above and purified by flash column chromatography (0 to 0.1% ethyl acetate in hexanes) in 61% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.86 (d, *J* = 8.0 Hz, 1H), 7.62 – 7.15 (m, 27H), 7.13 – 7.03 (m, 2H), 7.00 (s, 1H), 6.89 (s, 1H), 3.81 (d, *J* = 20.3 Hz, 6H). ¹³C NMR (101 MHz, CDCl₃) δ 144.41, 141.50, 138.53,

138.18, 137.84, 137.26, 137.21, 135.48, 130.36, 129.76, 129.66, 129.37, 129.25, 128.63, 128.34, 128.19, 128.03, 128.01, 127.92, 127.54, 127.46, 127.41, 126.48, 126.28, 125.97, 124.49, 122.18, 121.72, 121.28, 121.11, 120.10, 119.46, 119.35, 114.02, 109.69, 109.27, 77.52, 77.20, 76.88, 32.96, 32.91. HRMS: m/z calcd for C₂₃H₁₉N: 310.1596; found: 310.1590;



3-(1,2-diphenylvinyl)benzo[b]thiophene (3.73b): The compound **3.73b** was prepared according to the general method described above and purified by flash column chromatography (in hexanes) in 55% yield as a colorless-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.95 (dd, *J* = 15.8, 8.0 Hz, 2H), 7.65 (d, *J* = 8.0 Hz, 1H), 7.55 – 7.04 (m, 29H). ¹³C NMR (101 MHz, CDCl₃) δ 142.20, 140.72, 140.40, 140.20, 140.12, 138.54, 138.00, 137.28, 137.15, 135.78, 135.48, 130.85, 130.12, 129.74, 129.68, 129.12, 128.63, 128.49, 128.16, 128.14, 127.80, 127.70, 127.14, 126.99, 126.92, 125.81, 125.39, 124.47, 124.29, 124.17, 123.75, 123.56, 122.95, 122.73, 77.44, 77.12, 76.80. HRMS: m/z calcd for C₂₂H₁₆S: 313.1051; found: 313.1042;



3-(1,2-bis(4-(tert-butyl)phenyl)vinyl)benzofuran (3.73c): The compound **3.73c** was prepared according to the general method described above and purified by flash column

chromatography (in hexanes) in 61% yield as a colorless-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.72 – 6.90 (m, 26H), 1.43 (s, 8.3H), 1.38 (s, 8.7H), 1.33 (s, 8.4H), 1.30 (s, 9H). ¹³C NMR (101 MHz, CDCl₃) δ 155.98, 155.49, 150.77, 150.69, 150.11, 149.77, 144.26, 143.76, 139.46, 136.92, 134.48, 134.25, 132.96, 130.29, 130.12, 129.39, 129.16, 129.06, 128.85, 127.71, 127.05, 126.74, 126.29, 125.61, 125.28, 125.21, 125.06, 124.99, 124.40, 124.33, 122.83, 122.64, 121.47, 121.45, 119.93, 111.75, 111.44, 77.38, 77.06, 76.74, 34.70, 34.59, 34.54, 31.48, 31.36, 31.28, 31.24. HRMS: m/z calcd for C₃₀H₃₂O: 409.2531; found: 409.2525;



3-(1,2-bis(4-(tert-butyl)phenyl)vinyl)-1-methyl-1H-pyrrole (3.73d): The compound **3.73d** was prepared according to the general method described above and purified by flash column chromatography 0 to 0.1% ethyl acetate in hexanes) in 71% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.48 – 7.36 (m, 3H), 7.33 – 7.24 (m, 3H), 7.18 – 7.11 (m, 2H), 6.96 – 6.84 (m, 3H), 6.62 (t, *J* = 2.5 Hz, 1H), 6.48 – 6.42 (m, 1H), 6.29 (t, *J* = 2.0 Hz, 1H), 3.66 (s, 0.5H), 3.60 (s, 3H), 1.44 (s, 9H), 1.39 (s, 1.5H), 1.36 (s, 1.5H), 1.30 (s, 9H). ¹³C NMR (101 MHz, CDCl₃) δ 150.02, 149.88, 149.16, 148.48, 142.03, 138.06, 136.31, 136.14, 135.95, 135.12, 129.18, 128.91, 128.88, 128.69, 128.64, 128.54, 127.52, 125.99, 125.40, 125.16, 125.11, 125.06, 124.80, 124.77, 122.64, 122.44, 122.29, 121.69, 121.58, 121.52, 110.43, 105.80, 77.38, 77.06, 76.75, 36.23, 36.20, 34.63, 34.53, 34.39, 31.54, 31.43, 31.38, 31.30. HRMS: m/z calcd for C₂₇H₃₃N: 372.2691; found: 372.2681;



3-(1,2-diphenylvinyl)thiophene (3.73e): The compound 5e was prepared according to the general method described above and purified by flash column chromatography 0 to 0.1% ethyl acetate in hexanes) in 81% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.67 (d, *J* = 7.3 Hz, 0.12H), 7.58 – 7.38 (m, 8H), 7.36 – 7.10 (m, 8H), 7.06 (dd, *J* = 2.9, 1.4 Hz, 1H).¹³C NMR (101 MHz, CDCl₃) δ 145.23, 143.31, 140.46, 140.39, 137.73, 137.55, 137.44, 137.19, 130.04, 129.63, 129.58, 129.51, 129.06, 128.93, 128.88, 128.42, 128.24, 128.17, 127.84, 127.82, 127.70, 127.69, 127.10, 126.86, 126.73, 126.12, 125.81, 125.63, 125.14, 123.22, 77.60, 77.28, 76.97. HRMS: m/z calcd for C₁₈H₁₄S: 263.0894; found: 263.0884;



3-(1,2-bis(4-bromophenyl)vinyl)thiophene (3.73g): The compound **3.73g** was prepared according to the general method described above and purified by flash column chromatography 0 to 0.1% ethyl acetate in hexanes) in 67% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.57 – 7.45 (m, 2H), 7.39 – 7.22 (m, 4H), 7.19 – 7.12 (m, 2H), 6.99 (s, 1H), 6.95 (dd, *J* = 3.0, 1.4 Hz, 1H), 6.91 – 6.85 (m, 2H). ¹³C NMR (101 MHz, CDCl₃) δ 144.24, 142.65, 142.26, 141.76, 139.41, 138.90, 138.71, 137.00, 136.92, 136.16, 136.01, 135.64, 132.15, 132.13, 132.02,

131.64, 131.43, 131.34, 131.32, 131.29, 131.07, 130.92, 130.86, 129.13, 129.11, 128.42, 128.02, 127.85, 127.68, 127.41, 126.11, 126.04, 125.88, 125.31, 123.49, 121.98, 121.91, 121.06, 120.84. HRMS: m/z calcd for C₁₈H₁₂Br₂S: 420.9084; found: 420.9081;



3-(1,2-bis(4-(trifluoromethyl)phenyl)vinyl)thiophene: The compound **3.73h** was prepared according to the general method described above and purified by flash column chromatography 0 to 0.1% ethyl acetate in hexanes) in 83% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.75 – 7.61 (m, 3H), 7.56 – 7.30 (m, 9H), 7.30 – 7.24 (m, 0.4H), 7.20 – 7.06 (m, 4H), 7.04 (s, 0.15H), 6.97 (dd, *J* = 2.9, 1.4 Hz, 1H), 6.91 (dd, *J* = 5.0, 1.3 Hz, 0.13H). ¹³C NMR (101 MHz, CDCl₃) δ 143.7, 143.6, 143.5 – 143.3 (m), 140.2 – 139.5 (m), 138.2, 130.8, 130.3, 130.0, 129.6, 129.5 (d, *J* = 5.3 Hz), 129.2, 128.9 (d, *J* = 3.3 Hz), 128.5 (d, *J* = 6.0 Hz), 128.3 (d, *J* = 13.4 Hz), 128.1 (d, *J* = 2.2 Hz), 127.9, 127.7 (d, *J* = 3.8 Hz), 126.4, 126.3, 126.0, 125.9 (q, *J* = 3.7 Hz), 125.0, 125.6, 125.4 (d, *J* = 2.0 Hz), 125.3 (q, *J* = 3.5 Hz), 125.1 (d, *J* = 3.7 Hz), 124.1, 122.9 – 122.4 (m), 120.0 (d, *J* = 1.2 Hz), 77.3, 77.0, 76.7. ¹⁹F NMR (376 MHz, CDCl₃) δ -62.48, -62.55, -62.57, -62.58. HRMS: m/z calcd for C₂₀H₁₂F₆S: 399.0642; found: 399.0638;



Dimethyl 2-(thiophen-3-yl)maleate (3.73i): The compound **3.73i** was prepared according to the general method described above and purified by flash column chromatography 0 to 0.1% ethyl acetate in hexanes) in 61% yield as a light yellow-oil. ¹H NMR (400 MHz, CDCl₃) δ 7.48 (dd, *J* = 2.9, 1.4 Hz, 1H), 7.38 (dd, *J* = 5.1, 2.9 Hz, 1H), 7.26 (dd, *J* = 5.1, 1.4 Hz, 1H), 6.26 (s, 1H), 3.96 (s, 3H), 3.79 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 168.24, 165.70, 143.47, 135.12, 127.30, 127.01, 124.94, 115.42, 77.38, 77.07, 76.75, 52.78, 52.02. HRMS: m/z calcd for C₁₀H₉O₄SNa:249.0197; found: 249.0196;



(E)-3-(1-phenylprop-1-en-2-yl)thiophene (3.73j): The compound 3.73j was prepared according to the general method described above and purified by flash column chromatography 0 to 0.1% ethyl acetate in hexanes) in 63% yield as a colorless oil. ¹H NMR (400 MHz, CDCl₃) δ 7.55 – 7.30 (m, 8H), 7.08 (s, 1H), 2.35 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 145.11, 138.11, 132.11, 129.30, 128.26, 126.55, 126.49, 125.66, 125.33, 120.36, 77.47, 77.16, 76.84, 17.24. HRMS: m/z calcd for C₁₃H₁₂S: 201.0738; found: 201.0731;



(E)-tert-butyl 3-(1-methyl-1H-indol-3-yl)acrylate(3.75a): The compound 3.75a was prepared according to the general method described above and purified by flash column chromatography (0 to 4% ethyl acetate in hexanes) in 82% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.95 (dt, *J* = 7.6, 1.1 Hz, 1H), 7.84 (d, *J* = 15.9 Hz, 1H), 7.37 – 7.23 (m,

4H), 6.40 (d, *J* = 15.9 Hz, 1H), 3.76 (s, 3H), 1.61 (s, 9H). ¹³C NMR (101 MHz, CDCl₃) δ 167.78, 138.04, 136.96, 132.81, 126.09, 122.84, 121.12, 120.59, 114.57, 112.10, 109.92, 79.72, 77.46, 77.14, 76.82, 33.07, 28.42. HRMS: m/z calcd for C₁₆H₁₉NO₂: 258.1494; found: 258.1494;



(E)-tert-butyl 4-(1-methyl-1H-pyrrol-3-yl)-2-oxobut-3-enoate (3.75b): The compound 3.75b was prepared according to the general method described above and purified by flash column chromatography (0 to 3% ethyl acetate in hexanes) in 62% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.50 (d, *J* = 15.7 Hz, 1H), 6.87 – 6.74 (m, 1H), 6.62 – 6.49 (m, 1H), 6.39 – 6.28 (m, 1H), 6.02 (d, *J* = 15.7 Hz, 1H), 3.62 (s, 3H), 1.52 (s, 9H). ¹³C NMR (101 MHz, CDCl₃) δ 167.51, 137.83, 124.66, 123.73, 120.80, 114.62, 106.82, 79.52, 77.45, 77.14, 76.82, 36.30, 28.31. HRMS: m/z calcd for C₁₂H₁₈NO₂:208.1338; found: 208.1335;



(E)-tert-butyl 2-oxo-4-(thiophen-3-yl)but-3-enoate (3.73c): The compound 3.73c was prepared according to the general method described above and purified by flash column chromatography (0 to 1% ethyl acetate in hexanes) in 68% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.59 (dd, *J* = 15.9, 0.7 Hz, 1H), 7.46 (dd, *J* = 2.9, 1.3 Hz, 1H), 7.37 – 7.23 (m, 2H), 6.21 (d, *J* = 15.9 Hz, 1H), 1.54 (s, 9H). ¹³C NMR (101 MHz, CDCl₃) δ 166.52, 137.75, 137.05, 127.46, 126.79, 125.18, 119.88, 80.39, 77.38, 77.07, 76.75, 28.22. HRMS: m/z calcd for C₁₁H₁₄O₂SNa: 233.0612; found: 233.0611;



(E)-benzyl 3-(benzo[b]thiophen-3-yl)acrylate (3.75d): The compound 3.75d was prepared according to the general method described above and purified by flash column chromatography (0 to 1% ethyl acetate in hexanes) in 49% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 8.11 – 7.98 (m, 2H), 7.97 – 7.83 (m, 1H), 7.78 (s, 1H), 7.56 – 7.34 (m, 7H), 6.63 (d, *J* = 16.1 Hz, 1H), 5.34 (s, 2H). ¹³C NMR (101 MHz, CDCl₃) δ 166.97, 140.54, 137.11, 136.92, 136.11, 131.61, 128.67, 128.40, 128.38, 128.34, 125.10, 125.01, 123.05, 122.15, 118.30, 77.43, 77.11, 76.79, 66.48. HRMS: m/z calcd for C₁₈H₁₄O₂S: 295.0793; found: 295.0791;



(E)-benzyl 3-(benzofuran-3-yl)acrylate (3.75e): The compound 3.75e was prepared according to the general method described above and purified by flash column chromatography (0 to 1% ethyl acetate in hexanes) in 35% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.95 – 7.81 (m, 2H), 7.61 – 7.31 (m, 7H), 6.66 (d, *J* = 16.1 Hz, 1H), 5.33 (s, 2H). ¹³C NMR (101 MHz, CDCl₃) δ 166.91, 156.14, 148.02, 136.11, 135.05, 128.66, 128.41, 128.34, 125.44, 124.75, 123.83, 121.06, 118.02, 117.89, 112.04, 77.45, 77.13, 76.81, 66.45. HRMS: m/z calcd for C₁₈H₁₄O₃: 279.1021; found: 279.1016;



(E)-tert-butyl 3-(1-methyl-1H-indol-2-yl)acrylate (3.73f): The compound 3.75f was prepared according to the general method described above and purified by flash column chromatography (0 to 1% ethyl acetate in hexanes) in 63% yield as a light-yellow oil. ¹H NMR (400 MHz, CDCl₃) δ 7.76 (d, *J* = 15.7 Hz, 1H), 7.66 (dt, *J* = 8.0, 1.0 Hz, 1H), 7.39 – 7.25 (m, 2H), 7.17 (ddd, *J* = 8.1, 6.3, 1.7 Hz, 1H), 6.97 (s, 1H), 6.49 (d, *J* = 15.8 Hz, 1H), 3.80 (s, 3H), 1.64 (s, 9H). ¹³C NMR (101 MHz, CDCl₃) δ 166.44, 138.97, 135.16, 131.73, 127.54, 123.42, 121.34, 120.39, 120.26, 109.63, 103.37, 80.63, 77.49, 77.17, 76.85, 29.98, 28.32. HRMS: m/z calcd for C₁₆H₁₉NO₂: 258.1494; found: 258.1494;

3.11. References

 (a) Jia, C.; Piao, D.; Oyamada, J.; Lu, W.; Kitamura, T.; Fujiwara, Y. Efficient Activation of Aromatic C-H Bonds for Addition to C-C Multiple Bonds. *Science*. 2000, 287, 1992-1995. (b) Jia, C.; Kitamura, T.; Fujiwara, Y. Catalytic Functionalization of Arenes and Alkanes via C–H Bond Activation. *Acc. Chem. Res.* 2001, 34, 633-639. (c) Nevado, C.; Echavarren, A. M. Transition Metal-Catalyzed Hydroarylation of Alkynes. *Synthesis*. 2005, 167-182. (d) Nakao, Y. Hydroarylation of Alkynes Catalyzed by Nickel. *Chem. Rec.* 2011, 11, 242-251. (e) Bandini, M. Gold-catalyzed Decorations of Arenes and Heteroarenes with C-C Multiple Bonds. *Chem. Soc. Rev.* 2011, 40, 1358-1367. (f) Mendoza, P. D.; Echavarren, A. M. Synthesis of Arenes and Heteroarenes by Hydroarylation Reactions Catalyzed by Electrophilic Metal Complexes. *Pure Appl. Chem.* 2010, 82, 801-820. (g) Kitamura, T. Transition-Metal-Catalyzed Hdroarylation Reactions of Alkynes Through Direct Functionalization of C-H Bonds. A Convenient Tool for Organic Synthesis. *Eur. J. Org. Chem.* **2009**, 1111-1125.

2. (a) Song, C. E.; Jun, D.-u.; Choung, S.-Y.; Roh, E. J.; Lee, S.-g. Dramatic Enhancement of Catalytic Activity in an Ionic Liquid: Highly Practical Friedel-Crafts Alkenylation of Arenes with Alkynes Catalyzed by Metal Triflates. Angew. Chem. Int. Ed. 2004, 43, 6183-6185. (b) Li, R.; Wang, S. R.; Lu, W. FeCl₃-Catalyzed Alkenylation of Simple Arenes with Aryl-Substituted Alkynes. Org. Lett. 2007, 9, 2219-2222. (c) Yoon, M. Y.; Kim, J. H.; Choi, D. S.; Shin, U. S.; Lee, J. Y.; Song, C. E. Metal Triflate-Catalyzed Regio- and Stereoselective Friedel–Crafts Alkenylation of Arenes with Alkynes in an Ionic Liquid: Scope and Mechanism. Adv. Synth. Catal. 2007, 349, 1725-1737. (d) Choi, D. S.; Kim, J. H.; Shin, U. S.; Deshmukh, R. R.; Song, C. E. Thermodynamically and Kinetically-Controlled Friedel-Crafts Alkenylation of Arenes with Alkynes Using an Acidic Fluoroantimonate(v) Ionic Liquid as Catalyst. Chem. Commun. 2007, 3482-3484. (e) Soriano, E.; Marco-Contelles, J. Mechanisms of the Transition Metal-Mediated Hydroarylation of Alkynes and Allenes. Organometallics 2006, 25, 4542-4553. (f) Tsuchimoto, T.; Maeda, T.; Shirakawa, E.; Kawakami, Y. Friedel-Crafts Alkenylation of Arenes Using Alkynes Catalysed by Metal Trifluoromethanesulfonates. Chem. Commun. 2000, 1573-1574. (g) Bandini, M.; Emer, E.; Tommasi, S.; Umani-Ronchi, A. Innovative Catalytic Protocols for the Ring-Closing Friedel–Crafts-Type Alkylation and Alkenylation of Arenes. Eur. J. Org. Chem. 2006, 2006, 3527-3544. (h) Yamamoto, Y.; Gridnev, I. D.; Patil, N. T.; Jin, T. Alkyne Activation with Bronsted Acids, Iodine, or Gold Complexes, and Its Fate Leading to Synthetic Application. Chem. Commun. 2009, 5075-5087.

- (a) Cacchi, S. The Palladium-Catalyzed Hydroarylation and Hydrovinylation of Carbon-Carbon Multiple Bonds: New Perspectives in Organic Synthesis. *Pure Appl. Chem.* 1990, 62, 713-722. (b) Cacchi, S.; Fabrizi, G.; Goggiamani, A.; Persiani, D. Palladium-Catalyzed Hydroarylation of Alkynes with Arenediazonium Salts. *Org. Lett.* 2008, *10*, 1597-1600. (c) Artok, L.; Kuş, M.; Aksın-Artok, Ö.; Dege, F. N.; Özkılınç, F. Y. Rhodium Catalyzed Reaction of Internal Alkynes with Organoborons under CO Atmosphere: a Product Tunable Reaction. *Tetrahedron.* 2009, *65*, 9125-9133. (d) Xu, X.; Chen, J.; Gao, W.; Wu, H.; Ding, J.; Su, W. Palladium-Catalyzed Hydroarylation of Alkynes with Arylboronic Acids. *Tetrahedron.* 2010, *66*, 2433-2438. (e) Liu, S.; Bai, Y.; Cao, X.; Xiao, F.; Deng, G.-J. Palladium-Catalyzed Desulfitative Hydroarylation of Alkynes with Sodium Sulfinates. *Chem. Commun.* 2013, *49*, 7501-7503.
- For leading and recent reviews on C-H bond activations, see: a) Activation and Functionalization of C-H Bonds (Ed.: A. S. Goldman, K. I. Goldberg), ACS Symposium Series 885, Washington DC, 2004; b) Handbook of C-H Transformations, Vol. 1 (Ed.: G. Dyker), Wiley-VCH, Weinheim, 2005; c) Chen, X.; Engle, K. M.; Wang, D.-H.; Yu, J.-Q. Palladium(II)-catalyzed C-H activation/C-C cross-coupling reactions: versatility and practicality. *Angew. Chem.* 2009, *121*, 5196-5217. *Angew. Chem. Int. Ed.* 2009, *48*, 5094-5115. d) Colby, D. A.; Bergman, R. G.; Ellman, J. A. Rhodium-Catalyzed C–C Bond Formation via Heteroatom-Directed C–H Bond Activation. *Chem. Rev.* 2010, *110*, 624-655. e) Mkhalid, I. A.; Barnard, J. H.; Marder, T. B.; Murphy, J. M.; Hartwig, J. F. C–H Activation for the Construction of C–B Bonds. *Chem. Rev.* 2010, *110*, 890-931. f) Lyons, T. W.; Sanford, M. S. Palladium-Catalyzed Ligand-Directed C–H Functionalization Reactions. *Chem. Rev.* 2010, *110*, 1147-1169. g) Wencel-Delord, J.; Roge, T.; Liu, F.;

Glorius, F. Towards Mild Metal-Catalyzed C–H Bond Activation. *Chem. Soc. Rev.* 2011, *40*, 4740-4761. h) Sun, C.-L.; Li, B.-J.; Shi, Z.-J.; *Chem. Rev.* 2011, *111*, 1293-1314. h)
Song, G.; Wang, F.; Li, X.; C–C, C–O and C–N Bond Formation *via* Rhodium(III)Catalyzed Oxidative C–H Activation. *Chem. Soc. Rev.* 2012, *41*, 3651-3678. i) Daugulis,
O.; Do, H.-Q. Shabashov, D.; Palladium- and Copper-Catalyzed Arylation of
Carbon–Hydrogen Bonds. *Acc. Chem. Res.* 2009, *42*, 1074-1086. j) Albrecht, M.;
Cyclometalation Using d-Block Transition Metals: Fundamental Aspects and Recent
Trends. *Chem. Rev.* 2010, *110*, 576-623.

- Murai, S.; Kakiuchi, F.; Sekine, S.; Tanaka, Y.; Kamatani, A.; Sonoda, M.; Chatani, N. Efficient Catalytic Addition of Aromatic Carbon-Hydrogen Bonds to Olefins. *Nature*.1993, *366*, 529-531.
- Kakiuchi, F., Yamamoto, Y., Chatani, N. & Murai, S. Catalytic Addition of Aromatic C-H Bonds to Acetylenes. *Chem. Lett.* 1995, 681-682.
- Kitamura, T. Transition-Metal-Catalyzed Hydroarylation Reactions of Alkynes Through Direct Functionalization of C–H Bonds: A Convenient Tool for Organic Synthesis. *Eur. J. Org. Chem.* 2009, 1111- 1125.
- Bandini, M. Gold-Catalysed Decorations of Arenes and Heteroarenes with C–C Multiple Bonds. *Chem. Soc. Rev.* 2011, 40, 1358-1367.
- Schipper, D. J., Hutchinson, M. & Fagnou, K. Rhodium(III)-Catalyzed Intermolecular Hydroarylation of Alkynes. *J. Am. Chem. Soc.* 2010, *132*, 6910-6911.
- Hashimoto, Y., Hirano, K., Satoh, T., Kakiuchi, F. & Miura, M. Ruthenium(II)-Catalyzed Regio- and Stereoselective Hydroarylation of Alkynes via Directed C–H Functionalization. *Org. Lett.* 2012, *14*, 2058-2061.

- Hashimoto, Y., Hirano, K., Satoh, T., Kakiuchi, F. & Miura, M. Regioselective C–H Bond Cleavage/Alkyne Insertion under Ruthenium Catalysis. *J. Org. Chem.* 2013, 78, 638-646.
- Min, M.; Kim, D.; Hong, S. AgSbF₆-Controlled Diastereodivergence in Alkyne
 Hydroarylation: Facile Access to Z- and E-Alkenyl Arenes. *Chem. Commun.* 2014, *50*, 8028-8031.
- Liang, L.; Fu, S.; Lin, D.; Zhang, Q.; Deng, Y.; Jiang, H.; Zeng, W. Ruthenium(II)-Catalyzed Direct Addition of Indole/Pyrrole C₂–H Bonds to Alkynes. *Journal of Organic Chemistry*. 2014 79, 9472-9480.
- Wong, M.; Yamakawa, T.; Yoshikai, N. Iron-Catalyzed Directed C₂-Alkylation and Alkenylation of Indole with Vinylarenes and Alkynes. *Org. Lett.* 2015, *17*, 3, 442-445.
- 15. (a) Fujiwara, Y.; Moritani, I.; Matsuda, M.; Teranishi, S. Aromatic Substitution of Olefin. IV Reaction with Palladium Metal and Silver Acetate. *Tetrahedron Lett.* 1968, *35*, 3863-3865. (b) Fujiwara, Y.; Moritani, I.; Danno, S.; Asano, R.; Teranishi, S. Aromatic substitution of olefins. VI. Arylation of olefins with palladium(II) acetate. *J. Am. Chem. Soc.* 1969, *91*, 7166-7169.
- Miura, M.; Tsuda, T.; Satoh, T.; Nomura, M. Palladium-Catalyzed Oxidative Cross-Coupling of 2-Phenylphenols with Alkenes. *Chem. Lett.* 1997, *26*, 1103-1104.
- (a) Miura, M.; Tsuda, T.; Satoh, T.; Pivsa-Art, S.; Nomura, M. Oxidative Cross-Coupling of N-(2' Phenyl phenyl) benzenesulfonamides or Benzoic and Naphthoic Acids with Alkenes Using a Palladium-Copper Catalyst System Under Air. *J. Org. Chem.* 1998, *63*, 5211-5215. (b) Boele, M. D. K.; van Strijdonck, G. P. F.; de Vries, A. H. M.; Kamer, P. C. J.; de Vries, J. G.; van Leeuwen, P. W. N. M. Selective Pd-Catalyzed Oxidative

Coupling of Anilides with Olefins through C-H Bond Activation at Room Temperature. *J. Am. Chem. Soc.* **2002**, *124*, 1586-1587. (c) Engle, K. M.; Wang, D.-H.; Yu, J.-Q. Ligand-Accelerated C-H Activation Reactions: Evidence for a Switch of Mechanism. *J. Am. Chem. Soc.* **2010**, *132*, 14137-14151. (d) Shi, B.-F.; Zhang, Y.-H.; Lam, J. K.; Wang, D.-H.; Yu, J.-Q. Pd(II)-Catalyzed Enantioselective C–H Olefination of Diphenylacetic Acids. *J. Am. Chem. Soc.* **2010**, *132*, 460-461. (e) Wang, D.-H.; Engle, K. M.; Shi, B.-F.; Yu, J.-Q. Ligand-Enabled Reactivity and Selectivity in a Synthetically Versatile Aryl C– H Olefination. *Science.* **2010**, *327*, 315-319. (f) Lu, Y.; Wang, D.-H.; Engle, K.-M.; Yu, J.-Q. Pd(II)-Catalyzed Hydroxyl-Directed C–H Olefination. Enabled by Monoprotected Amino Acid Ligands. *J. Am. Chem. Soc.* **2010**, *132*, 5916-5921.

- Bras, J. L.; Muzart, J. Intermolecular Dehydrogenative Heck Reactions. *Chem. Rev.* **2011**, *111*, 1170-1214.
- (a) Umeda, N.; Hirano, K.; Satoh, T.; Miura, M. Rhodium-Catalyzed Mono- and Divinylation of 1-Phenylpyrazoles and Related Compounds via Regioselective C–H Bond Cleavage. *J. Org. Chem.* 2009, *74*, 7094-7099. (b) Ueura, K.; Satoh, T.; Miura, M. An Efficient Waste-Free Oxidative Coupling via Regioselective C–H Bond Cleavage: Rh/Cu-Catalyzed Reaction of Benzoic Acids with Alkynes and Acrylates under Air. *Org. Lett.* 2007, *9*,1407-1409. (c) Guimond, N.; Fagnou, K. Isoquinoline Synthesis via Rhodium-Catalyzed Oxidative Cross-Coupling/Cyclization of Aryl Aldimines and Alkynes. *J. Am. Chem. Soc.* 2009, *131*, 12050-120501. (d) Guimond, N.; Gouliaras, C.; Fagnou, K. Rhodium(III)-Catalyzed Isoquinolone Synthesis: The N–O Bond as a Handle for C–N Bond Formation and Catalyst Turnover. *J. Am. Chem. Soc.* 2010, *132*, 6908-6909. (e) Stuart, D. R.; Alsabeh, P.; Kuhn, M.; Fagnou, K. Rhodium(III)-Catalyzed

Arene and Alkene C–H Bond Functionalization Leading to Indoles and Pyrroles. *J. Am. Chem. Soc.* 2010, *132*, 18326-18339. (f) Rakshit, S.; Patureau, F. W.; Glorius, F. Pyrrole
Synthesis via Allylic sp³ C–H Activation of Enamines Followed by Intermolecular
Coupling with Unactivated Alkynes. *J. Am. Chem. Soc.* 2010, *132*, 9585-9587. (g)
Patureau, F. W.; Wencel-Delord, J.; Glorius, F. Cp*Rh-Catalyzed C–H Activations:
Versatile Dehydrogenative Cross-Couplings of C_{sp2} C–H Positions with Olefins,
Alkynes, and Arenes. *Aldrichimica*. 2012, 45, 79062-79187.

- Ackermann, L.; Pospech, J. Ruthenium-Catalyzed Oxidative C–H Bond Alkenylations in Water: Expedient Synthesis of Annulated Lactones. *Org. Lett.* 2011, *13*, 4153-4155.
- (a) Reddy, M. C.; Jeganmohan M. Ruthenium-Catalyzed Cyclization of Aromatic Nitriles with Alkenes: Stereoselective Synthesis of (*Z*)-3-Methyleneisoindolin-1-ones. *Org. Lett.* 2014, *16*, 4866–4869. (b) Padala K.; Jeganmohan. M. Highly Regio- and Stereoselective Ruthenium(II)-Catalyzed Direct *ortho*-Alkenylation of Aromatic and Heteroaromatic Aldehydes with Activated Alkenes under Open Atmosphere. *Org. Lett.* 2012, *14*, 1134–1137. (c) Manoharan, R.; Jeganmohan, M. Ruthenium-Catalyzed C–H Amidation and Alkenylation of Cyclic *N*-Sulfonyl Ketimines. *Eur. J. Org. Chem.* 2016, 4013–4019. (d) Reddy, M. C.; Jeganmohan, M. Ruthenium-Catalyzed Selective Aerobic Oxidative *ortho*-Alkenylation of Substituted Phenols with Alkenes through C–H Bond Activation. *Eur. J. Org. Chem.* 2013, 1150–1157 (e) Padala K.; Pimparkar, S.; Madasamy.; P.; Jeganmohan. M. Ruthenium-catalyzed regioselective oxidative coupling of aromatic and heteroaromatic esters with alkenes under an open atmosphere. *Chem. Commun.* 2012, *48*, 7140-7142.

- Manikandan, R.; Madasamy, P.; Jeganmohan[,] M. Ruthenium-Catalyzed *ortho* Alkenylation of Aromatics with Alkenes at Room Temperature with Hydrogen Evolution. *ACS Catal.* 2016, *6*, 230–234.
- Ueyama, T.; Mochida, S.; Fukutani, T.; Hirano, K.; Satoh, T.; Miura, M.; Ruthenium-Catalyzed Oxidative Vinylation of Heteroarene Carboxylic Acids with Alkenes via Regioselective C-H Bond Cleavage. *Org. Lett.* 2011, *13*, 706-708.
- Dana, S.; Mandal, A.; Sahoo, H.; Mallik, S.; Grandhi,G. S.; Baidya, M.; Ru(II)-Catalyzed Oxidative Heck-Type Olefination of Aromatic Carboxylic Acids with Styrenes through Carboxylate-Assisted C–H Bond Activation. *Org. Lett.* 2018, 20, 716–719.
- Ishiyama, T.; Takagi, J.; Ishida, K.; Miyaura, N. Mild Iridium-Catalyzed Borylation of Arenes. High Turnover Numbers, Room Temperature Reactions, and Isolation of a Potential Intermediate. *J. Am. Chem. Soc.* 2002, *124*, 390-391.
- Phipps, R. J.; Gaunt, M. J. A Meta-Selective Copper-Catalyzed C–H Bond Arylation.
 Science. 2009, 323, 1593-1597.
- (a) Leow, D.; Li, G.; Mei, T.-S.; Yu, J.-Q. Activation of Remote *meta*-C-H Bonds Assisted by an End-on Template. *Nature*. 2012, *486*, 518-522. (b) J.-Q. Conformation-Induced Remote *meta*-C-H Activation of Amines. *Nature*. 2014, *507*, 215-220. (c) Xu, H.-J.; Lu, Y.; Farmer, M. E.; Wang, H.-W.; Zhao, D.; Kang, Y.-S.; Sun, W.-Y.; Yu, J.-Q. Rh(III)-Catalyzed *meta*-C–H Olefination Directed by a Nitrile Template. *J. Am. Chem. Soc.* 2017, *139*, 2200-2203. (d) Wan, L.; Dastbaravardeh, N.; Li, G.; Yu, J.-Q. Cross Coupling of Remote *meta*-C–H Bonds Directed by a U-Shaped Template. *J. Am. Chem. Soc.* 2013, *135*, 18056-18059.

- Bag, S.; Patra, T.; Modak, A.; Deb, A.; Maity, S.; Dutta, U.; Dey, A.; Kancherla, R.;
 Maji, A.; Hazra, A.; Bera, M.; Maiti, D.Remote *para*-C–H Functionalization of Arenes
 by a D-Shaped Biphenyl Template-Based Assembly. *J. Am. Chem. Soc.* 2015, *137*, 11888–11891.
- (a) Wang, X-C.; Gong, W.; Fang, L.; Zhu, R-Y.; Li, S.; Engle, K. M.; Yu, J. Q. Ligand-enabled *meta*-C–H activation using a transient mediator. *Nature*. 2015, *519*, 334-338. (b) Dong, Z.; Wang, J.; and Dong, G. Simple Amine-Directed *meta*-Selective C–H Arylation via Pd/Norbornene Catalysis. *J. Am. Chem. Soc.* 2015, *137*, 5887–5890.
- 30. (a) Gooßen, L. J.; Rodríguez, N.; Gooßen, K. Carboxylic Acids as Substrates in Homogeneous Catalysis. *Angew. Chem. Int. Ed.* 2008, *47*, 3100-3120; (b) Rodriguez, N.; Goossen, L. J. Decarboxylative Coupling Reactions: a Modern Strategy for C–C-bond Formation. *Chem. Soc. Rev.* 2011, *40*, 5030-5048; (c) Cornella, J.; Larrosa, I. Decarboxylative Carbon-Carbon Bond-Forming Transformations of (Hetero)aromatic Carboxylic Acids. *Synthesis* 2012, 653-676; (d) Drapeau, M. P.; Gooßen, L. J. Carboxylic Acids as Directing Groups for C–H Bond Functionalization. *Chem. Eur. J.* 2016, *22*, 18654–18677. e) Wei, Y.; Hu, P.; Zang, M.; Su. W.; Metal-Catalyzed Decarboxylative C–H Functionalization. *Chem. Rev.* 2017, *117*, 8864–8907.
- 31. (a) Chiong, H. A.; Pham, Q.-N.; Daugulis, O. Two Methods for Direct *ortho*-Arylation of Benzoic Acids. *J. Am. Chem. Soc.* 2007, *129*, 9879-9884; (b) Wang, D.-H.; Mei, T.-S.;
 Yu, J.-Q. Versatile Pd(II)-Catalyzed C–H Activation/Aryl–Aryl Coupling of Benzoic and Phenyl Acetic Acids. *J. Am. Chem. Soc.* 2008, *130*, 17676-17677; (c) Shi, B.-F.; Zhang,
 Y.-H.; Lam, J. K.; Wang, D.-H.; Yu, J.-Q. Pd(II)-Catalyzed Enantioselective C–H Olefination of Diphenylacetic Acids. *J. Am. Chem. Soc.* 2009, *132*, 460-461; (d)

Ackermann, L.; Pospech, J.; Potukuchi, H. K. Well-Defined Ruthenium(II) Carboxylate as Catalyst for Direct C–H/C–O Bond Arylations with Phenols in Water. *Org. Lett.* 2012, *14*, 2146-2149; (e) Ackermann, L.; Pospech, J. Ruthenium-Catalyzed Oxidative C–H Bond Alkenylations in Water: Expedient Synthesis of Annulated Lactones. *Org. Lett.* 2011, *13*, 4153-4155.

- Ueura, K.; Satoh, T.; Miura, M. Rhodium- and Iridium-Catalyzed Oxidative Coupling of Benzoic Acids with Alkynes via Regioselective C–H Bond Cleavage. *J. Org. Chem.* 2007, 72, 5362-5367.
- Maehara, A.; Tsurugi, H.; Satoh, T.; Miura, M. Regioselective C–H Functionalization Directed by a Removable Carboxyl Group. *Organic Letters*. 2008, *10*, 1159-1162.
- Mochida, S.; Hirano, K.; Satoh, T.; Miura, M. Synthesis of Stilbene and Distyrylbenzene Derivatives through Rhodium-Catalyzed *ortho*-Olefination and Decarboxylation of Benzoic Acids. *Org. Lett.* 2010, *12*, 5776-5779.
- 35. Cornella, J.; Righi, M.; Larrosa, I. Carboxylic Acids as Traceless Directing Groups for Formal *meta*-Selective Direct Arylation. *Angew. Chem. Int. Ed.* **2011**, *50*, 9429-9432.
- Luo, J.; Preciado, S.; Larrosa, I. Overriding *ortho–para* selectivity via a traceless directing group relay strategy: The *meta-selective arylation of Phenols. J. Am. Chem.* Soc. 2014, 136, 4109-4112.
- Qin, X.; Qiulin, S. Cheng, Y.; Lan, J.; and You, J. Rh(III)-Catalyzed
 Decarboxylative *ortho* Heteroarylation of Aromatic Carboxylic Acids by Using the
 Carboxylic Acid as a Traceless Directing Group. *Org. Lett.* 2015, *17*, 1762–1765.

- 38. Shi, X-Y.; Dong, X-F.; Fan, J.; Liu, K-Y.; Wei, J-F.; Li, C-J.; Ru(II)-catalyzed *ortho*amidation and decarboxylation of aromatic acids: a versatile route to *meta*-substituted Naryl benzamides. *Sci. China: Chem.* **2015**, *58*, 1286–1291.
- 39. Zhang, J.; Shrestha, R.; Hartwig, J. F.; Zhao, P. A decarboxylative approach for regioselective hydroarylation of alkynes. *Nature Chemistry*. **2016**, *8*, 1144-1151.
- 40. Rotem, M.; Shvo, Y. Addition of Carboxylic Acids to Alkynes Catalyzed by Ruthenium Complexes. Vinyl Ester Formation. *Organometallics*. **1983**, *2*, 1689-1691.
- 41. (a) Ackermann, L.; Pospech, J.; Graczyk, K.; Rauch, K. Versatile Synthesis of Isocoumarins and α-Pyrones by Ruthenium-Catalyzed Oxidative C–H/O–H Bond Cleavages. *Org. Lett.* 2012, *14*, 930-933. (b) Chinnagolla, R. K.; Jeganmohan, M. Regioselective Synthesis of Isocoumarins by Ruthenium-Catalyzed Aerobic Oxidative Cyclization of Aromatic Acids with Alkynes. *Chem. Commun.* 2012, *48*, 2030-2032.
- Mio, M. J.; Kopel, L. C.; Braun, J. B.; Gadzikwa, T. L.; Hull, K. L.; Brisbois, R. G.;
 Markworth, C. J.; and Grieco, P. A. One-Pot Synthesis of Symmetrical and
 Unsymmetrical Bisarylethynes by a Modification of the Sonogashira Coupling Reaction. *Org. Lett.* 2002, *4*, 3199.
- Li, J.; Ackermann, L. Cobalt-Catalyzed C-H Cyanation of Arenes and Heteroarenes.
 Angew. Chem. Int. Ed. 2014, 54, 3635.
- Rebstock, A.-S.; Mongin, F.; Trécourt, F.; Quéguiner, G. Synthesis and metallation of 2-(pyridyl)benzoic acids and ethyl 2-(pyridyl)benzoates: a new route to azafluorenones. *Tetrahedron.* 2003, 59, 4973.

4. CHELATION-ASSISTED DECARBOXYLATIVE ARENE AMIDATION

4.1. Introduction

Aryl amines are often found in many natural products, pharmaceutical agents, and functional materials. ¹ This leads to a significant interest and desire among researchers to develop efficient and convenient methods for the construction of arene carbon-nitrogen bond.. Ullman first revealed a protocol for the synthesis of aryl amines (**4.03**) using a copper catalyst (Scheme 4.1. a)², but the current protocol suffers from certain flaws such as harsh reaction conditions and poor substrate scope. In 1994, Buchwald and Hartwig independently reported a palladium catalyzed aryl amine synthesis using aryl halides (**4.01**) as coupling partners (Scheme 4.1.b).³





Though this transformation was significantly better in terms of the compatibility with the broad range of functional group groups and with various aryl halides, it has its own disadvantages in terms of the requirement of pre-functionalized starting materials and generation of stoichiometric amounts of metal halide waste. In contrast, direct C-H amination of arenes occurs with high step economy and reduced waste.⁴ The classical method for the synthesis of aryl amines involves electrophilic nitration followed by reductive hydrogenation (Scheme 4.2). The harsh conditions such as strong acidic conditions are limiting the scope and practicality of this protocol.⁵



Scheme 4.2. Classical nitration pathway.

Based on the mechanism, C–H amination of arenes can be classified into three categories: (i) C–H insertion catalysis (ii) Single-electron transfer catalysis (iii) C-H activation catalysis.⁶

In C-H insertion catalysis, construction of C-N bond occurs through the catalytic transfer of nitrene moiety. The key intermediate in this reaction is metal–nitrenoid (**4.09**), which is a product of the reaction occurring between aminating agent and metal center. The interaction of metal–nitrenoid (**4.09**) species with the hydrocarbon (**4.10**) results in C-N bond formation through C-H bond functionalization. The C-H bond functionalization can occur in two different mechanisms as shown in Scheme 4.3. First one is a concerted C–H insertion and the second is a stepwise process of hydrogen atom abstraction (HAA) followed by radical recombination.⁷



Scheme 4.3. C-H insertion catalysis.

Breslow and coworkers first demonstrated the catalytic nitrene transfer through the amidation of cyclohexane using cytochrome P-450.⁸ Later, the group of Mansuy described the nitrene transfer in the aziridination of olefins utilizing Fe and Mn porphyrin catalysts.⁹ These results motivated various groups thus, diverting their attention towards the construction of C-N bond through direct functionalization of C-H bonds via insertion mechanism. In the recent years, C-H insertion catalysis has been employed for the functionalization of arene C-H bonds to construct arene C-N bond. Driver et al. reported Rh(II)-catalyzed intramolecular $C(sp^2)$ –H amination for the synthesis of indole and carbazole derivative which are often found in many biologically active compounds and functional materials (Scheme 4.4). Authors proposed that the reaction proceeded through the formation of rhodium-nitrenoid intermediate followed by C-N bond formation through 4π -electron-5-atom electrocyclization (**4.21**) which was supported by the Hammett plot and intramolecular competition experiments.¹⁰.



Scheme 4.4. Intramolecular C-H amination via metal-nitrenoid intermediate.

Che and coworkers demonstrated the intermolecular C–H amination of various heterocycles using Ru(II) porphyrin Catalyst.¹¹ Recently Falck et al. revealed Rh₂(esp)₂ mediated mild protocol for the synthesis of aryl amines using N-methyl-O-tosylhydroxylamine (**4.23**) as the aminating agent (Scheme 4.5).¹² Authors proposed that the reaction proceeded through the formation of rhodium-imido (**4.25**) species followed by electrophilic addition of arenes (**4.21**) furnished aryl amine product (**4.24**). This particular transformation is simple, scalable and occurs with good yields. This strategy was further extended for the synthesis of various N-heterocycles through direct intramolecular cyclization (aza-annulation) of arenes, but the drawback was that it worked only with electron-rich arenes and often produced a mixture of Regio isomers.



Scheme 4.5. Rh(II) Catalyzed direct intermolecular C-H amination of arenes.

In the recent years, photoredox catalysis has emerged as a valuable tool for C-H amination of arenes via single electron transfer mechanism. Significant advancements were made for the generation of nitrogen-centered radicals through visible light photo redox catalysis. Upon irradiation of visible light, organo radical species are produced in the presence of photocatalyst, which reacts with the hydrocarbon to furnish the desired aryl amine as a product (Scheme 4.6).^{13,14}



Scheme 4.6. C-H amination via photoredox catalysis.

Yu et al. reported a protocol for the synthesis of various Aza heterocycles such as phenanthridines, quinolines and pyridines through intramolecular C-H amination using iridium photocatalysis (Scheme 4.7). This transformation proceeds at room temperature with broad substrate scope and functional group tolerance. The proposed mechanism is depicted in the Scheme 4.8.¹⁵



Scheme 4.7. Ir-catalysed intramolecular C-H amination via photoredox catalysis.

Upon irradiation with visible light, photocatalyst undergoes excitation and the excited species transfers a single electron to the N-acyl oxime (**4.29**) which produces nitrogen-centered radical (**4.31**), and acetate anion. The intermediate **4.31** undergoes homolytic aromatic substitution to generate radical **4.32** which is further oxidized to form cationic intermediate **4.33** and regenerate the photocatalyst. Finally, deprotonation of **4.33** delivers the final product (**4.30**).



Scheme 4.8. The mechanism for intramolecular catalyzed C-H amination using a photocatalyst.

In 2014, Sanford and coworkers reported an efficient protocol for the intermolecular C-H amination of arenes and heteroarenes using N-acyloxyphthalimides as aminating agent.¹⁶ This method enables the generation of nitrogen-based radical in the presence of photocatalyst through single electron transfer. This protocol occurs at room temperature with high functional group tolerance. Recently, the group of Konig established a new protocol for direct C-H amidation of heteroarenes using Ruthenium photocatalyst as depicted in Scheme 4.9.¹⁷

This protocol occurs under a mild condition and suitable for C-H amidation of various heteroarenes such as indole, furan, benzofuran, thiophene, and pyrrole. Unlike aforementioned single electron process, the present reaction occurs via energy transfer mechanism. Upon energy transfer from excited photocatalyst to benzoyl azide (**4.36**) produces benzoyl nitrene (**4.39**), which reacts with the heteroarene in the presence of an acid additive to furnish amidation product (**4.37**).



Scheme 4.9. Intermolecular C-H amination of heteroarenes via energy transfer mechanism.

Although significant advancements have been made for C-H amination through insertion catalysis or photoredox catalysis, it is still not viable for the practical applications. To counteract this, another important strategy for C-H amination is "C-H activation catalysis (Scheme 4.10)." In this mechanism, the interaction of metal catalyst with carbon-hydrogen bond of a hydrocarbon produces organometallic intermediate (C-M), which reacts with the aminating agent to furnish desired amination products. ¹⁸



Scheme 4.10. C-H amination through transition metal catalyzed C-H activation. The groups of Mori¹⁹ and Schreiber²⁰ independently developed new protocols for the C-H amination of heterocycles such as benzimidazoles, benzoxzole and benzothiazoles in the

presence of a copper catalyst using molecular oxygen as oxidant. The reaction proceeds via dehydrogenative coupling of heterocycles, and secondary amines and it occurs at the most acidic C-H bond. In 2011, the group of Miura established a redox neutral method using O-acylated hydroxyl amines (**4.45**) as an aminating agent which also acts as an internal oxidant and circumvents the need for eternal oxidant (Scheme 4.11).²¹



Scheme 4.11. Copper-catalyzed direct amination of electron-deficient arenes with hydroxylamines.

In addition to heterocycles, this transformation is also suitable for C-H amination of perfluoroarenes, and it works at room temperature. The proposed mechanism is depicted in the Scheme 4.12. The Cu(II) precatalyst produces Cu(I) intermediate vis disproportionation reaction. The Cu(I) active catalyst reacts with arene (**4.43**) to give Cu(I)-aryl intermediate (**4.47**) which undergoes oxidative addition with the aminating agent (**4.45**) and generates Cu(III) intermediate (**4.48**). Cu(III) intermediate upon reductive elimination produces the desired product and releases the active catalyst.



Scheme 4.12. The mechanism for copper-catalyzed intermolecular C-H amination. In 2013, Hartwig et al. reported a palladium catalyzed intermolecular C-H amination of arenes using an N-aryl phthalimides (4.51) aminating agent and Diacetoxyiodobenzene oxidant. Despite the broad substrate scope and high yields, the reaction occurs with less of regioselectivity and the reaction results in a mixture of regio isomers in most of the cases as shown in Scheme 4.13.²²



Scheme 4.13. Palladium-catalyzed intermolecular amination of arenes.

Achieving regioselectivity is a critical challenge in an intermolecular C-H amination. In this regard, chelation assisted C-H amination is an allusive strategy to achieve regioselectivity. In 2005, Buchwald revealed an efficient protocol for the synthesis of the carbazole (4.53) from acetaminobiphenyl (4.52) through intramolecular C-H amination (Scheme 4.14. b). This reaction proceeds via C-H bond activation followed by C-N formation.²³ The catalytic cycle starts with the pre-association of pd catalyst with amide to generate metal amido intermediate. Subsequently, Palladium undergoes heck type or wacker type addition on to the arene and generates corresponding Pd(II) intermediate, which upon β -hydride elimination delivers the carbazole product and Pd (0). Pd(II) catalyst can be regenerated by using copper oxidant. In 2008, Gaunt and coworkers revealed another protocol for the synthesis of carbazole using intramolecular C-H amination in the presence of palladium catalyst and PhI(OAc)₂ oxidant.²⁴ This reaction proceeds through Pd(II)/Pd(iv) catalytic system and occurs with high functional group tolerance and good yields. Later, Hartwig and co-workers established a new method for Pd (0) catalyzed Indole (4.55) synthesis via intramolecular C-H amination using N-acyl oxime (4.54) as an aminating agent and no external oxidant is required for this reaction (Scheme 4. 14. b).²⁵



Scheme 4.14. a) Pd(II)-catalyzed carbazole synthesis through oxidative C–H bond amination b) Palladium-catalyzed intramolecular C–H amination of aromatics with oxime esters.

After these seminal reports, many research groups developed various methods for the cross dehydrogenative coupling for the synthesis of azaheterocycles (Scheme 4.15). ^{26, 27, 28, 29}



Scheme 4.15. Pd catalyzed intramolecular C-H amination for the synthesis of heterocycles.

In 2007, Sanford and coworkers demonstrated the chelation assisted intermolecular C-H amination through the stoichiometric amination of palladacyclic complexes with PhI=NTs as shown in Scheme 4.16.³⁰ Authors proposed that the reaction proceeds via a stepwise mechanism which involves the formation of Pd^{IV} imido intermediate (**4.59**).

Stepwise reaction of PhI=NTs with PdII-C bond



Scheme 4.16. Amination of palladacycle with an iminoiodinane reagent.

Che and coworkers first revealed a Pd-catalyzed directed oxidative C-H amination of phenyl pyridine using a stoichiometric amount of $K_2S_2O_8$ oxidant.³¹ Later, Liu et al. accomplished Pd

catalyzed dehydrogenative coupling of aromatic ketones and amides for intermolecular C-N bond formation using N-fluoro-2, 4,6-trimethyl-pyridinium triflate as oxidant.³² Later in 2010, the group of Yu reported a redox-neutral method for intermolecular C-H amidation of anilides using N-Nosylcarbamates as aminating agent.³³ The reaction proceeded with good functional group tolerance and regioselectivity. This protocol enables the facile synthesis of 2aminoanilines which are essential scaffolds in many biologically active compounds. The usage of electrophilic aminating agents containing weak N-O bond is an attractive strategy as it occurs under redox neutral conditions. This strategy was extended for the intermolecular C-H amination of various directing groups through chelation assisted C-H bond activation (Scheme 4.17).³⁴



Scheme 4.17. Intermolecular C-H amination using electrophilic aminating agents.

The use of preactivated aminating agents in directed C-H amination also produces stoichiometric amounts of byproducts. In this regard, organic azides have been considered as efficient aminating agents for intermolecular C-H amination because of prominent features such as i) easy synthesis of azides ii) generation of N₂ as a sole byproduct in C-N bond forming reaction. iii) an internal oxidant which obviates the need for the external oxidant.



Scheme 4.18. Rhodium-catalyzed intermolecular C-H amination using tosyl azide as aminating source.

Chang and coworkers first reported a rhodium catalyzed protocol for the directed C-H amination of arenes using sulfonyl azide as an aminating source.³⁵ This protocol is suitable for broad range of arenes with good yield and high functional group tolerance. The mechanism of the reaction is described in the Scheme 4.18. The reaction proceeds via formation of rhodacycle **4.74** followed by addition tosyl azide coordination leads to the generation of intermediate **4.76**. Later, the formation of metal amido intermediate **4.78** can occur in two possible pathways: i)

through the stepwise formation of Rh(V) metal nitrenoid intermediate followed by insertion of a nitrenoid moiety to cyclometalate. ii) Through concerted mechanism. Depending on experimental and DFT calculations authors proposed that stepwise mechanism is more probable than concerted mechanism.³⁶ Finally, protonation of metal amido species releases the aminated product and regenerates the active catalyst.

Kanai and coworkers extend this strategy for the C-H amination of indole derivatives using tosyl azide as an aminating agent in the presence of airstable Co(III) catalyst (**4.19 a**).³⁷ Later in 2013 Chang and coworkers extended the directed C-H amination in the presence of iridium catalyst using acyl azides as aminating agent (**4.19 b**). The current transformation works under mild reaction conditions with excellent substrate scope.³⁸



Scheme 4.19. a) Cobalt-catalyzed directed C-H amidation of indole derivatives using organic azides b) Iridium-catalyzed directed C-H of amides using organic azides.

Diaryl amines are widely present in many pharmaceuticals, agrochemicals, and functional materials. Particularly diarylamines bearing *ortho* carbonyl group are vital components of many biologically active compounds. Classical methods for the synthesis of diarylamines produce stoichiometric amounts of waste. Chang et al. disclosed a method for the rhodium-catalyzed C-H amination of benzamides and imines using aryl azides as aminating agents to produces

diarylamines (Scheme 4.20). This protocol works without any external oxidant, and the only byproduct is nitrogen. ³⁹



Scheme 4.20. Rhodium-catalyzed diarylamine synthesis using aryl azides as an aminating reagent.

In 2015, Chang's group presented a new aminating agent 1,4,2-Dioxazol-5-one which enabled the C-H amination of arenes at room temperature. This protocol works without oxidant and releases carbon dioxide as a byproduct which makes the current transformation environmentally benign. The newly developed rhodium catalyzed C-H amination is highly efficient and compatible with a broad range of arenes. Authours proposed that the reaction proceed via formation of cyclometallated intermediate followed by coordination of aminating agent leading to the generation of metal-amido intermediate **4.91**. Finally, protonation of rhodium-amido intermediate produces the final product. ⁴⁰



Scheme 4.21. Rhodium-catalyzed direct C–H amidation with dioxazolones.
Later, Ackermann and many other groups used 1,4,2-Dioxazol-5-one as an aminating agent for the C-H amination of various directing groups such as oxazolines, ketones and pyrimidine.^{41, 42, 43}

In addition to C-H bond functionalization, directing group also has been employed for the C-C bond cleavage. Suggs and coworkers first demonstrated the C-C bond cleavage of ketones using directing group. In 1995, the group of Jun reported catalytic C-C bond cleavage through the assistance of 2-amino-3-picoline using rhodium(I) catalyst. The mechanism for the C-C bond cleavage is described in the Scheme 4.22. The transformation starts with the formation of ketimine **4.97**, followed by C-C bond cleavage via oxidative addition leading to the formation of **4.98**, which upon β -elimination produces rhodium hydride intermediate (**4.99**). Insertion of incoming alkene into Rh-H bond followed by reductive elimination of **4.100** delivers the final product and regenerates the catalyst through ligand association. ⁴⁴



Scheme 4.22. Rh(I) catalyzed C-C bond activation of ketimines.

This seminal work is the stepping stone for the advancement of C-C bond cleavage through the assistance of directing group. Shi and coworkers reported many methods for C-C bond cleavage using pyridyl directing group. In the initial report, authors reported C-C bond cleavage of diarylmethanols (**4.102**) for the synthesis of alkenyl arenes and biaryl derivatives (Scheme 4.23 a).⁴⁵ In 2012, the same group disclosed a new protocol for the rhodium catalyzed decarbonylative biaryl synthesis through directed C-C bond cleavage (Scheme 4.23 b).⁴⁶ These reactions proceed via rhodium catalyzed beta carbon elimination through the assistance of pyridyl directing group followed by formation of a five-membered rhodacycle intermediate which undergoes alkenylation and biaryl coupling.



Scheme 4.23. a) Rh-catalyzed alkenylation through directed C-C bond cleavage. b) Rh-catalysed decarbonylative biaryl coupling through directed C-C bond activation.

Recently Dong et al. used the same strategy for the synthesis of 2-quinolinones (**4.108**) through the coupling of isatins (**4.106**) and alkynes (**4.107**). The transformation proceeds via rhodium catalyzed directed C-C bond cleavage followed by decarbonylation and then insertion of alkynes. This reaction enables the straightforward synthesis of quinolone derivatives, and is compatible with broad range of isatin derivatives and alkynes (Scheme 4.24 a). This concept was also extended for the synthesis of benzimidazolidinones using isocyanates.⁴⁷ In 2016, Dong

disclosed a method for the selective cleavage of a C-C bond in cyclopentanone (**4.109**) using 2aminopyridine as a transient directing group (Scheme 4.24 b).⁴⁸



Scheme 4.24. a) Rh-Catalyzed decarbonylative coupling of alkynes and isatins for the synthesis of 2-quinolinones b) Rh-catalysed C-C bond activation of cyclopentanones using transient directing group.

Chatani and co-workers demonstrated the decarbonylative biaryl coupling of ketones using stoichometric amounts of the Ni catalyst which occurs through the C-C bond cleavage.⁴⁹ Recently, the group of Wei established a new protocol for Nickel catalyzed directed C-C bond cleavage using pyrimidine directing group (Scheme 4.25). This reaction is an alternative strategy for the synthesis of the 2-aryl indoles (**4.113**) and occurs with broad substrate scope and functional group tolerance.⁵⁰



Scheme 4.25. Ni-Catalysed decarbonylative biaryl coupling through directed C-C Bond. Cleavage.

In this chapter we describe Rh(III)/Ag(I) catalysed chelation assisted decarboxylative amidation of *ortho*-substituted benzoic acids using 3-aryldioxazolones as a coupling partner. The regioselecvity for amidation is regulated by the *ortho* substituent on the benzoic acid.

4.2. Initial Results

We commenced our study by taking *ortho* methoxy benzoic acid (**4.114 a**) and tosyl azide (**4.115 a**) as coupling Partners. After screening various Ruthenium, Iridium, Cobalt and Rhodium catalysts, we were delighted to observe the desired amidation product in 21% yield at 140 °C. In oder to improve efficiency of the reaction, we screened various additives, solvents and the best yield is obtained with 5 mol% [RhCl₂(Cp^{*})]₂, 20 mol% AgSbF₆ and 50 mol% Ag₂CO₃ in DMAC solvent at 140 °C.

OMe O		Catalyst, Additive	NHTs 	
HOH +	TsN 3	Temperature (°C)	\rightarrow	
		Solvent	OMe	
4.114 a	4.115 a		4.116	
Catalyst (mol%)	Additive (X mo	ol%) Temp(^O C)	Solvent Co	onversion(%) ^a
$[\operatorname{RuCl}_2(p\text{-cymene})]_2 (5)$	NaOAc (20)	120	Dioxane	0
$[\operatorname{RuCl}_2(p\text{-cymene})]_2 (5)$	Pivalic acid (5)	0) 120	Dioxane	0
$[Ru(OAc)_2(p-cymene)]$ (10)) Pivalic acid (5)	0) 120	Dioxane	0
$[Ru(OAc)_2(p-cymene)]$ (10)) $Ag_2CO_3(50)$	120	Dioxane	Traces
$[Ru(OAc)_2(p-cymene)]$ (10)) –	120	Dioxane	Traces
$[IrCl_2Cp^*]_2(5)$	$AgSbF_{6}(20)$	120	Dioxane	0
$[IrCl_2Cp^*]_2(5)$	$\operatorname{AgSbF}_{6}(20) + \operatorname{Ag}_{2}C$	CO ₃ (50) 120	Dioxane	0
$[IrCl_2Cp^*]_2(5)$	$\operatorname{AgSbF}_{6}(20) + \operatorname{Ag}_{2}C$	CO ₃ (50) 140	Dioxane	Traces
$[RhCl_2Cp^*]_2(5)$	$AgSbF_6(20)$	120	Dioxane	0
$[RhCl_2Cp^*]_2(5)$	$\operatorname{AgSbF}_{6}(20) + \operatorname{Ag}_{2}C$	CO ₃ (50) 120	Dioxane	Traces
$[RhCl_2Cp^*]_2(5)$	$AgSbF_6 (20) + Ag_2 C$	CO ₃ (50) 140	DMF	21
$[RhCl_2Cp^*]_2(5)$	$AgSbF_6(20) + Ag_2C$	CO ₃ (50) 150	DMF	17
$[RhCl_2Cp^*]_2(5)$	$AgSbF_6 (20) + Ag_2 C$	CO ₃ (50) 140	DMAC	23
$[RhCl_2Cp^*]_2(5)$	$\operatorname{AgSbF}_{6}(20) + \operatorname{Ag}_{2}C$	CO ₃ (50) 140	NMP	19
$[RhCl_2Cp^*]_2(5)$	$\operatorname{AgSbF}_{6}(20) + \operatorname{Ag}_{2}C$	CO ₃ (50) 140	DMSO	13
$[RhCl_2Cp^*]_2(5)$	$AgSbF_6(20) + Ag_2$	O (50) 140	DMAc	11
$[RhCl_2Cp^*]_2(5)$	$AgSbF_{6}(20) + Cu_{2}$	O (50) 140	DMAc	14
$[RhCl_2Cp^*]_2(5)$	$AgSbF_6(20) + Cuc$	O (50) 140	DMAc	Traces

Table 4.1. Optimization of rhodium catalyzed decarboxylative amidation (Part I).

Reaction conditions: **1** (0.15 mmol, 1 equiv), **2** (0.30 mmol, 2 equiv), Solvent (1 mL); a: GC yields

4.3. Optimization of Reaction Condition

At this stage we decided to examine various aminating agents (4.115) as shown in Table

2. On the basis of initial result we screened the aminating agents using Rh(III)/Ag(I) catalytic

system. The reaction efficiency is dramatically decreased with **4.15** b-**4.15** e aminating agents. On the other hand reaction proceeded smoothly with 3-phenldioxazolones (**4.15** f) and the corresponding product was obtained in 42% yield. Screening of various solvents reavealed dioxane as optimal solvent. Screening of various bases and other silver or copeer salts did not improve the reaction efficiency. Pleasigly, switching the equivalents of the reactants significantly improved the yield and the desired amidation product is obtained in 76% isolated yield.

Table 4.2. Optimization of rhodium catalyzed decarboxylative amidation (Part II).

R O 4.114 a	OH ₊ Amir	nating agent 4.115	Catalyst, Ad Temperatur Solvent	dditive e (°C) ►	NHR ₁ , , , , , , , , , , , , , , , , , , ,
Aminating TsN agent	N ₃ O N ₃ Ph	$>_{0}^{0}$	N-O II Ph OS=O	Ph O	Ph O
a	b	с	d	e	f
Catalyst (mol%	6) Additive	Amin agent	ating t Temp (°C) Solvent	Convsn(%) ^a
[RhCl ₂ Cp [*]] ₂	$AgSbF_6 + Ag$	g ₂ CO ₃ 4.1 1	15 a 140	DMAC	23
[RhCl ₂ Cp [*]] ₂	$AgSbF_6 + Ag$	g ₂ CO ₃ 4.1	15 b 140	DMAC	11
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag$	$g_2 CO_3$ 4.1	15 c 140	DMAC	0
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag$		15 d 140	DMAC	0
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag$	g ₂ CO ₃ 4.1	15 e 140	DMAC	<10
[RhCl ₂ Cp [*]] ₂	$AgSbF_6 + Ag$	g ₂ CO ₃ 4.1 1	1 5 f 140	DMAC	42

Reaction conditions: 1 (0.15 mmol, 1 equiv), 2 (0.225 mmol, 1.5 equiv), Solvent (1 mL), Catalyst (5 mol%), AgSbF₆ (20 mol%), Ag₂CO₃ (50 mol%), a: GC yields

R O	N-0	Catalyst,	Additive	NHCOPh
OH +		Temperat	ure (°C)	
	Ph O	Solve	nt	R
4.114	4.115 f			4.117
Catalyst (mol%)	Additive (mol%)	Temp (°C)	Solvent	Convsn(%)
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$ + K_2HPO_4	140	DMAC	0
$[RhCl_2Cp^*]_2$	AgSbF ₆	140	DMAC	0
-	$AgSbF_6 + Ag_2CO_3$	140	DMAC	0
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$	140	Dioxane	47
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$	140	Toluene	13
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$	140	Xylene	15
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$	140	Acetonitrile	8
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$	140	DCE	31
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$	140	NMP	41
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3$	140	DMF	38
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3(5)$	0) 140	DME	35
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3(5)$	0) 140	DMSO	26
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3(5)$	0) 140	Dimethyl carb	onate 43
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3(5)$	0) 140	Diethyl carbor	nate 36
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3(5)$	0) 140	Dioxane + DM	1AC 49
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2O(50)$	140	Dioxane + DM	1AC 11
$[RhCl_2Cp^*]_2$	$AgSbF_6 + CuO(50)$	140	Dioxane + DM	1AC 22
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Cu_2O(50)$	140	Dioxane + DM	IAC 9
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Cu(OAc)_2$	(50) 140	Dioxane + DN	IAC Traces
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3(5)$	0) 140	Dioxane + DN	IAC 83 ^b
$[RhCl_2Cp^*]_2$	$AgSbF_6 + Ag_2CO_3(50)$	0) 140	Dioxane	86 ^b (77) ^c

 Table 4.3. Optimization of rhodium catalyzed decarboxylative amidation (Part III).

_

Reaction conditions: 1 (0.15 mmol, 1 equiv), 2 (0.225 mmol, 1.5 equiv), Solvent (1 mL), Catalyst (5 mol%), AgSbF₆ (20 mol%), a: GC yields, b: 1 (2 equiv) & 2 (1 equiv), C: Isolated yield;

4.4. Substrate Scope

With optimized conditions in hand we evaluated the substrate scope for the decarboxylative amidation of benzoic acids as shown in Figure 4.1. Under established conditions, broad range of *ortho* substituted benzoic acids (**4.114**) reacted smoothly with phenyl dioxazolone (**4.115** f) to afford the corresponding *meta* amidation products.



Figure 4.1. Substrate scope for the decarbocxylative C-H amidation through C-H bond activation.

ortho benzoic acids bearing methoxy, fluoro, chloro, trifluoromethoxy and acetyl substrates were smoothely transformed into corresponding amidtion products (**4.117a - 4.117e**)

with satisfying yields from 77% to 42%. Next, various disubstituted benzoic acids were subjected to decarboxylative amidation and products (**4.117f** – **4.117l**) are afforded in good yields. The reactivity of trisubstituted and tetra substituted benzoic acids significatly diminished. The reaction of trifluoro benzoic acid and tetrafluoro benzoic acid with phenyl dioxazolone gave 33% and 45% yields respectively (**4.117m** and **4.117n**). Next we compared the reactivity of dioxazolones by raecting with *ortho* methoxy benzoic acid. 3-(4-Fluoro phenyl)-1,4,2-dioxazolone and the corresponding products are obtained in 74% and 63% yields (**4.117o** & **4.117p**).



Scheme 4.26. Plausible reaction mechanism for decarboxylative *ortho* amidation of arene carboxylic acids through chelation assisted C-H bond activation.

The current catalytic system proceeds via Rh(III) catalysed carboxylate-directed C-H amidation followed by Ag(I) catalysed decarboxylation. This protocol is compatible with broad range of arene carboxylic acids with good functional group tolerance. The plausible mechanism for this transformation is depicted in the Scheme **4.26**. The catalytic cycle begins with the formation of silver arylcarboxylate (**A**) through the reaction of arylcarboxylic acid with silvercarbonate. Upon reaction of Rh(III) catalyst with **A** generates the cylometalated intermediate **B** through carboxylate directed C-H bond activation. Next, reaction of dioxazolone with cyclometalated intermediate **B** results in the formation of metal-amido intermediate **D**. Protonation of intermediate **D** regenerates Rh(III) catalyst and produces silver arylcarboxylate intermediate **E**. Decarboxylation of **E** follwed by protonation of **F** delivers the final product.

Surprisingly, when Rhodium/Ag catalytic system is employed to the directing group substituted benzoic acids led to the formation of *ipso*-amidation products. Unlike abovementioned amidation which occurs through carboxyl directed C-H amidation followed by decarboxylation, this reaction proceeds via DG-assisted decarboxylation and subsequent amidation. Encouraged by this initial result we investigated the best reaction condition for the decarboxylative *ipso* amidation. After having established condition inhand we examined the scope of the reaction as shown in Figure 4.2.



Reaction conditions: 1 (0.18 mmol, 1.2 equiv) & 2 (0.15 mmol 1 equiv)



As shown above in Figure 4.2, the reaction is suitable for pyridyl, pyrazole and benzo[*h*]quinoline substituted benzoic acids and the corresponding products are obtained in excellent yields (**4.119a** to **4.119c**). The present reaction is also compatible for N-pyridyl indole-2-carboxylic acid for chelation assisted decarboxylative *ipso* amidation with 66% yield. Alteration of electronics on the phenyl ring affected the catalytic activity of the reaction and the substrate bearing electron with-drawing group on the phenyl ring in less reactive compared to the substrates bearing electron rich substituents, and corresponding products are afforded in good yields (**4.119e** to **4.119f**). Subsequently, variation of electronics on dioxazolone was examined and significant loss of reactivity was observed with 4-methoxyphenyl dioxazolone.

To further gain insight into the mechanism, we performed control experiments as shown in Scheme 4.27 . When the present reaction is performed using cationic Rh(III) catalyst, the reaction proceeded smoothly and afforded the desired product in 91% yield (Scheme 4.27 a). Next, we carried out decarboxylative *ipso* amidation using cyclometalated Rh(III) complex as catalyst, we obtained the corresponding product in 93% yield (Scheme 4.27 b). These experiments revealed that the reaction proceeds via Rh(III) mediated *ipso* decarboxylation followed by amidation.



Scheme 4.27. Control experiments for Rh(III) catalysed *ipso* amidation of arenecarboxylic acids through chelation assisted C-C bond activation.

From the above observations, the plausible pathway for *ipso* amidation is proposed in the following Scheme 4.28. Active catalyst generated in the reaction mixture reacts with arene carboxylic acid and generates Rh(III) carboxylate intermediate **A.** Coordination of Dioxazolidinone followed by decarboxylation produces cyclometalated Rh(III) intermediate **C**. Subsequently Rhodium-amido species (**D**) is generated from intermediate **C**, which undergoes protonation with the incoming substrate and delivers the final product.



Scheme 4.28. Mechanism for Rh(III) catalysed chelation assisted *ipso* amidation. In conclusion we disclosed a new protocol for Rh(III)/Ag catalysed amidation of arene carboxylic acids using aryl dioxazolones. The reaction proceeds in two different pathways depending on the *ortho* substituents of arene carboxylic acids. With nondirecting substituents the reaction proceeds via Rhodium catalysed chelation assisted C-H amidation followed by silver catalysed decarboxylation. On the otherhand with directing group substituents the reaction occurs via Rhodium catalysed chelation assisted decarboxylation followed by amidation. The current reaction proceeds with good yields and high functional group tolerance.

4.5. Experimental Procedures

4.5.1. General Experimental Procedures and Reagent Availability

Unless otherwise noted, all manipulations were carried out under a nitrogen atmosphere using standard Schlenk-line or glove box techniques. All glasswares were oven-dried for at least 1 h prior to use. Toluene and hexane solvents were degassed by purging with nitrogen for 45 min and dried with a solvent purification system (MBraun MB-SPS). THF and dichloromethane were dried over activated 3Å molecular sieves and degassed by purging with nitrogen. Other reagents and starting materials for substrate synthesis were purchased from commercial vendors and used as received. TLC plates were visualized by exposure to ultraviolet light.

Organic solutions were concentrated by rotary evaporation at ~10 torr. Flash column chromatography was performed with 40–63 microns silica gel. ¹H NMR and ¹³C NMR spectra were obtained on a 400 MHz spectrometer, and chemical shifts were recorded relative to residual protiated solvent. Both ¹H and ¹³C NMR chemical shifts were reported in parts per million downfield from tetramethylsilane ($\delta = 0$ ppm). ¹⁹F NMR spectra were obtained at 282.4 MHz, and all chemical shifts were reported in parts per million upfield of CF₃COOH ($\delta = -78.5$ ppm). High resolution mass spectra were obtained at a Waters HRMS spectrometer.

4.5.2. Synthesis of Reagents for C-H Amidation

4.5.2.1. Preparation of TsN_3^{51}





Into a 100 mL round-bottom flask equipped with a magnetic stir-bar was added solution of sodium azide (30 mmol) in water (20 mL). To the above solution, *p*-toluene sulfonyl chloride (20 mmol) in acetone (40 mL) was added dropwise at 0 °C. The reaction mixture was heated to room temperature and stirred for overnight. After removing the acetone under reduced pressure, the mixture was extracted with ethylacetate (40 mL). The combined organic layers were washed with 5% Na₂CO₃ (2 × 50 mL), water (2 × 50 mL) and then dried over MgSO₄. The solvent was removed under reduced pressure to afford tosyl azide (14 mmol) as a colorless oil.

4.5.2.2. Preparation of 5,5-dimethyl-3-phenyl-1,4,2-dioxazole⁵²



Scheme 4.30. Synthesis of 5,5-dimethyl-3-phenyl-1,4,2-dioxazole.

Procedure: Into a 250mL round bottom flask equipped with magnetic stir-bar was added a solution of benzohydroxamic acid (1.4 g, 10 mmol) and 2,2-dimethylpropane (3.1g, 30 mmol) in 150 mL of dichloromethane. To the above solution camphorsulfonic acid (2.3 g, 10 mmol) was added and stirred at room temperature for 3 h. After three hours, the reaction mixture was quenched with sodium bicarbonate (20 mL) and the aqueous layer was extracted with diethyl ether (100 mL x 3). The combined organic layers were dried over magnesium sulfate and concentrated under reduced pressure. The crude mixture was purified by column chromatography (hexane/EtOAc) to obtain the desired product.



Scheme 4.31. Synthesis of 5-phenyl-1,3,2,4-dioxathiazole 2-oxide.

Into a 250 mL round bottom flask was added a solution of thionyl chloride (40.0 mmol) in dichloromethane (40 mL). To the above solution benzohydroxamic acid (1.4 g, 10 mmol) was added at room temperature and stirred for 6 h. After six hours excessive thionyl chloride and solvent were removed under reduced pressure to give 5-phenyl-1,3,2,4-dioxathiazole 2-oxide (81%) as light yellow liquid.

4.5.2.4. Preparation of 3-substituted-1,4,2-dioxazol-5-one⁵⁴



Scheme 4.32. Synthesis of 3-substituted-1,4,2-dioxazol-5-one.

Into a 250mL round bottom flask was added a solution of hydoxamic acid (10.0 mmol) in dichloromethane (150 mL). To the above solution 1,1'-carbonyldiimidazole (1.62 g,10.0 mmol) was added in one portion at room temperature. After stirring the solution for 30 min, the reaction mixture was quenched with 1 N HCl (100 mL) and extracted with dichloromethane three times (100 mL x 3) and dried over magnesium sulfate. The solvent was removed under reduced pressure and product was recrystallized with acetone/hexane, to afford 3-substituted 1,4,2-dioxazol-5-ones.

$$H_2N_{N_{H}} \xrightarrow{0} + NaNO_2 \xrightarrow{CH_3COOH} N_{N_{N}} \xrightarrow{0} 0$$

Scheme 4.33. Synthesis of *tert*-Butyloxycarbonylazide.

Procedure: Into a 25mL round bottom flask was added a solution of tert-Butylcarbazate (3.00 g, 22.7 mmol, 1.0 eq.) in a mixture of water (7.0 mL) and acetic acid (5.0 mL). To the above solution NaNO₂ (1.72 g, 25.0 mmol, 1.1 eq.) was added at 0 °C over 5 minutes and the resulting mixture was stirred for two hours. After two hours, the reaction mixture was brought to room temperature and then 20 mL Water was added and the mixture was extracted with Et₂O (3×70 mL). The organic layer was washed with half saturated NaHCO₃(2×50 mL), brine (30 mL) and then dried over Na₂SO₄ and filtered. The solvent was removed in vacuum at room temperature. The BocN₃ (3.25 g, 22.7 mmol, quant.) was obtained as a yellow liquid and was pure enough for further applications.

4.5.2.6. Preparation of Benzoyl azide⁵⁶



Scheme 4.34. Synthesis of benzoyl azide.

Procedure: Into a 100mL round bottom flask was added a solution of sodium azide (1.99 g, 30 mmol) in water (10 mL). The solution was cooled to 0 °C and then acyl chloride (20 mmol) in acetone (20 mL) was added dropwise. The resulting mixture mixture was warmed up to room temperature and stirred for overnight. Acetone was removed under reduced pressure and the reaction mixture was extracted with EtOAc (50 mL x 3). The combined organic layers were

dried over MgSO₄and solvent was removed under reduced pressure. The crude mixture was purified using silica gel column chromatography (n-Hexane/EtOAc).

4.5.3. General Procedure for Decarboxylative *ortho* Amidation of Arenecarboxylic Acids through Chelation Assisted C-H Bond Activation

Into a 4 mL scintillation vial equipped with a magnetic stir bar was placed $[RhCl_2Cp^*]_2$ (5 mol%), AgSbF₆ (20 mol%), Ag₂CO₃ (50 mol%), phenyl-1,4,2-dioxazolone (0.15 mmol). arenecarboxylicacid (0.30 mmol) and Dioxane (1.0 mL). The reaction mixture was stirred at room temperature for 5 minutes and the vial was sealed with a silicone-lined screw-cap and electrical tape before being transferred out of the glovebox and stirred at 140 °C for 36 hours. The reaction mixture was cooled to room temperature and volatiles were removed under reduced pressure. The crude product was purified by flash-column chromatography on silica.

4.5.4. General Procedure for Decarboxylative *ipso* Amidation of Arenecarboxylic Acids through Chelation Assisted C-C Bond Activation

Into a 4 mL scintillation vial equipped with a magnetic stir bar was placed $[RhCl_2Cp^*]_2$ (2.5 mol%) and AgSbF₆(10 mol%). Phenyl-1,4,2-dioxazolone (0.15 mmol), arenecarboxylicacid (0.30 mmol) and Dioxane (1.0 mL) were subsequently added. The reaction mixture was stirred at room temperature for 5 minutes and the vial was sealed with a siliconelined screw-cap, and electrical tape before being transferred into the preheated oil bath. After stirring the reaction mixture at 140 °C for 36 hours, it was cooled to room temperature and volatiles were removed under reduced pressure to afford crude product. The crude product was purified by flash-column chromatography on silica.

4.6. Spectral Data



N-(3-methoxyphenyl)benzamide (4.117a): The compound **4.117a** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 7% ethyl acetate in hexanes) in 77% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.01 – 7.82 (m, 3H), 7.52 (dt, *J* = 31.6, 7.8 Hz, 4H), 7.31 – 7.23 (m, 1H), 7.13 (d, *J* = 8.0 Hz, 1H), 6.73 (ddd, *J* = 8.2, 2.4, 0.9 Hz, 1H), 3.84 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 165.8, 160.2, 139.1, 134.9, 131.8, 129.7, 128.7, 127.0, 112.3, 110.5, 105.8, 77.3, 77.0, 76.7, 55.3. HRMS: m/z calcd for C₁₄H₁₃NO₂: 228.1024; found: 228.1037.



N-(3-fluorophenyl)benzamide (4.117b):The compound **4.117b** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 4 % ethyl acetate in hexanes) in 65% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.17 – 7.97 (m, 1H), 7.94 – 7.77 (m, 2H), 7.69 – 7.42 (m, 4H), 7.36 – 7.23 (m, 2H), 6.87 (ddt, J = 7.9, 6.7, 2.5 Hz, 1H). ¹³C NMR (101 MHz CDCl₃) δ 165.9, 163.0 (d, J = 244.8 Hz), 139.4 (d, J = 10.9 Hz), 134.6, 132.0, 130.1 (d, J = 9.4 Hz), 128.8, 127.0, 115.4 (d, J = 3.0 Hz), 111.2 (d, J = 21.3 Hz), 107.7 (d, J = 26.4 Hz), 77.3, 77.0, 76.7. ¹⁹F NMR (376 MHz, CDCl₃) δ -111.31. HRMS: m/z calcd for C₁₃H₁₀FNO: 216.0825; found: 216.0819.



N-(3-chlorophenyl)benzamide(4.117c): The compound **4.117c** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 4 % ethyl acetate in hexanes) in 71 % yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.22 (s, 1H), 7.89 – 7.74 (m, 3H), 7.59 – 7.39 (m, 4H), 7.26 (t, *J* = 8.1 Hz, 1H), 7.12 (ddd, *J* = 8.0, 1.9, 0.9 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 166.1, 139.1, 134.6, 134.4, 132.0, 130.0, 128.7, 127.1, 124.6, 120.5, 118.4, 77.3, 77.0, 76.7. HRMS: m/z calcd for C₁₃H₁₀NOCl:232.0529; found: 232.0536.



N-(3-(trifluoromethoxy)phenyl)benzamide (4.117d): The compound **4.117d** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 4% ethyl acetate in hexanes) in 52% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.23 (s, 1H), 7.95 – 7.79 (m, 2H), 7.78 – 7.64 (m, 1H), 7.61 – 7.23 (m, 5H), 7.02 (dd, J = 8.1, 2.3 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 166.0, 149.6 (q, J = , 2.0 Hz), 139.3, 134.4, 132.1, 130.0, 128.8, 127.0, 121.7, 119.1, 118.3, 116.6, 113.1, 77.3, 77.0, 76.7. ¹⁹F NMR (376 MHz, CDCl₃) δ -57.7. HRMS: m/z calcd for C₁₄H₁₀NO₂F₃:282.0742; found: 282.0755.



N-(2-acetylphenyl)benzamide (4.117e): The compound **4.117e** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 10 % ethyl acetate in hexanes) in 42% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.69 (s, 1H), 8.22 (t, *J* = 1.9 Hz, 1H), 8.12 (ddd, *J* = 8.2, 2.2, 1.0 Hz, 1H), 7.98 – 7.86 (m, 2H), 7.70 (ddd, *J* = 7.8, 1.7, 1.0 Hz, 1H), 7.60 – 7.50 (m, 1H), 7.45 (td, *J* = 7.9, 6.5 Hz, 3H), 2.57 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 198.3, 166.2, 138.7, 137.6, 134.5, 132.0, 129.3, 128.7, 127.2, 125.1, 124.3, 119.8, 77.3, 77.0, 76.7, 26.6. HRMS: m/z calcd for C₁₅H₁₃NO₂: 240.1024; found: 240.1037.



N-(3-fluoro-5-methoxyphenyl)benzamide (4.117f): The compound **4.117f** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 72% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.37 (s, 1H), 7.88 – 7.75 (m, 2H), 7.56 – 7.46 (m, 1H), 7.46 – 7.35 (m, 2H), 7.16 – 7.03 (m, 2H), 6.41 (dt, *J* = 10.5, 2.3 Hz, 1H), 3.75 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 166.2, 163.6 (d, *J* = 242.9 Hz), 161.1 (d, *J* = 13.0 Hz), 139.9 (d, *J* = 13.7 Hz), 134.5, 132.0, 128.7, 127.1, 101.7 (d, *J* = 2.9 Hz), 100.1 (d, *J* = 26.6 Hz), 98.0 (d, *J* = 25.2 Hz), 77.4, 77.0,

76.7 , 55.5. ^{19}F NMR (376 MHz, CDCl₃) δ -110.6. HRMS: m/z calcd for C14H12NO2F: 246.0930; found: 246.0938.



N-(3-chloro-5-methoxyphenyl)benzamide (4.117g): The compound **4.117g** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 79% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.18 (s, 1H), 7.91 – 7.74 (m, 2H), 7.62 – 7.36 (m, 3H), 7.26 (dt, *J* = 10.6, 1.9 Hz, 2H), 6.69 (t, *J* = 2.1 Hz, 1H), 3.78 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 166.0, 160.6, 139.7, 135.0, 134.4, 132.0, 128.7, 127.0, 112.7, 110.7, 104.5, 77.3, 77.0, 76.7, 55.5. HRMS: m/z calcd for C₁₄H₁₂NO₂Cl: 262.0635; found: 262.0645;



N-(3,5-dimethoxyphenyl)benzamide (4.117h): The compound **4.117h** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 12 % ethyl acetate in hexanes) in 55% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.24 (s, 1H), 7.92 – 7.77 (m, 2H), 7.57 – 7.35 (m, 3H), 6.95 (d, *J* = 2.2 Hz, 2H), 6.28 (t, *J* = 2.2 Hz, 1H), 3.76 (s, 6H). ¹³C NMR (101 MHz, CDCl₃) δ 166.0, 161.0, 139.8, 134.8, 131.8, 128.6, 127.0, 98.5, 97.1, 77.4, 77.1, 76.8, 55.3. HRMS: m/z calcd for C₁₅H₁₅NO₃: 258.1130; found: 258.1133.



N-(3-bromo-5-methoxyphenyl)benzamide (**4.117i**): The compound **4.117i** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 76 % yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.41 (s, 1H), 7.82 (d, *J* = 7.8 Hz, 2H), 7.59 – 7.19 (m, 5H), 6.82 (s, 1H), 3.74 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 166.2, 160.6, 140.0, 134.4, 132.0, 128.7, 127.1, 122.7, 115.7, 113.6, 105.2, 77.4, 77.1, 76.8, 55.5. HRMS: m/z calcd for C14H12NO₂Br: 306.0130; found: 306.0139.



N-(3-methoxy-5-methylphenyl)benzamide(4.117j): The compound **4.117j** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 82% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.05 (s, 1H), 7.92 – 7.80 (m, 2H), 7.58 – 7.39 (m, 3H), 7.27 – 7.20 (m, 1H), 7.01 (q, *J* = 1.4, 0.9 Hz, 1H), 6.54 (ddd, *J* = 2.2, 1.4, 0.7 Hz, 1H), 3.79 (s, 3H), 2.32 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 165.8, 160.1, 139.9, 138.9, 135.0, 131.7, 128.7, 127.0, 113.2, 111.4, 103.0, 77.4, 77.0, 76.7, 55.2, 21.6. HRMS: m/z calcd for C₁₅H₁₅NO₂: 242.1181; found: 242.1188.



N-(3,5-difluorophenyl)benzamide (4.117k): The compound **4.117k** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 4% ethyl acetate in hexanes) in 53% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.10 (s, 1H), 7.92 – 7.77 (m, 2H), 7.62 – 7.54 (m, 1H), 7.48 (tt, *J* = 6.7, 1.5 Hz, 2H), 7.34 – 7.24 (m, 2H), 6.61 (tt, *J* = 8.9, 2.3 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 165.9, 164.4 (d, *J* = 14.7 Hz), 162.0 (d, *J* = 14.6 Hz), 140.0 (t, *J* = 13.3 Hz), 134.2, 132.3, 128.9, 127.0, 103.4 – 102.8 (m), 99.8 (t, *J* = 25.6 Hz), 77.3, 77.0, 76.7. ¹⁹F NMR (376 MHz, CDCl₃) δ -108.7. HRMS: m/z calcd for C₁₃H₉NOF₂: 234.0730; found: 234.0731.



N-(3-chloro-5-fluorophenyl)benzamide (4.117l): The compound **4.117l** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 4% ethyl acetate in hexanes) in 53% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.26 (s, 1H), 7.89 – 7.75 (m, 2H), 7.61 – 7.37 (m, 5H), 6.87 (dt, J = 8.3, 2.1 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 166.1, 162.8 (d, J = 247.8 Hz), 140.0 (d, J = 12.2 Hz), 135.2 (d, J = 12.4 Hz), 134.1, 132.3, 128.8, 127.1, 115.9 (d, J = 3.2 Hz), 112.1 (d, J = 25.0 Hz), 106.0 (d, J = 26.4 Hz), 77.3, 77.0, 76.7. ¹⁹F NMR (376 MHz, CDCl₃) δ -109.5. HRMS: m/z calcd for C₁₃H₉CIFNO: 250.0435; found: 250.0440.



N-(3,4,5-trifluorophenyl)benzamide (4.117m): The compound **4.117m** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 6 % ethyl acetate in hexanes) in 33 % yield as a white solid. ¹H NMR (400 MHz, DMSO-*d*₆) δ 10.55 (s, 1H), 7.97 – 7.89 (m, 2H), 7.74 (ddd, *J* = 10.5, 6.6, 2.0 Hz, 2H), 7.65 – 7.48 (m, 3H). ¹³C NMR (101 MHz, DMSO) δ ¹³C NMR (101 MHz, DMSO-*d*₆) δ 166.4, 166.3, 151.6 (dd, *J* = 9.9, 5.3 Hz), 149.1 (dd, *J* = 10.0, 5.3 Hz), 136.1 (d, *J* = 3.7 Hz), 136.0 (d, *J* = 3.6 Hz), 135.9 (d, *J* = 2.7 Hz), 134.5 (d, *J* = 5.8 Hz), 132.5, 128.9, 128.1, 104.7 (dd, *J* = 25.0, 7.9 Hz). ¹⁹F NMR (376 MHz, DMSO) δ -135.0, -135.0, -135.0, -167.7, -167.8, -167.8. HRMS: m/z calcd for C₁₃H₈F₃NO: 252.0636; found: 252.0642.



4-fluoro-N-(3-methoxyphenyl)benzamide (4.117o): The compound 4.117o was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 8 % ethyl acetate in hexanes) in 74% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 8.00 (s, 1H), 7.93 – 7.80 (m, 2H), 7.41 (t, *J* = 2.2 Hz, 1H), 7.31 – 7.20 (m, 1H), 7.19 – 7.06 (m, 3H), 6.72 (dd, *J* = 8.2, 2.0 Hz, 1H), 3.82 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 165.5 (d, *J* = 131.0 Hz), 163.6 , 160.2 , 139.0 , 131.1 (d, *J* = 3.1 Hz), 129.7, 129.4 (d, *J* = 9.0 Hz), 115.8 (d, *J* = 22.0 Hz), 112.5, 110.6, 106.0, 77.3, 77.0, 76.7,

55.32. ¹⁹F NMR (376 MHz, CDCl₃) δ -107.41. HRMS: m/z calcd for C₁₄H₁₃FNO:246.0930; found: 246.0936.



4-methoxy-N-(3-methoxyphenyl)benzamide (4.117p): The compound **4.117p** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 15% ethyl acetate in hexanes) in 63% yield as a white solid. ¹H NMR (400 MHz, CDCl₃): δ 7.85 (d, *J* = 8.8 Hz, 3H), 7.46 (t, *J* = 2.2 Hz, 1H), 7.32 – 7.19 (m, 1H), 7.16 – 7.06 (m, 1H), 6.97 (dd, *J* = 8.8, 1.6 Hz, 2H), 6.71 (ddd, *J* = 8.3, 2.5, 0.9 Hz, 1H), 3.86 (d, *J* = 18.2 Hz, 6H). ¹³C NMR (101 MHz, CDCl₃) δ 165.2, 162.4, 160.2, 139.4, 129.6, 128.8, 127.1, 113.9, 112.2, 110.3, 105.7, 77.3, 77.0, 76.7, 55.4, 55.3. HRMS: m/z calcd for C₁₅H₁₅NO₃: 258.1130; found: 258.1140.



N-(2-(pyridin-2-yl)-phenyl)benzamide (4.119a): The compound 4.119a was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 6% ethyl acetate in hexanes) in 89% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 13.30 (s, 1H), 8.99 – 8.47 (m, 2H), 8.07 (dd, *J* = 8.0, 1.8 Hz, 2H), 7.96 – 7.67 (m, 3H), 7.63 – 7.41 (m, 4H), 7.39 – 7.09 (m, 2H). ¹³C NMR (101 MHz, CDCl₃) δ 165.5, 158.1,

147.1, 138.1, 138.0, 135.6, 131.5, 130.3, 128.8, 128.6, 127.3, 125.6, 123.6, 123.0, 122.0, 77.4, 77.1, 76.8. HRMS: m/z calcd for C₁₈H₁₄N₂O: 275.1184; found: 275.1194.



N-(2-(1H-pyrazol-1-yl)phenyl)benzamide (4.119b): The compound **4.119b** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 6% ethyl acetate in hexanes) in 86% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 11.34 (s, 1H), 8.74 (dd, *J* = 8.3, 1.4 Hz, 1H), 8.06 – 7.82 (m, 4H), 7.61 – 7.35 (m, 5H), 7.23 (td, *J* = 7.7, 1.4 Hz, 1H), 6.55 (t, *J* = 2.1 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 165.3, 141.1, 134.8, 131.8, 131.8, 130.2, 129.1, 128.6, 128.1, 127.3, 124.0, 122.9, 122.2, 107.3, 77.3, 77.0, 76.7. HRMS: m/z calcd for C₁₆H₁₃N₃O: 264.1137; found: 264.1149.



N-(benzo[h]quinolin-10-yl)benzamide (4.119c): The compound **4.119c** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 94% yield as a light yellow solid. ¹H NMR (400 MHz, CDCl3) δ 15.65 (s, 1H), 9.35 (dt, *J* = 8.3, 1.5 Hz, 1H), 8.82 (ddd, *J* = 6.6, 4.4, 2.2 Hz, 1H), 8.36 – 8.09 (m, 3H), 7.74 (dtd, *J* = 7.7, 4.9, 2.3 Hz, 2H), 7.66 – 7.44 (m, 6H). ¹³C NMR (101 MHz, CDCl₃) δ 166.3, 147.6, 145.4, 139.6, 136.6, 136.5, 135.0, 131.4, 129.4, 129.0, 128.6, 127.6, 127.5, 125.0, 123.1, 120.7, 118.3, 117.8, 77.3, 77.0, 76.7. HRMS: m/z calcd for C₂₀H₁₄N₂O: 299.1184; found: 299.1190.



N-(1-(pyridin-2-yl)-1H-indol-2-yl)benzamide (4.119d): The compound 4.119d was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 10% ethyl acetate in hexanes) in 66% yield as a white solid. ¹H NMR (400 MHz, Chloroform-*d*) δ 11.95 (s, 1H), 8.55 (dd, J = 5.2, 1.9 Hz, 1H), 8.00 – 7.90 (m, 2H), 7.83 (td, J = 7.8, 2.0 Hz, 1H), 7.70 – 7.61 (m, 2H), 7.59 – 7.45 (m, 4H), 7.37 (s, 1H), 7.26 – 7.13 (m, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 163.6, 151.8, 148.1, 139.5, 134.9, 134.1, 132.0, 131.9, 129.6, 128.8, 127.0, 122.1, 121.8, 120.8, 120.6, 117.7, 110.6, 93.7, 77.6, 77.3, 77.0. HRMS: m/z calcd for C₂₀H₁₅N₃O: 314.1293; found: 314.1300.



N-(5-methoxy-2-(pyridin-2-yl)phenyl)benzamide (4.119e): The compound 4.119e was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 92% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 13.84 (s, 1H), 8.66 – 8.59 (m, 2H), 8.13 – 8.05 (m, 2H), 7.85 – 7.66 (m, 3H), 7.60 – 7.49 (m, 3H), 7.23 (ddd, *J* = 7.3, 4.9, 1.3 Hz, 1H), 6.77 (dd, *J* = 8.8, 2.7 Hz, 1H), 3.95 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 165.8, 161.0, 158.1, 146.9, 140.0, 137.7, 135.8, 131.5, 129.5, 128.6, 127.4, 121.9, 121.1, 117.6, 110.6, 105.5, 77.3, 77.0, 76.7, 55.4. HRMS: m/z calcd for C₁₉H₁₆N₂O₂:305.1290; found: 305.1299.



N-(5-fluoro-2-(pyridin-2-yl)phenyl)benzamide (4.119f): The compound **4.119f** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 5 % ethyl acetate in hexanes) in 85% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 13.70 (s, 1H), 8.81 – 8.53 (m, 2H), 8.13 – 7.96 (m, 2H), 7.90 – 7.65 (m, 3H), 7.54 (qd, J = 8.7, 7.8, 3.5 Hz, 3H), 7.29 (dd, J = 7.4, 4.9 Hz, 1H), 6.88 (ddd, J = 8.9, 7.6, 2.8 Hz, 1H). ¹³C NMR (101 MHz, CDCl₃) δ 165.74 , 163.50 (d, J = 247.7 Hz), 157.46 , 147.14 , 140.11 (d, J = 11.9 Hz), 137.98 , 135.40 , 131.77 , 130.04 (d, J = 9.9 Hz), 128.68 , 127.39 , 122.57 , 121.97 , 121.12 (d, J = 3.2 Hz), 110.32 (d, J = 22.2 Hz), 108.67 (d, J = 27.6 Hz), 77.42 , 77.10 , 76.79 . ¹⁹F NMR (376 MHz, CDCl₃) δ -108.8. HRMS: m/z calcd for C₁₈H₁₃FN₂O: 293.1090; found: 293.1100.



N-(5-methyl-2-(pyridin-2-yl)phenyl)benzamide (4.119g): The compound 4.119g was synthesized according to the general procedure described above and purified by flash column

chromatography (0 to 6% ethyl acetate in hexanes) in 93% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 13.47 (s, 1H), 8.80 – 8.54 (m, 2H), 8.18 – 7.96 (m, 2H), 7.88 – 7.73 (m, 2H), 7.64 (d, *J* = 8.0 Hz, 1H), 7.60 – 7.45 (m, 3H), 7.26 (ddd, *J* = 6.7, 5.6, 3.1 Hz, 1H), 7.03 (dd, *J* = 8.1, 1.9 Hz, 1H), 2.47 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 165.5, 158.3, 147.1, 140.6, 138.1, 137.7, 135.8, 131.4, 128.6, 128.5, 127.3, 124.4, 122.7, 122.5, 122.2, 121.6, 77.4, 77.1, 76.7, 21.7. HRMS: m/z calcd for C₁₉H₁₆N₂O: 289.1341; found: 289.1348.



N-(2-(pyridin-2-yl)-5-(trifluoromethyl)phenyl)benzamide (4.119h): The compound **4.119h** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 63% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 13.47 (s, 1H), 9.21 (d, *J* = 1.8 Hz, 1H), 8.73 (dd, *J* = 4.6, 2.0 Hz, 1H), 8.05 (dd, *J* = 7.9, 1.6 Hz, 2H), 7.98 – 7.78 (m, 3H), 7.63 – 7.49 (m, 3H), 7.49 – 7.34 (m, 2H). ¹³C NMR (101 MHz, Chloroform-*d*) δ 165.7, 157.0, 147.5, 138.7, 138.2, 135.2, 131.9 (q, *J* = 32.6 Hz), 131.8, 129.0, 128.7, 127.9, 127.4, 123.9 (q, *J* = 272.5 Hz), 123.3, 122.9, 119.8 (q, *J* = 3.9, 3.5 Hz), 118.8.0 (q, *J* = 4.2 Hz), 77. 4, 77.1, 76.7. ¹⁹F NMR (376 MHz, CDCl₃) δ -62.8, -62.8. HRMS: m/z calcd for C₁₉H₁₃N₂OF₃:343.1058; found: 343.1068.



4-methoxy-N-(2-(pyridin-2-yl)phenyl)benzamide (4.119i): The compound **4.119i** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 8% ethyl acetate in hexanes) in 54% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 13.24 (s, 1H), 8.82 (dd, *J* = 8.4, 1.2 Hz, 1H), 8.68 (ddd, *J* = 4.9, 1.8, 0.9 Hz, 1H), 8.11 – 7.96 (m, 2H), 7.89 – 7.66 (m, 3H), 7.47 (ddd, *J* = 8.5, 7.3, 1.6 Hz, 1H), 7.34 – 7.11 (m, 2H), 7.08 – 6.91 (m, 2H), 3.87 (s, 3H). ¹³C NMR (101 MHz, CDCl₃) δ 165.1, 162.2, 158.3, 147.2, 138.3, 137.8, 130.2, 129.2, 128.7, 128.0, 125.4, 123.3, 122.9, 121.9, 121.8, 113.8, 77.4, 77.1, 76.8, 55.4. HRMS: m/z calcd for C₁₉H₁₆N₂O₂:305.1290; found: 305.1295.



4-fluoro-N-(2-(pyridin-2-yl)phenyl)benzamide (4.119j): The compound **4.119j** was synthesized according to the general procedure described above and purified by flash column chromatography (0 to 5% ethyl acetate in hexanes) in 85% yield as a white solid. ¹H NMR (400 MHz, CDCl₃) δ 13.36 (s, 1H), 8.88 – 8.58 (m, 2H), 8.13 – 8.02 (m, 2H), 7.93 – 7.79 (m, 2H), 7.76 (dd, J = 7.9, 1.6 Hz, 1H), 7.49 (ddd, J = 8.6, 7.3, 1.6 Hz, 1H), 7.37 – 7.27 (m, 1H), 7.26 – 7.14 (m, 3H). ¹³C NMR (101 MHz, CDCl₃) ¹³C NMR (101 MHz, CDCl₃) δ 164.8 (d, J = 251.9 Hz), 164.4, 158.3, 147.2, 138.1, 137.9, 131.9 (d, J = 3.2 Hz), 130.3, 129.7 (d, J = 8.9 Hz), 128.7, 125.4, 121.95 (d, J = 21.4 Hz), 115.7, 115.5, 77.4, 77.0, 76.7. ¹⁹F NMR (376 MHz, CDCl₃) δ - 108.3. HRMS: m/z calcd for C₁₈H₁₃FN₂O: 293.1090; found: 293.1102.

4.7. References

 (a) Hili, R.; Yudin, A. K. Making Carbon-Nitrogen Bonds in Biological and Chemical Synthesis. *Nat. Chem. Biol.* 2006, *2*, 284–287. (b) Ricci, A. *Amino Group Chemistry:* *From Synthesis to the Life Sciences*; 1st ed.; Wiley-VCH: Weinheim, Germany, 2007. (c)
Kim, J.; Movassaghi, M. Biogenetically-Inspired Total Synthesis
Epidithiodiketopiperazines and Related Alkaloids. *Acc. Chem. Res.* 2015, *48*,
1159–1171. (d) Quintas-Cardama, A.; Cortes, J. Therapeutic Options Against BCR-ABL1 T315I-Positive Chronic Myelogenous Leukemia. *Clin. Cancer Res.* 2008, *14*,
4392–4399. (e) Wakamiya, A.; Nishimura, H.; Fukushima, T.; Suzuki, F.; Saeki, A.;
Seki, S.; Osaka, I.; Sasamori, T.; Murata, M.; Murata, Y.; Kaji, H. On-Top π-Stacking of
Quasiplanar Molecules in Hole-Transporting Materials: Inducing Anisotropic Carrier
Mobility in Amorphous Films. *Angew. Chem., Int. Ed.* 2014, *53*, 5800–5804.

- 2. Ullmann, F. Ber. Dtsch. Chem. Ges. 1903, 36, 2382–2384.
- 3. (a) Guram, A. S.; Buchwald, S. L. Palladium-Catalyzed Aromatic Aminations with in situ Generated Aminostannanes *J. Am. Chem. Soc.* 1994, *116*, 7901–7902. (b) Paul, F.; Patt, J.; Hartwig, J. F. Palladium-catalyzed formation of carbon-nitrogen bonds. Reaction intermediates and catalyst improvements in the hetero cross-coupling of aryl halides and tin amides. *J. Am. Chem. Soc.* 1994, *116*, 5969–5970.
- Shin, K.; Kim, H.; Chang, S. Transition-Metal-Catalyzed C–N Bond Forming Reactions Using Organic Azides as the Nitrogen Source: A Journey for the Mild and Versatile C–H Amination. Acc. Chem. Res. 2015, 48, 1040–1052.
- Booth, G. Nitro Compounds, Aromatic. Ullmann's Encyclopedia of Industrial Chemistry;
 Wiley-VCH: Weinheim, 2005; Chapter 2.
- Park, Y.; Kim, Y.; Chang, S. Transition Metal-Catalyzed C–H Amination: Scope, Mechanism, and Applications. *Chem. Rev.* 2017, *117*, 9247–9301.

- Aguila, M. J. B.; Badiei, Y. M.; Warren, T. H. Mechanistic Insights into C–H Amination via Dicopper Nitrenes. J. Am. Chem. Soc. 2013, 135, 9399–9406.
- 8. (a) Breslow, R.; Gellman, S. H. Tosylamidation of Cyclohexane by a Cytochrome P-450 Model. *J. Chem. Soc., Chem. Commun.* 1982, 1400–1401. (b) Breslow, R.; Gellman, S. H. Intramolecular Nitrene Carbon–Hydrogen Insertions Mediated by Transition-Metal Complexes as Nitrogen Analogs of Cytochrome P-450 Reactions. *J. Am. Chem. Soc.* 1983, *105*, 6728–6729.
- Mansuy, D.; Mahy, J.-P.; Dureault, A.; Bedi, G.; Battioni, P. Iron- and Manganese-Porphyrin Catalysed Aziridination of Alkenes by Tosyl- and Acyl-Iminoiodobenzene. *J. Chem. Soc., Chem. Commun.* 1984, 1161–1163.
- (a) Stokes, B. J.; Dong, H.; Leslie, B. E.; Pumphrey, A. L.; Driver, T. G. Intramolecular C–H Amination Reactions: Exploitation of the Rh₂ (II)-Catalyzed Decomposition of Azido acrylates. *J. Am. Chem. Soc.* 2007, *129*, 7500–7501. (b) Stokes, B. J.; Jovanovic, B.; Dong, H.; Richert, K. J.; Riell, R. D.; Driver, T. G. Rh₂(II)-Catalyzed Synthesis of Carbazoles from Biaryl Azides. *J. Org. Chem.* 2009, *74*, 3225–3228. (c) Stokes, B. J.; Richert, K. J.; Driver, T. G. Examination of the Mechanism of Rh₂(II)-Catalyzed Carbazole Formation Using Intramolecular Competition Experiments. *J. Org. Chem.* 2009, *74*, 6442–6451.
- 11. He, L.; Chan, P. W. H.; Tsui, W.-M.; Yu, W.-Y.; Che, C.-M. Ruthenium(II) Porphyrin-Catalyzed Amidation of Aromatic Heterocycles. *Org. Lett.* **2004**, *6*, 2405–2408.
- Paudyal, M. P.; Adebesin, A.; Burt, S. R.; Ess, D. H.; Ma, Z.; Kürti, L.; Falck, J. R.
 Science. 2016, 353, 1144-1147.

- Skubi, K. L.; Blum, T. R.; Yoon, T. P. Dual Catalysis Strategies in Photochemical Synthesis. *Chem. Rev.* 2016, *116*, 10035–10074.
- (a) Narayanam, J. M. R.; Stephenson, C. R. J. Visible Light Photoredox Catalysis: Applications in Organic Synthesis. *Chem. Soc. Rev.* 2011, *40*, 102–113. (b) Kärkä s, M. D.; Porco, J. A.; Stephenson, C. R. J. Photochemical Approaches To Complex Chemotypes: Applications in Natural Product Synthesis. *Chem. Rev.* 2016, *116*, 9683–9747.
- Jiang, H.; An, X.; Tong, K.; Zheng, T.; Zhang, Y.; Yu, S. Visible-Light-Promoted Iminyl-Radical Formation from Acyl Oximes: A Unified Approach To Pyridines, Quinolines, and Phenanthridines. *Angew. Chem., Int. Ed.* 2015, *54*, 4055–4059.
- Allen, L. J.; Cabrera, P. J.; Lee, M.; Sanford, M. S. N-Acyloxy phthalimides as Nitrogen Radical Precursors in the Visible Light Photocatalyzed Room Temperature C–H Amination of Arenes and Heteroarenes. *J. Am. Chem. Soc.* 2014, *136*, 5607–5610.
- Brachet, E.; Ghosh, T.; Ghosh, I.; König, B. Visible Light C–H Amidation of Heteroarenes with Benzoyl Azides. *Chem. Sci.* 2015, *6*, 987–992.
- 18. (a) Labinger, J. A.; Bercaw, J. E. Understanding and Exploiting C–H Bond Activation. *Nature*. 2002, *417*, 507–514. (b) Jazzar, R.; Hitce, J.; Renaudat, A.; Sofack-Kreutzer, J.; Baudoin, O. Functionalization of Organic Molecules by Transition-Metal Catalyzed C(sp³)–H Activation. *Chem. - Eur. J.* 2010, *16*, 2654–2672.
- Monguchi, D.; Fujiwara, T.; Furukawa, H.; Mori, A. Direct Amination of Azoles via Catalytic C–H, N–H Coupling. *Org. Lett.* 2009, *11*, 1607–1610.
- Wang, Q.; Schreiber, S. L. Copper-Mediated Amidation of Heterocyclic and Aromatic C–H Bonds. Org. Lett. 2009, 11, 5178–5180.

- 21. Matsuda, N.; Hirano, K.; Satoh, T.; Miura, Copper-Catalyzedzed Direct Amination of Electron-Deficient Arenes with Hydroxylamines. *Org. Lett.* **2011**, *13*, 2860–2863.
- Shrestha, R.; Mukherjee, P.; Tan, Y.; Litman, Z. C.; Hartwig, J. F. Sterically Controlled, Palladium-Catalyzed Intermolecular Amination of Arenes. *J. Am. Chem. Soc.* 2013, *135*, 8480–8483.
- 23. Tsang, W. C. P.; Zheng, N.; Buchwald, S. L. Combined C–H Functionalization/C–N
 Bond Formation Route to Carbazoles. *J. Am. Chem. Soc.* 2005, *127*, 14560–14561.
- Jordan-Hore, J. A.; Johansson, C. C. C.; Gulias, M.; Beck, E. M.; Gaunt, M. J. Oxidative Pd(II)-Catalyzed C–H Bond Amination to Carbazole at Ambient Temperature. *J. Am. Chem. Soc.* 2008, *130*, 16184–16186.
- Tan, Y.; Hartwig, J. F. Palladium-Catalyzed Amination of Aromatic C–H Bonds with Oxime Esters. J. Am. Chem. Soc. 2010, 132, 3676–3677.
- Miura, T.; Ito, Y.; Murakami, M. Synthesis of Oxindoles by Palladium-catalyzed C–H Bond Amidation. *Chem. Lett.* 2009, *38*, 328–329.
- Inamoto, K.; Saito, T.; Katsuno, M.; Sakamoto, T.; Hiroya, K. Palladium-Catalyzed C–H Activation/Intramolecular Amination Reaction: A New Route to 3-Aryl/Alkyl indazoles. *Org. Lett.* 2007, *9*, 2931–2934.
- Clagg, K.; Hou, H.; Weinstein, A. B.; Russell, D.; Stahl, S. S.; Koenig, S. G. Synthesis of Indole-2-carboxylate Derivatives via Palladium-Catalyzed Aerobic Amination of Aryl C–H Bonds. Org. Lett. 2016, 18, 3586–3589.
- Xiao, Q.; Wang, W. H.; Liu, G.; Meng, F. K.; Chen, J. H.; Yang, Z.; Shi, Z. J. Direct Imidation To Construct 1H-Benzo[d]imidazole through Pd^{II} -Catalyzed C–H Activation Promoted by Thiourea. *Chem. - Eur. J.* 2009, *15*, 7292–7296.

- Dick, A. R.; Remy, M. S.; Kampf, J. W.; Sanford, M. S. Carbon–Nitrogen Bond-Forming Reactions of Palladacycles with Hypervalent Iodine Reagents. *Organometallics*. 2007, 26, 1365–1370.
- Thu, H.-Y.; Yu, W.-Y.; Che, C.-M. Intermolecular Amidation of Unactivated sp² and sp³
 C–H Bonds via Palladium-Catalyzed Cascade C–H Activation/Nitrene Insertion. *J. Am. Chem. Soc.* 2006, *128*, 9048–9049.
- Xiao, B.; Gong, T.-J.; Xu, J.; Liu, Z.-J.; Liu, Palladium-Catalyzedzed Intermolecular Directed C–H Amidation of Aromatic Ketones. *J. Am. Chem. Soc.* 2011, *133*, 1466–1474.
- 33. Ng, K.-H.; Chan, A.S.C.; Yu,W.-Y. Pd-Catalyzed Intermolecular *ortho*-C–H Amidation of Anilides by N-Nosyloxycarbamate. *J. Am. Chem. Soc.* **2010**, *132*, 12862–12864.
- 34. (a) Yoo, E. J.; Ma, S.; Mei, T.-S.; Chan, K. S. L.; Yu, J.-Q. Pd-Catalyzed Intermolecular C–H Amination with Alkylamines. *J. Am. Chem. Soc.* 2011, *133*, 7652–7655. (b)
 Grohmann, C.; Wang, H.; Glorius, F. Rh[III]-Catalyzed C–H Amidation Using Aroyloxycarbamates to Give N-Boc Protected Arylamines. *Org. Lett.* 2013, *15*, 3014–3017.

(c) Raghuvanshi, K.; Zell, D.; Rauch, K.; Ackermann, L. Ketone Assisted Ruthenium(II) Catalyzed C–H Imidatioaccess tosTo Primary Aminoketones by Weak Coordination.
 ACS Catal. 2016, 6, 3172–3175.

35. Kim, J. Y.; Park, S. H.; Ryu, J.; Hwan, S. C.; Kim, S. H.; Chang, S. Rhodium-Catalyzed Intermolecular Amidation of Arenes with Sulfonyl Azides via Chelation-Assisted C–H Bond Activation. J. Am. Chem. Soc. 2012, 134, 9110–9113.
- Park, S. H.; Kwak, J.; Shin, K.; Ryu, J.; Park, Y.; Chang, S. Mechanistic Studies of the Rhodium-Catalyzed Direct C–H Amination Reaction Using Azides as the Nitrogen Source. J. Am. Chem. Soc. 2014, 136, 2492–2502.
- 37. Sun, B.; Yoshino, T.; Matsunaga, S.; Kanai, M. Air-Stable
 Carbonyl(pentamethylcyclopentadienyl)cobalt Diiodide Complex as a Precursor for
 Cationic (Pentamethylcyclopentadienyl)cobalt(III) Catalysis: Application for Directed C2 Selective C–H Amidation of Indoles. *Adv. Synth. Catal.* 2014, *356*, 1491–1495.
- Ryu, J.; Kwak, J.; Shin, K.; Lee, D.; Chang, S. Ir(III)-Catalyzed Mild C–H Amidation of Arenes and Alkenes: An Efficient Usage of Acyl Azides as the Nitrogen Source. *J. Am. Chem. Soc.* 2013, *135*, 12861–12868.
- Ryu, J.; Shin, K.; Park, S. H.; Kim, J. Y.; Chang, Rhodium-Catalyzedzed Direct C–H Amination of Benzamides with Aryl Azides: A Synthetic Route to Diarylamines. *Angew. Chem., Int. Ed.* 2012, *51*, 9904–9908.
- 40. Park, Y.; Park, K. T.; Kim, J. G.; Chang, S. Mechanistic Studies on the Rh(III)-Mediated Amido Transfer Process Leading To Robust C–H Amination with a New Type of Amidating Reagent. *J. Am. Chem. Soc.* **2015**, *137*, 4534–4542.
- 41. Patel, P.; Chang, S. Cobalt(III)-Catalyzed C–H Amidation of Arenes using Acetoxycarbamates as Convenient Amino Sources under Mild Conditions *ACS Catal*.
 2015, 5, 853-858.
- 42. (a) Wang, H.; Lorion, M. M.; Ackermann, L. Overcoming the Limitations of C–H Activation with Strongly Coordinating N-Heterocycles by Cobalt Catalysis *Angew*. *Chem., Int. Ed.*, **2016**, *55*, 10386-10390; (b) Wang, F.; Wang, H.; Wang, Q.; Yu, S.; and Li, X. Highly Chemoselective Reduction of Carbonyl Groups in the Presence of

Aldehydes. *Org. Lett.* **2016**, *18*, 1306-1309; (c) Wang, X.; Lerchen, A.; and Glorius, F. A Comparative Investigation: Group 9 Cp*M(III)-Catalyzed Formal [4 + 2] Cycloaddition as an Atom-Economic Approach to Quinazolines. *Org. Lett.* **2016**, *18*, 2090-2093.

- Wang, S.; Chen, S-Y.; Yu, X-Q. C–H functionalization by high-valent Cp*Co(III) catalysis.
 Chem. Commun. 2017, *53*,3165-3180.
- 44. Jun, C.-H.; Lee, H.; Lim, S.-G. The C-C Bond Activation and Skeletal Rearrangement of Cycloalkanone Imine by Rh(I) Catalysts. *J. Am. Chem. Soc.* **2001**, *123*, 751–752.
- 45. (a) Li, H.; Li, Y.; Zhang, X.-S.; Chen, K.; Wang, X.; Shi, Z.-J. Pyridinyl Directed Alkenylation with Olefins via Rh(III)-Catalyzed C-C Bond Cleavage of Secondary Arylmethanols. *J. Am. Chem. Soc.* 2011, *133*, 15244–15247. (b) Chen, K.; Li, H.; Li, Y.; Zhang, X.-S.; Lei, Z.-Q.; Shi, Z.-J. *Chem. Sci.* 2012, *3*, 1645–1649.
- Lei, Z.-Q.; Li, H.; Li, Y.; Zhang, X.-S.; Chen, K.; Wang, X.; Sun, J.; Shi, Z.-J. Extrusion of CO from Aryl Ketones: Rhodium(I)-Catalyzed C-C Bond Cleavage Directed by a Pyridine Group. *Angew. Chem., Int. Ed.* 2012, *51*, 2690–2694.
- 47. Zeng, R.; Dong, G. Rh-Catalyzed Decarbonylative Coupling with Alkynes via C–C
 Activation of Isatins. *J. Am. Chem. Soc.* 2015, *137*, 1408–1411.
- 48. Xia, Y.; Lu, G.; Liu, P.; Dong, G. Catalytic activation carbon-carbonbon bonds in cyclopentanones. *Nature*. **2016**, *539*, 546–550.
- Morioka, T.; Nishizawa, A.; Furukawa, T.; Tobisu, M.; Chatani, N. Nickel-Mediated Decarbonylation of Simple Unstrained Ketones through the Cleavage Carbon-Carbonbon Bonds. J. Am. Chem. Soc. 2017, 139, 1416-1419.

- Zhao, T-T.; Xu, W-H.; Zheng, Z-J.; Xu, P-F.; Wei, H.; Directed Decarbonylation of Unstrained Aryl Ketones vNickel-Catalyzedzed C-C Bond Cleavage. *J. Am. Chem. Soc.* 2018, *140*, 586–589.
- Nanteuil, F.; and Waser, J. Catalytic [3+2] Annulation of Aminocyclopropanes for the Enantiospecific Synthesis of Cyclopentylamines. *Angew. Chem. Int. Ed.* 2011, *50*, 12075-12079.
- Couturier, M.; Tucker, J. L.; Proulx, C.; Boucher, G.; Dubé, P.; Andresen, B. M.; and Ghosh, A.; 5,5-Dimethyl-1,4,2-dioxazoles as Versatile Aprotic Hydroxamic Acid Protecting Groups. *J. Org. Chem.* 2002, 67, 4833-4838.
- 53. Uchida, Y.; Kozuka, S. Reaction of N-Aroyl-N-t-butylhydroxylamines with Thionyl Chloride. Synthesis of Substituted Benzohydroximoyl Chlorides. *Bull. Chem. Soc. Jpn.* 1984, 57, 2011-2012.
- Bizet, V.; Buglioni, L; Bolm, C. Light-Induced Ruthenium-Catalyzed Nitrene Transfer Reactions: A Photochemical Approach towards N-Acyl Sulfimides and Sulfoximines. *Angew. Chem., Int. Ed.* 2014, *53*, 5639-5642.
- a) Carpino, L. A.; Carpino, B. A.; Crowley, P. J.; Giza, C. A.; Terry, P. H.; Synthesis of *t*-butyl azidoformate. *Org. Synth.* 1964, *44*, 15. b) Wippich, J.; Truchan, N.; Bach, T. Rhodium-Catalyzed *N-tert*-butoxycarbonyl (Boc) Amination by Directed C-H Bond Activation. *Adv. Synth. Catal.* 2016, *358*, 2083-2087.
- (a). Liu, J., Mandel, S., Hadad, C. M., Platz, M. S. A Comparison of Acetyl- and Methoxycarbonylnitrenes by Computational Methods and a Laser Flash Photolysis Study of Benzoylnitrene. *J. Org. Chem.* 2004, *69*, 8583-8593. b). Ryu,J.; Kwak, J.; Shin, K.; Lee, D.; Chang, S. *J. Am. Chem. Soc.*, 2013, *135*, 12861.

5. CONCLUSION

In summary, we have developed three new catalytic methods through directing groupassisted transition metal catalysed C-H bond functionalisation. One of the main challenges in directed C-H bond activation is installation and removole of directing groups which hurts the step and atom economy of direct C-H bond functionalisation. To overcome the aforementioned challenge in directed C-H bond functionalisation we aimed at heteroatom-based directing groups (DGs) that are multi-tasking, transformable, and/or readily removable.

We demonstrated a ruthenium/NHC based catalytic system with intramolecular alkene hydroarylation of N-aryl acrylamides to form oxindoles, which are important synthetic targets due to their biological activities. The transformation proceeds via tandem sequence of amidedirected aromatic C–H bond activation and intramolecular alkene arylmetalation. This Rucatalyzed cyclization process does not require additional DG assistance and the directing group is incrporated into the product backnone. With the redox-neutral nature and high atom-efficiency of alkene hydroarylation, the reaction is compatible with broad range of aromatic and vinyl substituents.

Next, we explored carboxylic acids as a traceless or removable directing group for chelation assited C-H bond functionaisation. we disclosed a protocol for the alkenylation of various heteroarenes through decarboxylative hydroheteroarylation of alkynes. The reaction occrs via carboxylate-directed C-H activation, alkyne insertion, and subsequent decarboxylation. This method enbles the alkenylation of heteroarens such as Indole, Benzofuran, Benzothiophene, Thiophene and Pyrrole at both C-2 and C-3 positions by overcoming their inherent reactivity. In addition, we have extended the scope of this decarboxylative C-H functionalization by coupling hateroarene carboxylicacids and acrylates under modified catalytic conditions.

Metal-catalyzed decarboxylation enables the use of arenecarboxylic acids as readily available aromatic building blocks. Long-standing challenges in this area include harnessing the effects of aromatic substituents on decarboxylation activity and achieving broader scopes and controlled regiochemistry for subsequent arene functionalization. Towards these goals, we have developed a Rh/Ag-bimetallic catalyst for decarboxylative amidation of *ortho*-substituted benzoic acids with 3-aryldioxazolones. Regioselectivity of this reaction is determined by nature of *ortho*-substituents through two forms of proposed chelation assistance: (1) A wide range of non-directing *ortho*-substituents led to *ortho*-amidation products via carboxylate-directed C-H amidation and subsequent decarboxylation. (2) 2-Pyridyl and analogous DGs led to *ipso*amidation products via DG-assisted decarboxylation and subsequent amidation. Although this protocol works well for the *ortho* substituted arene carboxylic acids but it's not suitable for other arenen carboxylic acids such as *para* and *meta* substituted arene carboxylative C-H bond functionalisation which is suitable for broad range of arenes carboxylic acids.

Although significant advancements have been made in the field of transition metal catalysed chelation assisted C-H bond functionalisation, there is still room for the improvement. Notably, it is highly desirable to develop methods for C-H bond functionalisation with low catalytic loading and low temperature. It is necessary to explore less explore less expensive metals such as Cobalt, Nickel and Copper for the C-H bond functionalisation.